Prairie View A&M University
Digital Commons @PVAMU

All Dissertations

Dissertations

5-2024

An Examination Of Risk Factors For Suicidality Among Adolescents In The United States

Kaviya Shankar Prairie View A&M University, kshankar@pvamu.edu

Follow this and additional works at: https://digitalcommons.pvamu.edu/pvamu-dissertations

Recommended Citation

Shankar, K. (2024). An Examination Of Risk Factors For Suicidality Among Adolescents In The United States. Retrieved from https://digitalcommons.pvamu.edu/pvamu-dissertations/40

This Dissertation is brought to you for free and open access by the Dissertations at Digital Commons @PVAMU. It has been accepted for inclusion in All Dissertations by an authorized administrator of Digital Commons @PVAMU. For more information, please contact hvkoshy@pvamu.edu.

AN EXAMINATION OF RISK FACTORS FOR SUICIDALITY AMONG ADOLESCENTS IN THE UNITED STATES

A Dissertation

by

KAVIYA SHANKAR

Submitted to the Office of Graduate Studies of Prairie View A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

May 2024

Major Subject: Juvenile Justice

AN EXAMINATION OF RISK FACTORS FOR SUICIDALITY AMONG

ADOLESCENTS IN THE UNITED STATES

A Dissertation

by

KAVIYA SHANKAR

Submitted to the Office of Graduate Studies of Prairie View A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Approved as to style and content by:

Camille Gibson Chair of Committee Interim Dean of College Myrna Cintron Member Head of Department

Ling Wu Member Gbolahan Solomon Osho Member

Tyrone Tanner Dean of Graduate Studies

May 2024

Major Subject: Juvenile Justice

ABSTRACT

An Examination of Risk Factors for Suicidality Among Adolescents in the United States (May 2024)

Kaviya Shankar, B. Sc., Jain University

M.Sc., University of Madras

Chair of Advisory Committee: Dr. Camille Gibson

Suicide is considered a significant public health issue, identified as the second leading cause of early mortality in the United States among adolescents (CDC, 2023). The National Institute of Mental Health (2023) defined it as "death induced by self-directed destructive behavior with intent to die" (p. 4). For youth aged 10-14 suicide increased by 36% from 2000 to 2021 in the United States. In recent decades, suicide among adolescents has increased despite estimates of stable or dropping suicide rates in developed countries. Every year, 703,000 youth worldwide commit suicide, and many more attempt (WHO, 2022).

This study used a nationally representative sample of adolescents from the 2021 High School Youth Risk Behavior surveys. Data were examined with a General Strain theoretical framework utilizing logistic regression, and linear regression to understand the impact of empirical risk factors, physical, dating, sexual victimization, bullying, and cyberbullying on youth mental health issues, physical well-being that includes exercises and workouts, and suicidality among high school students. During the early months of the COVID-19 pandemic (2020 into 2021), youth experiences of physical and sexual violence/victimization may have increased, given their isolation. While dating violence

iii

increased, it is not clear how this impacted suicides. Youth likely experienced more cyberbullying, given their increased interactions online.

This study revealed significant relationships between various forms of victimization, that is physical, sexual, dating violence, bullying, and cyberbullying, and the mental health and physical well-being of high school students, which in turn influenced their risk of suicidality. The regression analyses highlighted that these forms of victimization were predictors of increased mental health issues, which were directly linked to higher suicidality rates among adolescents. The findings are consistent with the literature, as victimization indicators are expected to be related to mental health issues and physical wellness. Sexual victimization was more impactful than physical victimization and dating victimization on mental health and cyberbullying was more impactful than more traditional bullying on mental health. Relatedly mental health issues significantly affected high school students' physical wellness. This study's results offer empirical details that should be informative for policymakers in prioritizing their efforts to reduce youth health risk and victimization.

Keywords: suicidality, victimization, mental health issues, physical wellness, adolescents.

iv

AN EXAMINATION OF RISK FACTORS FOR SUICIDALITY AMONG ADOLESCENTS IN THE UNITED STATES

@ COPYRIGHT 2024

KAVIYA SHANKAR

All Rights Reserved

DEDICATION

This dissertation is dedicated to my Appa, Shankar; my Amma, Uma; and my sister, Bhuvanasree, for always being there for me. To Vijaykrishnan, thank you for being my distant yet closest companion and sending virtual hugs that feel warmer. Special thanks to Anbu Anna for believing in me, Sherie Sam for keeping me going for every laugh and eye-roll at my good jokes and ravings, and her family for constant encouragement. To my Grandfathers (Perumal and Ayyasamy), Aathuma, Gomathi, Iyyappan, Elamurugu, Soundararajan, Ravi, Baskar, Kannan, and Praisy and family. To my incredible friends Muthu, Navya, Divya, Grace, R.S., Lavanya, Anamika, Maliha, Arpitha, Monica, Varsha, and Mokka family members for your friendship and comfort.

Finally, to all those who have supported me throughout my academic journey, including my mentors, teachers, professors, and the many individuals who have positively impacted my life. This work is not mine alone but proof of the love and belief you all have invested in me. And finally, to the tunes that kept me sane during every sleepless work night.

Thank you all from the bottom of my heart!

vi

ACKNOWLEDGEMENTS

This journey of completing the dissertation would not have been possible without the support, guidance, and encouragement of many people to whom I owe my deepest gratitude. First and foremost, I would like to thank my supervisor, Dr. Camille Gibson, for her support, patience, and invaluable guidance throughout this research, even with her busy schedule. Your expertise and insights have significantly shaped this work, and your encouragement kept me motivated during the toughest times.

I am also grateful to my committee members, Dr. Myrna Cintron, Dr. Gbolahan Solomon Osho, and Dr. Ling Wu, whose insightful critiques and valuable suggestions enriched this dissertation. Their expertise and encouragement were instrumental in sculpting this academic endeavor into its final form. I also thank Dr. Thiagarajan Ramakrishnan for sharing innovative research techniques and insights and walking through me in the analysis process. His willingness to share his expertise and time has been invaluable.

Thanks to Dr. Nabil Ouassini, Dr. Shantae Motley, Dr. Sesha Kethinini, Dr. Abdullah Cihan, Dr. Robin Jackson, and also Ms. Sandy Seigmund for fostering an environment of collaboration, and intellectual curiosity and for the moments of levity amidst the rigors of research from the beginning. Thanks to my colleagues and cohorts. This dissertation is a testament to the collective effort and encouragement of all those mentioned above. I am immensely grateful for each of you and proud to share this accomplishment with you!

vii

TABLE OF CONTENTS

| ABSTRACTiii |
|--|
| DEDICATIONvi |
| ACKNOWLEDGEMENTS vii |
| TABLE OF CONTENTS |
| LIST OF FIGURES x |
| LIST OF TABLES xi |
| CHAPTER |
| I INTRODUCTION |
| Significance of the Study6Theories7Research Questions and Hypotheses10Organization of the Study12Summary13 |
| II LITERATURE REVIEW 14 |
| Suicidality14Experiences of Victimization and Suicidality16Physical Victimization17Sexual Victimization19Dating Violence Victimization21Bullying and Suicidality24Cyberbullying and Suicidality27Physical Wellness and Suicidality31Mental Health and Suicidality37Summary42 |
| III METHOD |
| Research Design43Data Source – Youth Risk Behavior Survey (YRBS)45Data Collection46Characteristics of the Data47YRBS Overall Participants49Validity and Reliability of the Data50 |

| Endogenous (Outcome) Variable- Suicidality | 52 |
|---|-----|
| Control Variables | 53 |
| Exogenous (Indicator) Measures | 54 |
| Pre-Analysis | 56 |
| Analyses | 59 |
| Assumptions for Regression | 60 |
| Demographic Data | 61 |
| Summary | 62 |
| IV RESULTS | 63 |
| Examination of Risk Factors for Suicidality Among Adolescents | 66 |
| Summary | 78 |
| V. DISCUSSION | 79 |
| Summary of Findings | 79 |
| Contributions to the Literature | 81 |
| Support for Strain Theories | 85 |
| Comparing 2019 and 2021 Data | 86 |
| Limitations of the Study | 89 |
| Future Research and Recommendations | 89 |
| Conclusion | 91 |
| REFERENCES | 92 |
| APPENDICES | 135 |
| CURRICULUM VITAE | 137 |

LIST OF FIGURES

| FIGURE | Page |
|---|------|
| Figure 1: Strain Theory of Suicide by Zhang | 10 |
| Figure 2: Expected Model of Youth Suicidality | 11 |
| Figure 3: Pearson Correlation of Observed and Latent Constructs | 64 |

LIST OF TABLES

| TABLE | | Page |
|----------------------------|---|------------------|
| Table 1: Study | Variables | 54 |
| Table 2: Sample | le Demographics | 61 |
| Table 3: Pearson | on Correlation of Observed and Latent Constructs | 65 |
| Table 4: Regres Bullyin | ession Analysis Predicting Mental Health Issues from Victiming and Cyberbullying | nizations, 68 |
| Table 5: Regres through | ession Analysis Predicting Physical Wellness Directly and In gh Mental Health Issues | directly 71 |
| Table 6: Regres | ession Analysis for Mental Health Issues Predicting Physical | Wellness 72 |
| Table 7: Logisti | tic Regression Analysis Predicting Suicidality | 74 |
| Table 8: Logisti Victim | tic Regression Analysis Predicting Suicidality Based on For nization | ms of 77 |
| Table 9: Compa | parison of Logistic Regression Coefficients for YRBS 2019 | and 2021 87 |

CHAPTER I

INTRODUCTION

In the United States, suicide is the second cause of death of youth 10-14 (CDC, 2023). Yet not enough is known about the factors that lead adolescents to commit suicide (Ruch et al., 2019). Relatively few studies have examined the significant risk factors of multiple variables associated with various aspects, including physical and mental health and victimization of high school students with suicide. Youth who have suffered physical, sexual, and dating violence are two to three times more likely to attempt suicide (Labuhn et al., 2021).

Limited research focuses on the impact of repeated violence experiences linked with suicide (Franklin et al., 2017). Therefore, to develop comprehensive juvenile suicide prevention approaches, it is imperative to have a deeper knowledge of suicidality and concurrent victimization (Yang, 2023). Using a nationally representative sample of adolescents YRBS (Yang et al., 2021; 2020), this study examined physical, dating, and sexual victimization, bullying, and cyberbullying effects on mental health, physical wellbeing, and adolescents' suicidality, that is suicide plan and suicide attempt, during the lockdowns of the COVID-19 pandemic. Suicidality is "the risk of suicide, usually indicated by suicidal ideation or intent, especially as evident in the presence of a wellelaborated suicidal plan" (VandenBos, 2007 p. 1048).

According to Ehlman et al. (2022), suicide ranked as the country's 10th most common cause of death in 2018. Suicide rates among 14–18-year-olds increased

This dissertation follows the style of the Publication Manual of the American Psychological Association 7th edition. dramatically between 2009 and 2018. Therefore, youth mental distress and suicidal actions have emerged as serious concerns within public health (Casey et al., 2011; Hawton et al., 2013). Suicide is the second leading cause of death among high school students after accidents. Traditional suicide rates are more significant for men than women in all age categories. However, new CDC figures show that suicide rates among 15-to-19-year-old girls more than doubled from 2007 to 2015, compared to a 31% increase in boys. A separate analysis shows that suicide rates for 10–14-year-old girls tripled from 1999 to 2014 (Ruch et al., 2019).

Prior to the pandemic, many studies discussed the significance of suicide. For example, Ehlman et al. (2022) concentrated on US public health issues presented by suicidal adolescents and young adults aged 10–24. However, recent times have seen alterations and limitations in the lives of young individuals. Economic difficulty within the family, restricted access to healthcare and educational opportunities, and social and recreational activities may all contribute to a decrease in stress management, a rise in disputes, and intrafamilial violence (Fegert et al., 2020). All these variables may increase teenage anxiety and despair.

However, Manczak et al. (2019) stated that spending more time together at home might strengthen intrafamilial bonding. School closures may have lowered academic and peer stress. These positive factors may have reduced teenagers' anxiety and depression in the early pandemic and reduced their suicide risk (Orben et al., 2020). Yet, adolescents, those 10–24 years old, who tend to be very peer-oriented may suffer most from physical and social isolation. Social contact is delicate throughout adolescence (Sawyer et al., 2018). Given these points, this study addressed the belief that teenagers are

hypersensitive to social cues and the detrimental impacts of social isolation (Blakemore & Mills, 2014). According to the CDC, high school students reported having ongoing depressive or hopeless thoughts in 2021 (CDC, 2023).

Depression, anxiety, loneliness, and diminished social support due to the COVID -19 epidemic and lockdowns may have raised teenage suicide risk (Bridge et al., 2023). The suicide rate fell in 2019 and 2020 before almost reaching its 2018 high in 2021. The rate over this timeframe indicated a notable increase in African American adolescent suicide rates, which went from 8.2 per 100,000 in 2018 to 11.2 per 100,000 in 2021. With a suicide rate of 36.3 per 100,000 in 2021, American Indian and Alaska Native adolescents had significantly greater rates of suicide. In 2021, the rate for White children was 12.4 per 100,000, the rate for Hispanic or Latino youth was 7.9, and the rate for Asian youth was 9.4. (Pappas, 2023).

While puberty causes hormonal and biochemical changes, adolescence also involves significant psychological and social changes (Sawyer et al., 2018). Socialization and peer relationships become crucial throughout adolescence. Compared to children under 10, teenagers spend more time with peers and develop more complicated interactions with them (Somerville, 2013). Peer acceptance and influence become more important throughout adolescence. Adolescence improves cognitive capacities through self-referential processing, executive control, and mentalizing, helping youth to understand and accept others' points of view (Foulkes & Blakemore, 2016). Despite considerable research efforts about mental health issues, understanding the causes, warning signs, and effective prevention strategies for youth suicide remains incomplete, highlighting a significant gap in the knowledge. The initial months of the COVID-19 pandemic saw high rates of depression, anxiety, and other mental health issues for adults (Aknin et al., 2022). However, there are fewer concentrated studies on children and adolescents (John et al., 2020). While recent research suggests that suicide rates vary by country, in the United States, increased social isolation and mental and economic distress correlated with higher rates of suicidal behavior (Bersia et al., 2022; Dsouza et al., 2020; Gruber et al., 2021; Holmes et al., 2020; Isumi et al., 2020; Mamun & Griffiths, 2020).

Mental health and COVID alone did not impact suicidality among adolescents. However, research mainly focuses on the risk factors associated with mental health that predict suicidal ideations (Klonsky et al., 2016; Morneau-Vaillancourt et al., 2023). Other variables that make people consider suicide ideas to trying to kill themselves are still mostly unclear (Klonsky et al., 2016; Nock et al., 2008). The unknown variables highlight a significant shortfall in the existing knowledge on suicidality.

According to Farrokhi et al. (2024) and Pathania et al. (2023), physical well-being also matters in suicidal ideation. Some physical well-being may include acknowledging the need to exercise, consume nutritious meals, obtain sufficient sleep, avoid sickness and injury, or effectively manage chronic health disorders. Mental and physical health influence an individual's holistic well-being. Physical activity is any movement of the body that requires energy and is produced by the muscles (Caspersen et al., 1985). Substantial evidence indicates that engaging in physical activity demonstrates efficacy in diminishing various significant risk factors associated with suicide (Parker at al., 2021).

Although mental health and suicide are closely interconnected, it is crucial to recognize that not everyone with mental health issues experience suicidal thoughts or

behaviors. While a direct predictive relationship between physical exercise and suicidal thoughts has not been shown, previous research has indicated notable connections between this health habit and many mental health indicators among adolescents (Arat & Wong, 2017; Parker et al., 2021). Thus, this study explored how physical wellness may mediate any connection between the negative impact of youth victimization and suicidality.

Apart from physical and mental well-being, the experience of victimization also plays a significant role in suicidality. Notably, adolescence and early adulthood are developmental stages during which individuals may have heightened vulnerability to suicidal ideation and behavior. Studies, such as those conducted by Arango et al. (2016) and Modecki et al. (2014), suggested that those youth who experience a sense of oppression and do not have a sense of belonging may exhibit a lack of fear towards physical harm or death. This was especially true if they had experienced victimization by different types of violence repeatedly.

Several studies found a correlation between dating violence, sexual victimization, and youth suicidality (Henry et al., 2013; Litwiller & Brausch, 2013). Further, victims of bullying and dating violence have a higher likelihood of experiencing specific outcomes related to suicidality (Fedina et al., 2023). Due to the significant neurobiological, physiological, and social changes that occur during adolescence, these prior victimization experiences may have a strong mental impact that is influenced by their social environment and interactions with peers (Abrutyn & Mueller, 2014; Blakemore & Mills, 2014; Greydanus et al., 2013). Many studies categorized adverse childhood experience as a significant predictor of suicidal ideation, which includes emotional, physical, and sexual maltreatment as well as neglect (Brodsky & Stanley, 2008; Fuller-Thomson et al., 2016; Thompson et al., 2018). Favril et al. (2020) conducted a systematic review of a meta-analysis of 35 studies examining the prevalence and risk of suicide. The risk factors included singleness, violent offenses, prior incarceration, and psychological issues like trauma in a case of victimization. These factors were associated with self-harm in females more than males.

There is a notable disparity in suicide mortality rates based on gender. The prevalence of suicide attempts is higher among female adolescents, at nine percent, compared to male youth at five percent (Kann et al., 2018). However, the reverse gender trend is the case for adults. A significant public health concern is the correlation between adolescent dating violence and an increase in suicidal behavior (Greydanus et al., 2013). Nevertheless, depression remains one of the most significant risk factors for suicide (Hawton et al., 2013). This lack of a complete understanding is particularly alarming considering the increasing rates of suicide among young people, which emphasizes the urgent need for more targeted research and intervention strategies.

Significance of the Study

The purpose of the present study was to examine the experiences of various victimizations, including physical, sexual, dating victimizations, bullying, and cyberbullying, on physical and mental well-being and how these related to suicidality nationwide for high school students early during the COVID-19 pandemic lockdown. With the rise of social isolation and other public health issues during the pandemic 2020

and 2021, it is important to understand how the lockdown's unique dynamics affected youth suicidality.

Theories

This study draws upon the foundational ideas of Strain theories. While these theories are not tested, they provide valuable insight to examine and understand the risk factors of suicidality. The study reflects ideas from three theorists. First, Durkheim (1897), whose book *Suicide: A Study in Sociology*, investigated the connection between various social variables and suicide. His concept of *anomie* or strain referred to an absence of norms or a collapse of societal standards and values (Malik & Malik, 2022). According to Durkheim, people were more likely to commit suicide if they felt they were not fully integrated into society or if they were not integrated at all. Durkheim also stated that suicide is a rational choice for those persons who conclude that death is the best option for their future. Second, Merton (1938) expanded on how societal structure can lead to deviant behavior according to individuals' experience of strain. Third, the General Strain Theory (GST) proposed by Agnew in 1992 builds on earlier strain theories, offering a detailed perspective on how strain relates to deviant behavior.

Agnew (1992) expanded the concept of strain beyond mere socio-structural and economic goals to include strain from challenging circumstances and negative influences like physical and mental abuse, victimization, or discrimination. The General Strain Theory (GST) highlights the significance of strain in triggering negative emotions and the subsequent coping strategies adopted by individuals, particularly youth (Eitle, 2010). This background is instrumental in understanding the complexities surrounding adolescent suicidality. The particular stressors and challenges young people face, such as interpersonal conflicts, have all been shown to impact suicidality (Eitle, 2010). Suicide attempts are both an emotion-driven response to the incapacity to escape from stress and the unpleasant feelings that go along with it (Walls et al., 2007).

These ideas led to the Strain Theory of Suicide (STS) (Ivanich & Teasdale, 2017). STS states that "conflicting and competing pressures in an individual's life" precede suicide (Zhang & Lester, 2008, p.67). Zhang (2005) developed the Strain Theory of Suicide as a means to elucidate the socio-psychological factors that lead to instances of suicidal conduct. The STS posits that individuals may resort to suicidal behavior as a means to alleviate or eliminate psychological distress caused by pressures and conflicts arising from a misalignment between their behaviors and societal values (Zhang & Song 2006).

Adolescents exposed to various forms of victimization might experience psychological distress arising from a disconnect between their personal aspirations and ethical principles (David-Ferdon et al., 2021). Based on the General Strain Theory hypothesis of suicide, when adolescents encounter distress without adequate coping mechanisms, they may undergo coping strain. According to Merton's study conducted in 1957, these particular strains have the potential to contribute to the development of mental diseases. They may serve as precursors to other deviant actions, including suicide and other deviance. According to Zhang and Lester (2008), there exists a correlation between strain and suicide, whereby mental problems may serve as an intermediary factor that amplifies the connection between strain and suicide.

Many studies, such as those conducted by Bao et al. (2023), Islam et al. (2020), and Wilson et al. (2023), could not distinguish between the different types of bullying victimization and were unable to provide conclusive evidence on the relationships between suicidality and bullying victimization. According to Nixon (2014), there is a higher likelihood of poor mental health and suicidality among teenagers who report experiencing cyberbullying. Adolescent Behaviors and Experiences Survey (ABES) data during the pandemic indicated that 37.1% of teens in the US reported having poor mental health, 19.9% with suicide ideation, and 9.0% having attempted it in the year prior (Breuner & Bell, 2023). Most of the previous studies focused on a single additive measure were cumulative victimization subsequently increased mental health symptoms, including suicide risk.

As depicted in Figure 1, the Strain Theory of Suicide, introduced by Zhang in 2005, offers a comprehensive framework to understand the complex dynamics leading to suicidal behavior. However, the interconnection between these varied forms of victimization and their collective impact on an individual has not been thoroughly established in the existing literature. Thus, this study aimed to fill a critical gap by providing insights into how these complex victimization experiences influence overall physical wellness and mental health issues outcomes, explicitly addressing an area that has been relatively overlooked in previous research.

Figure 1

Strain Theory of Suicide by Zhang



The conceptual model diagram in Figure 1 illustrates a theoretical framework for understanding suicidal behavior by Zhang (2005). The framework posits that suicidal behavior can be the result of *strain*, which is caused by factors such as conflicting values, the gap between reality and aspirations, relative deprivation, and deficient coping mechanisms. This *strain* then interacts with social/psychological moderating factors and psychopathological intervening factors. These two sets of factors can influence the likelihood of suicidal behavior occurring. The arrows indicate the direction of influence, suggesting that while strain can lead directly to suicidal behavior, the impact is moderated or mediated by social/psychological and psychopathological factors.

Research Questions and Hypotheses

The expected model of youth suicidality in Figure 2 builds on Zhang's (2005) model. It outlines a structured pathway for understanding the relationship between different types of victimization such as physical, sexual, dating, bullying and

cyberbullying. Victimization experiences may have an effect on a person's physical and mental health, which may then influence suicidality.

Figure 2

Expected Model of Youth Suicidality



Hence, research questions:

1. What is the relationship between (a) physical victimization, (b) sexual victimization,

(c) dating violence, (d) bullying, and (e) cyberbullying on youth mental health issues

(MH Issues)?

Hypotheses:

- a. There is a positive relationship between physical victimization and MH issues.
- b. There is a positive relationship between sexual victimization and MH issues.
- c. There is a positive relationship between dating violence and MH issues.
- d. There is a positive relationship between bullying and MH issues.

- e. There is a positive relationship between cyberbullying and MH issues.
- Do experiences of (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying impact physical wellness directly or indirectly through mental health issues?

Hypothesis: There is a positive relationship between (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, (e) cyberbullying and physical wellness indirectly through mental health issues.

- What is the relationship between mental health issues and physical wellness?
 Hypothesis: There is an inverse relationship between mental health issues and physical wellness.
- What is the relationship between physical wellness and suicidality?
 Hypothesis: There is an inverse relationship between physical wellness and suicidality.
- 5. What is the relationship between mental health issues and suicidality?

Hypothesis: There is a positive relationship between mental health issues and suicidality.

6. Do victimizations relate to suicidality without intervention from mental health issues and, or physical wellness?

Hypothesis: Victimizations will have an indirect relationship with suicidality through mental health issues and, or physical well-being.

Organization of the Study

This chapter described the topic's importance, the study's purpose, the theoretical framework, and the significance of the study. Chapter II describes the literature on youth

suicidality as influenced by youth victimization, mental health, and physical wellness. Chapter III describes the research design, including the dataset utilized, measures, variables, and analytical methods employed. Chapter IV presents the study's empirical findings, providing concrete evidence to support or refute the proposed hypotheses. This chapter offers insights into the relationships between different forms of victimization, mental health issues, physical wellness, and suicidality among youth. Chapter V critically discusses these findings related to the Strain Theory of Suicide (Zhang, 2005), General Strain Theory (Agnew, 1992), and the existing literature. Chapter V also presents the study's limitations and suggestions for future research.

Summary

In this chapter, the topic's importance given recent trends in the rates of youth suicide was presented. The purpose of the study was clearly outlined, aiming to fill critical gaps in the existing literature by examining how various forms of victimization impact mental health and physical wellness, ultimately influencing youth suicidality during the COVID-19 pandemic. Furthermore, Chapter I introduced the theories, especially the Strain Theory of Suicide, laying the groundwork for understanding the complex relationships involved in suicidal behavior among adolescents.

CHAPTER II

LITERATURE REVIEW

This chapter reviews the literature and concepts employed in the current study. It describes empirical works that support the hypothesized relationships in the theoretical explanation of suicidality among high school students as a significant public health concern. It explores the roles of various forms of victimization as they relate to physical wellness, mental health, bullying, cyberbullying, and suicidality among male and female high school students. Understanding these interconnected factors is essential for developing effective prevention and intervention strategies.

Suicidality

Suicidal ideation or suicidality encompasses a wide range of thoughts and obsessions about suicide and death, making it difficult to establish a universally accepted definition (Ivey-Stephenson et al., 2020). This lack of consensus in defining suicidality poses difficulties for researchers, educators, and medical professionals, particularly when comparing study results. Despite the extensive clinical guidelines for evaluating and managing suicide risk, there is no universally accepted clinical standard for diagnosing and treating individuals at risk. This lack of standardization hinders the ability to compare research results and is frequently cited as a limitation in meta-analyses concerning suicidality (Ivey-Stephenson et al., 2020). Healthcare records usually oversimplify suicidal ideation by using a binary "yes/no" approach, even though it can range from distressing obsessions with self-annihilation to fleeting thoughts of wanting to sleep forever (Harmer et al., 2024). Im et al. (2017) attempted to predict suicidality by looking at both attempted and successful suicides and their risk factors, especially age and gender. Specific risk factors were more pronounced in teenagers than adults. The study revealed gender disparities, with more women reporting suicide ideation and more men attempting suicide. Age also matters. Mid-adolescence is when suicide ideation peaks, and adolescents who start suicidal ideation at age 15 are more likely to attempt suicide (Im et al., 2017).

It is crucial to remember that suicide is preventable (Cha et al., 2018), and adolescence provides a significant chance for intervention, possibly saving many years of life. Christine et al. (2017) stated that most people who consider or attempt suicide start thinking about it before their mid-20s. Cha et al. (2018) discussed how suicide fatalities rise mainly throughout early childhood to young adulthood. Thus, understanding how the adolescent suicide risk unfolds is necessary for early intervention (Cha et al., 2018).

Studies by Nock et al. (2008) and Wolff et al. (2017) evaluated the literature on youth suicidal ideation, attempts, and fatalities. Research into the epidemiology, mechanisms, treatment, and prevention of suicide provides a comprehensive framework for understanding, defining, and addressing this critical issue. Studies focusing on suicidal thoughts and behaviors among teenagers are prevalent, reflecting the dynamic and evolving nature of this field. Notable contributions by Cha et al. (2018), Wolff et al. (2017), and Nock et al. (2008) underlined the continuous shifts in the understanding of adolescent suicidality, highlighting the complexity and the need for ongoing research in this area.

To have an idea about the time before and after the COVID-19 pandemic situation, Sara et al. (2023) compared times before and after in terms of self-harm or

suicidal ideation presentations in emergency departments among 10- to 24-year-olds. An annual rate of 8.4% before COVID-19 increased to 19.2% annually after COVID-19, primarily due to a notable rise in presentations from adolescent girls between the ages of 13 and 17. However, males in the 10- to 24-year-old age range did not indicate a discernible increase. Remarkably, the post-COVID expansion was more noticeable in urban and socioeconomically affluent locations. The results underscore the need to comprehend the consistent rise and the novel pandemic implications, specifically impacting teenage girls during their pivotal formative years.

Experiences of Victimization and Suicidality

Research has established a link between childhood trauma and abuse and the development of suicidality. According to studies by Salmon et al. (2022), Macalli et al. (2021), and Turanovic and Siennick (2022), sexual assault victimization, physical abuse during childhood, and bullying by peers were significant risk factors for both suicidal ideation and attempts. These findings are the concept of poly victimization, where children subjected to multiple forms of victimization exhibit exacerbated symptoms of trauma, aligning with the *dose-response theory* that suggests an increased risk of suicidality with greater exposure to childhood adversity (Johns et al., 2020). Youth suicide ideation is strongly correlated with mood and anxiety disorders, and adolescents with these diagnoses are more likely to have been victimized (Cash & Bridge, 2009). Failure to address young people's mental problems, particularly internalizing disorders, may contribute to the link between victimization and suicidal thoughts (Macalli et al., 2021).

Violence against youth, encompassing intimate relationship violence, sexual violence, child abuse, neglect, and resultant suicidal behaviors, presents a multifaceted challenge (Turanovic & Siennick, 2022). These forms of violence are interconnected, often share common root causes, and can occur concurrently or at different stages of life, frequently within the same household, neighborhood, or community (Cash & Bridge, 2009). Understanding the interrelated causes of violence and implementing comprehensive measures to protect individuals and communities are critical steps in addressing the full spectrum of violence effectively.

Physical Victimization

Literature is scarce on co-occurring violent and suicidal behaviors in youth. Often, the two are related to young people's early alcohol use. Markedly, there is evidence that preventive measures may be able to lower early alcohol consumption (Swahn et al., 2013). According to a study by the CDC (2004), the students who reported attempted suicide within the previous year were almost four times more likely than those who did not to have also reported physically fighting. The correlation between physical violence and suicidal ideation was particularly prevalent among ninth-grade students.

Shaikh et al's. (2019) study highlighted that adolescents presenting specific risk factors were more susceptible to engaging in both suicidal behaviors and acts of interpersonal violence. These insights underline the importance of implementing preventative measures against suicide before students reach high school, with a focus on addressing both aggressive tendencies and suicidal thoughts. Injuries, threats, intimidation, fear and vulnerability, school absenteeism, school dropout, medical expenses, and interruption of studies have all been linked to physical fighting (Li et al., 2022). Similar patterns emerged in cases involving physical altercations and sexual violence.

For instance, Sigel et al. (2019), concluded that carrying a firearm raised the possibility of physical conflict and suicidal thoughts and actions. Based on the problem behavior theory:

Individuals exposed to concurrent physical fighting and sexual violence will have more health-risk behaviors than individuals with only one of these exposures, and they will, in turn, have a higher risk of suicidal behaviors even though the mechanisms underlying the interaction of multiple factors in the suicide pathways are complex and difficult to conceptualize. (Li et al., 2022, p. 2)

Dell et al. (2023) conducted a cross-sectional study of non-fatal firearm injuries, a form of physical victimization, among individuals 16 years of age or older. The National Electronic Injury Surveillance System- Firearm Injury Surveillance Research (NEISS-FISS) 1993–2020 data were used, even though the study included those experiencing homelessness (PEH) in the United States. They found that 82.64% of injuries among PEH were caused by assault, which was higher than the rates in the general population. While 22.73% of patients had documented substance use, this was less than would be predicted based on prior research. The researchers noted data collection issues, identified frequent missing data on contextual factors, and raised the possibility that underestimating PEH and drug use rates in gunshot accidents may result from dependence on narrative fields, where the substance misuse could lead to suicide or self-harm or an indirect relationship between variables.

Relatedly, recent school mass shootings have drawn more attention to school safety today. The study, *Safe Schools, Thriving Students: Evidence-Based Strategies for Creating Safe and Supportive Schools* by DePaoli and McCombs (2023), highlighted that, in addition to shootings, there were other physical safety risks that children must confront at schools, including robberies, physical assaults, threats, and sexual assault. These occurrences may have long-term consequences for students' welfare, such as repercussions on mental health, increased drug use, and increased risk of suicide. There was a consensus in the study that establishing a safe school environment was crucial. Still, debate persists regarding the most effective strategies to achieve this. What is clear, however, is that there is a need to provide safety in all settings, including homes and public places, where youth may be subjected to violence (DePaoli & McCombs, 2023).

Sexual Victimization

Sexual victimization in this study by Castro et al. (2019) referred to when a minor is sexually abused for the sexual gratification or financial gain of the abuser. According to the US Department of Health and Human Services, in 2016, one in nine girls and one in 53 boys under 18 years experienced sexual abuse (Banvard-Fox et al., 2020). In 2015, the estimated economic cost of child sexual abuse in the United States was \$9.3 billion. This figure encompassed expenses related to child welfare, special education, healthcare, criminal behavior, violence, suicide, and lost productivity (Letourneau et al., 2018). In addition to causing the victims' emotional pain, this kind of abuse upends family structures and engenders parental dissatisfaction (Letourneau et al., 2023).

Adolescent sexual abuse victims are more likely to have drug addiction, depression, PTSD, anxiety, poor self-esteem, somatic symptoms, concentration issues, sleeping disorders, and suicide ideation (Castro et al., 2019). Long-term effects of childhood sexual abuse include anxiety, rage, and sexual dysfunction, among other problems. Further, as people age, suicidal thoughts can become more prevalent. Older age groups have higher suicide rates (Banvard-Fox et al., 2020).

The National Comorbidity Survey (NCS) has been conducted in several iterations, starting with the original NCS in 1990-1992. This survey revealed that 78% of females and 82% of males reported sexual abuse and met the criteria for at least one lifetime psychiatric disorder, compared to 49% and 51%, respectively, who did not report. PTSD was observed in 57% of adolescents who had suffered sexual trauma (Kessler et al., 2009). They were also at risk for sexual dysfunction, unhappiness, anxiety, fear, anger, and suicidal thoughts (Mainali et al., 2023).

Martin et al. (2004) emphasized gender disparities in child sexual abuse (CSA) and suicidality. They noted that the psychological impact can last long into adulthood. Radell et al.'s (2023) longitudinal retrospective study showed that survivors of sexual abuse may experience severe psychological disruption that brings an elevated risk of suicidality. Martin et al. (2004) also found a robust correlation between CSA and suicide attempts, mainly when psychopathology plays a mediating role.

Similarly, Bikmazer et al. (2022) examined teens with a history of child sexual abuse (CSA) in terms of their suicidal behaviors and non-suicidal self-injury (NSSI) with dissociative symptoms. Dissociative symptoms were much greater among 100 adolescents involved in NSSI, even though there was no significant difference between those with and without suicide attempts. There was a relationship between increased incidence of NSSI and suicide attempts and genital contact as a form of sexual abuse. In a regression analysis, both a history of psychiatric treatment and non-suicidal self-injury (NSSI) were independently associated with suicidality. Additionally, suicide attempts and dissociative symptoms, as reported by parents, were separately connected to NSSI (Bikmazer et al., 2022).

Relatedly, Grant et al. (2022) examined a nationally representative dataset and found that students who identified as lesbian, homosexual, bisexual, or uncertain about their sexual orientation had a markedly increased risk of suicide. Drug use, sexual activity, and involvement in acts of violence or behaviors connected to violence in schools were associated with a higher risk of suicidality (Olshen et al., 2007). However, race and grade level were not important risk factors. The results highlight the significance of devising strategies to lessen the detrimental effects of sexual, violent, drug-related, and health-related behaviors on the mental health of adolescents (Grant et al., 2022).

Dating Violence Victimization

Adolescence is often the time of first romances and, for some, sexual experiences. However, negative effects, such as aggression, can be experienced in these partnerships. Adolescents are more likely than adults to be victims of intimate relationship violence. Women are particularly vulnerable to physical or sexual abuse at the hands of intimate partners or known criminals (Abrams, 2023). About 20% of female adolescents in high school reported being physically or sexually assaulted on a date. Girls between the ages of 16 and 24 had the highest risk of intimate partner violence (IPV), which was three times greater than the national average (Belshaw et al., 2012).

Suicidal thoughts are more common among victims of dating violence than in the general population (Wolford-Clevenger et al., 2016). Meanwhile, studies have

investigated the connection between dating violence and suicidal thoughts within the framework of an empirically validated suicide theory (Chu et al., 2017). According to an interpersonal-psychological perspective of suicide, suicidal thoughts are close antecedents to unfulfilled interpersonal demands, such as thwarted belongingness and perceived burdensomeness. Such interpersonal requirements may be frustrated by relationship violence, which raises the likelihood of suicidal thoughts (Chu et al., 2017; Wolford-Clevenger et al., 2016).

The co-occurrence of peer victimization and dating violence among teenagers was investigated by Smith et al. (2021). They hypothesized that depression-related beliefs and actions might be suicide factors. They found a connection between dating violence victimization of youth and peer victimization in school, using a genetically informed design with 806 twins. The results corroborated this theory by showing that common genetic vulnerabilities linked to symptoms of depression accounted for a significant fraction of the connection. To avoid being victimized by friends or romantic partners, the research emphasized how critical it is to treat early signs of depression (Smith et al., 2021).

Olshen et al. (2007), Baiden et al. (2021), and Abrams (2023) investigated the correlation between instances of dating violence in the recent past, a lifetime history of sexual assault, and suicide attempts among teenagers. Their findings indicated that among urban teenage females, the presence of a lifetime history of sexual assault did not exhibit an independent association with suicide attempts. In contrast, Baiden et al. (2021) found a substantial correlation between teenage male suicide attempts and the experience of sexual assault. The observed gender disparity is significant and justifies more inquiry.

This research also examined sexual orientation and found that adolescents who were classified as homosexual, lesbian, bisexual, or uncertain had a higher likelihood of reporting suicide attempts compared to their heterosexual counterparts (Olshen et al., 2007). Given that dating violence is a pattern of abusive, coercive, and controlling conduct in an intimate relationship, it may include physical abuse, sexual assault, and psychological hostility. Dating violence is regularly associated with adverse outcomes, such as depression and Post-Traumatic Stress Disorder (PTSD), and these have been connected to many health repercussions (e.g., headaches, back pain, and digestive issues) (Mazza et al., 2021).

Yanez-Peñúñuri et al. (2023) investigated the association between dating violence (DV) and the mental health of both victims and offenders among people aged 12 to 29 in a systematic literature analysis. It included 27 studies, 10 of which were longitudinal, published from 1981 to 2021. The results pointed to a substantial negative influence of dating violence on the mental health of adolescent victims; the most often mentioned outcome was depression, followed by anxiety, suicidal thoughts, low self-esteem, and poor emotional well-being. Indeed, other studies such as those conducted by Mazza (2021), Gasperecz et al. (2023), and Piolanti et al. (2023) over an extended period have shown links between victimization from physical, psychological, and sexual dating violence and a range of mental health consequences, such as anxiety, depression symptoms, post-traumatic stress disorder, and thoughts of suicide. They highlighted the need for developing, implementing, and evaluating preventative methods for dating violence. Gasperecz et al. (2023) also concluded that in adolescence and early adulthood,

victims of bullying, dating violence, and abuse are more likely than their peers to experience suicidality.

Bullying and Suicidality

Bullying is a well-established risk factor for suicidality among adolescents (Kim et al., 2018). Male students often experience direct physical or verbal bullying, which can lead to feelings of isolation and hopelessness (Klomek et al., 2009). In contrast, female students may encounter more relational aggression, such as social exclusion and rumors, which can have equally detrimental effects on mental health (Crick & Grotpeter, 1995).

Educational researchers and institutions, such as the US Department of Education and various bullying prevention organizations, define bullying at school as persistent acts of physical, verbal, or social hostility that occur in a learning environment, either on or off school grounds. Physical bullying includes acts such as kicking, striking, stealing, and causing property damage. Social bullying involves spreading rumors, shaming others, alienating others, and utilizing digital methods for intimidation. Verbal bullying comprises actions like name-calling, mocking, threats, and nasty insults. According to studies, bullies often behave aggressively and believe using violence as a legitimate form of communication. They could harass others in self-defense or strength because they have underlying fears. Many bullies also have impulsive behaviors (Shireen et al., 2014).

Bullying and suicidal thoughts have a complex connection that is impacted by many variables, including anxiety, depression, abuse, low self-esteem, loneliness, and poor academic achievement (Duah, 2024). The co-occurrence of these risk variables raises the possibility of suicidal thoughts. Bullying and other behavioral and emotional
stress experienced as a child may have long-lasting effects that persist into adulthood (Alavi et al., 2015).

Hong et al. (2023) investigated the connection between African American children in Chicago's Southside communities, ages 12 to 17, who experienced bullying and suicide. According to Bao et al. (2023), suicide ideation was not directly linked to bullying victimization. However, there was a positive correlation discovered between bullying victimization and emotional distress, which was linked to suicide ideation. Neighborhood factors moderated the connection between bullying victimization and suicide ideation, with emotional discomfort acting as a mediator. The results of Okobi et al. (2023) highlighted how important it is to address suicidal ideation and bullying victimization as significant issues for African American adolescents. In this context, the research emphasized cost-effective preventative and intervention techniques (Hong et al., 2023).

Approximately 20% to 56% of youth are victims or perpetrators of bullying (Hertz et al., 2013). Bullying rates vary by age, kind, and evaluation period. Middle schoolers are more inclined to bully than high schoolers. Hinduja and Patchin (2010), in their study of 2,000 middle school students, found that those who were victims of school bullying had a 1.7 times higher likelihood of having attempted suicide than those not involved in bullying. Moreover, the impact of social isolation and bullying, including cyberbullying, has been increasingly recognized as significant contributors to adolescent suicidality. Liu et al. (2023) found a strong correlation between experiences of bullying, both in-person and online, and an elevated risk of suicidal thoughts and behaviors among

adolescents, emphasizing the critical need for creating supportive and inclusive school environments.

Verbal bullying is more prevalent than physical or cyberbullying and lasts longer, even a year. Having been bullied is common among homosexual children. Bully victims are also more likely to witness family violence, experience family violence, and commit suicide (Hertz et al., 2013). There has been inconsistent reporting in the literature regarding the frequency of bullying during the COVID-19 pandemic. Studies by Gohal et al. (2023) and Turanovic and Siennick (2022) indicated that bullying was declining, except for cyberbullying. Others raised worries about a marked uptick in bullying, especially in the digital realm. The possible detrimental effects of the pandemic on children's and teenagers' well-being need immediate action and remedies (Forsberg & Thorvaldsen, 2022).

A recent study by Da et al. (2023) discussed peer interactions as a significant contributing factor to bullying. Concerns over the COVID-19 pandemic's effects on bullying have been expressed, particularly concerning how isolation brought on by the pandemic may disrupt peer connections, perhaps resulting in a decline in peer support and an increase in bullying incidences. While some research has been done during the pandemic, further comparison studies that looked at the variations in bullying prevalence before and during the COVID-19 era are needed (Da et al., 2023). Compared to their peers who are not engaged in bullying, young people who are active in bullying, whether as aggressors or victims, are more likely to experience sadness, anxiety, and suicidal thoughts and actions (Gohal et al., 2023).

The relationship between bullying and suicidal thoughts may be a more significant correlation between bullying victims and perpetrators. However, when depression is taken into account, the correlation seems to be most important for bully perpetrators (Copeland et al., 2013). Regarding bully victims, there are also conflicting findings (Brunstein- Klomek et al., 2007). At the same time, some studies found no differences in suicidal thinking or actions between bullied children and uninvolved youth, victims, or perpetrators (Espelage & Holt, 2013; Gordon, 2022). According to Tu et al. (2022), males are more likely to encounter physical victimization. At the same time, girls are more likely to experience relational victimization, which may have longer-lasting impacts.

Cyberbullying and Suicidality

The internet has a pervasive impact on people's lives, particularly youths' inclination for online social media. Therefore, the rise of digital technology has introduced cyberbullying as a unique risk factor (Hinduja & Patchin, 2010). In the United States, 20% of high school students reported experiencing bullying in person at school in 2019, while 16% claimed to have been the victim of cyberbullying the year before (NIH, 2022). Cyberbullying is defined as deliberate and persistent acts of aggression committed by individuals or groups against victims who cannot defend themselves by the CDC (CDC, 2023) and the Olweus Bullying Prevention Program. The remote nature of cyberbullying makes it difficult for those who engage in it to see the harm they do to their targets. Cyberbullying victimization rates range from 5.3% to 31.5% worldwide, according to particular figures. Among students, prevalence rates are as high as 14.6% and 17.3% (Álvarez-García et al., 2016).

Cyberbullying can be particularly damaging, as it is pervasive and difficult to escape (Hinduja & Patchin, 2010). Both male and female high school students can fall victim to cyberbullying, which has been associated with increased suicidality (Liu & Graves, 2011). Studies emphasize the need to address and understand the dynamics of cyberbullying in the context of the increasingly influential role of cyberspace in people's daily lives.

Older adolescents, usually between the ages of 15 and 17, have a higher propensity to perpetrate cyberbullying in comparison to younger adolescents (aged 10-14 years). Nonetheless, there are no appreciable variations in the prevalence of cyberbullying based on age. While younger people may attempt to address bullying via physical confrontation, older people are more likely to take their bullying online (Zhu et al., 2021).

The frequency of cyberbullying at the time of the COVID-19 epidemic was higher compared to traditional bullying (António et al., 2023). In contrast, conventional school bullying became less viable under lockdown measures. The pandemic's limitations, which curtailed in-person contact among children and adolescents, may have raised the possibility of engaging in cyberbullying and cyber victimization to new levels.

Siah et al. (2022) investigated the association between adolescent cyber victimization and suicidal behavior, with an emphasis on the mediating roles played by psychological issues and perceived social support. The results showed that being a victim of cyberbullying was associated with a risk of stress, anxiety, and sadness, which raised the possibility of suicidality. Álvarez-García et al. (2016) emphasized the protective effect of perceived social support, indicating that family and close friends' support might lessen the risk of suicidal behavior in students who have been the victims of cyber victimization.

A 2019 study of 4,972 American middle and high school students found a significant association between digital self-harm and an increased risk of suicidal ideation and attempts, suggesting the importance of health professionals and caregivers in addressing the underlying mental health issues that can lead to these behaviors (Patchin et al., 2023). Weir, (2023) noted that amidst rising concerns about the impact of social media on youth mental health there was a clear need for protective measures. Weir (2023) stressed the importance of a balanced approach to social media, recognizing its potential benefits for teen socialization while advocating for digital literacy and safe usage practices to mitigate risks from cyberbullying, exposure to harmful content, and exacerbating mental health issues.

Mantey et al. (2023) examined the relationship between digital screen time per day and suicidal thoughts in high school adolescents in the United States, taking victimization by cyberbullies into account as a mediating factor based on YRBS data (2011–2019). They found that 15.4% of students experienced cyberbullying, and 40.5% of students reported increased digital screen use. They used structural equation models to show that higher digital screen time and adverse mental health outcomes, such as depressive symptoms, suicidal thoughts, planned suicide attempts, and prior suicide attempts, were significantly mediated by cyberbullying. This mediation is consistent across a range of demographic variables. It highlights the significance of evaluating the quality of digital screen time rather than quantity when assessing its influence on adolescents' mental health (Mantey et al., 2023). Zhou et al. (2023), in a longitudinal study, investigated the connection between early adolescents' cyber-victimization and suicidal thoughts, looking at the moderating effects of mindfulness and perceived social support as well as the mediating function of despair. A year later, suicidality was predicted by cyber-victimization, with despair acting as a complete mediating factor, according to the data from 1,110 individuals. Furthermore, the predictive impact of cyber-victimization on despair was mitigated by mindfulness.

Still, the effect of hopelessness on suicidal ideation was mitigated by perceived social support. Reducing despair, cultivating mindfulness, and offering social support should lessen the harmful effects of cyber-victimization on suicidal thoughts (Eroglu et al., 2022). These are critical points for intervention plans aimed at addressing the adverse effects of teenage cyber victimization on mental health (Zhou et al., 2023). The research emphasized that educators, parents, school administrators, nurses, and other healthcare professionals must identify depression in young people early to avoid suicidality after bullying victimization (Nguyen et al., 2023).

Yosep et al. (2023) examined 12 articles to explore the effectiveness of schoolbased nursing interventions in preventing and reducing student bullying. The interventions, which included game programs, physical activities, training, and peergroup programs, encompassed strategies such as psychoeducation, empathy training, counseling, and self-management. The findings indicated that collaborative efforts between healthcare professionals and schools could significantly reduce bullying incidents. The interventions addressed the immediate effects of bullying and aimed to foster a supportive and empathetic school environment, pointing out the importance of a multidisciplinary approach in tackling this issue.

Physical Wellness and Suicidality

The role of physical wellness in suicidality among male and female high school students is complex and multifaceted (Okada et al., 2023). Physical wellness encompasses various aspects of a person's physical health, including diet, exercise, sleep, and overall well-being. While physical wellness on its own may not directly cause suicidality, it can significantly influence an adolescent's mental health and overall susceptibility to suicidal thoughts and behaviors.

A study by Grasdalsmoen et al. (2020) stated that even though regular physical activity has been shown to have several health advantages, a sizable segment of the world's population does not participate in enough physical activity. More than one in four people globally do not reach the recommended levels of physical exercise, which comprises at least 150 minutes of moderate to vigorous physical activity (MVPA) per week. Exercise not only helps with avoiding non-communicable illnesses like Type 2 diabetes and cardiovascular disease but several studies have also shown that exercise is good for mental health, especially when it comes to problems like depression (Grasdalsmoen et al., 2020; Lavie et al., 2019; Schuch & Vancampfort, 2021; Singh et al., 2023).

Mather et al. (2009), in a Canadian national study of persons 15 and older (N= 36, 984), found that obesity was positively related to suicide ideation. Regular physical activity has been linked to improved mental health, reduced stress, and enhanced mood

(Rebar et al., 2015). Physical activities can help high school students cope with academic and social pressures, potentially reducing the risk of developing suicidal thoughts.

Of course, the other aspect of physical wellness is diet. A balanced diet with adequate nutrients is essential for cognitive functioning and emotional regulation. Poor nutrition can lead to mood swings and increased vulnerability to mental health issues, including depression (Jacka et al., 2017). Both male and female students may be affected by inadequate nutrition. Sleep plays a critical role in mental health. Irregular sleep patterns or insufficient sleep can contribute to mood disorders and exacerbate stress (Walker et al., 2019). Sleep disturbances can impact both male and female high school students and increase their risk of suicidality. Good physical wellness contributes to wellbeing and self-esteem. Feeling physically healthy can enhance a person's resilience to stressors and make it easier to seek help and support when facing challenges (Lee et al., 2012).

Male high school students who maintain good physical wellness may experience improved self-esteem and mood regulation, reducing their risk of suicidality (Ning et al., 2022). Engaging in sports or physical activities can give male students a sense of belonging and camaraderie, potentially acting as a protective factor against social isolation, a risk factor for suicidality (Brown et al., 2023). The American Psychological Association (2023) stated that females may face unique body image and appearance pressures. Maintaining physical wellness in a healthy and balanced way is essential to mitigate the risk of developing eating disorders, which can contribute to suicidal thoughts (Stice, 2002). Physical wellness programs promoting body positivity and self-acceptance can particularly benefit female students. Therefore, physical wellness is a crucial aspect of overall health that can indirectly influence suicidality among male and female high school students. While it may not be a direct cause, poor physical wellness can contribute to mental health challenges and exacerbate stressors, making students more susceptible to suicidal thoughts and behaviors (Grasdalsmoen et al., 2020). Schools and communities should prioritize comprehensive well-being programs that address physical, emotional, and social aspects to promote mental health and reduce the risk of suicidality among high school students (Guthold et al., 2018).

Vancampfort et al. (2018) found that the relationship between teenage suicide thoughts and physical activity levels was complex and not well-understood. They thought that it was potentially influenced by confounding factors such as hyperactivity, which may be associated with underlying conditions like eating disorders or attentiondeficit/hyperactivity disorder. These complicate the link between exercise and mental health. Physical activity might enhance body image and lessen depression in teenagers. Cross-sectional relationships between suicide ideation and issues like depression may become less pronounced when these factors are controlled (LaRocca et al., 2023). The kind of physical exercise, along with group activities and social support are likely critical factors in lowering the risk of suicide (Ning et al., 2022).

Physical exercise may help reduce suicide ideation, according to the data currently available from intervention trials. However, it is still unclear what kind and how much physical activity is best. Pursuant to research by Ning et al. (2022), high school students in the US who engaged in physical exercise had a lower risk of suicide. Teenagers who engaged in regular physical exercise had a much-decreased likelihood of having suicidal thoughts.

Increased physical exercise is a beneficial strategy for avoiding suicidal ideation because, according to a comprehensive review, it lowers suicidal ideation in both adults and adolescents. Although the exact processes behind its putative positive benefits are still being investigated, there may be connections to better mood, sleep, and overall quality of life (Kim et al., 2019). It is hypothesized that suicidality may negatively correlate with neurobiological changes from physical exercise (Ning et al., 2022).

Gu (2022) and Ames et al. (2022) found a correlation between consistent engagement in physical exercise and decreased depressive symptoms among adolescents. This reflects the process whereby physical activity facilitates the secretion of endogenous chemicals known as endorphins, which can enhance a person's emotional state. Lubans et al. (2012) found a significant reduction in depression symptoms among teenagers who engaged in a well-organized physical exercise program. Engaging in physical exercise has been shown to assist adolescents in coping effectively with stress, which is a prominent determinant of mental health concerns that can lead some youth to contemplate suicide (Kandola et al., 2016). Further engaging in physical activities can enhance self-esteem, especially among teenagers who see advancements in physical fitness and body image (Babiss & Gangwisch, 2009).

On the contrary, Fabiano et al. (2023) noted that high-intensity physical activity in girls was associated with an increased risk of suicidality. Although a clear causal relationship between physical activity and suicidal thoughts is not clear, empirical studies have demonstrated a significant correlation between physical activity and many

34

indicators of mental well-being among adolescents (Dammeyer & Nunez, 1999). Nevertheless, the available research indicates that promoting consistent physical exercise among teenagers might positively impact adolescents' mental well-being and reduce the likelihood of suicidality.

The prevalence of obesity among adolescents has become a significant concern in recent years. Physical activity plays a pivotal role in weight management, as it helps adolescents maintain a healthy body mass index (BMI) and reduce the risk of obesity-related conditions such as Type 2 diabetes (Farpour-Lambert et al., 2018). Engaging in physical activities helps burn excess calories, build lean muscle, and regulate appetite, all contributing to maintaining a healthy weight. Physical exercise has been widely acknowledged as a fundamental aspect of a well-balanced lifestyle, providing advantages to people of all ages. The significance of consistent physical exercise among teenagers should not be underestimated (Granero-Jiménez et al., 2022). Adolescence is a pivotal stage characterized by significant physical and psychological growth. It is a suitable phase for cultivating enduring behaviors that foster overall physical and mental health (Romeo, 2013).

Iwatate et al. (2023) studied the possible confounding influence of psychosocial variables on the independent relationship between physical wellness and suicidality among adolescents in the United States. They looked at demographics, psychosocial characteristics such as feeling depressed or hopeless, drug use, bullying, and sexual abuse, and recent suicide behaviors like attempt, ideation, and plan, using the YRBS 2019 data. While the prevalence of obesity was 15.5%, the rates of suicidality were 8.90%, 18.75%, and 15.71%, respectively. Adolescents who were obese were more likely

than those who were not to try, consider, or plan suicide after controlling for psychosocial factors, indicating an independent relationship. The study recommended further investigation into these associations' causes and temporal aspects.

Vancampfort et al. (2018) concluded that physical activity guidelines could have a preventative impact on risk behaviors associated with suicide. They suggested that future research look into age- and gender-specific suggestions. Such shortcomings in the literature give the present study more significance in understanding the relationship between physical activities and suicide. It is important to note that the relationship between physical wellness and suicidality is complex, and multiple factors can interact to increase or decrease risk. Therefore, a holistic approach to well-being that considers individual differences and needs is essential.

Boone et al. (2023) studied the association between young adults and involvement in extracurricular activities (ECAs) and their health, suicidality, and suicide attempts. Data from 583 Southeastern University participants were examined in parallel mediation models. The findings indicated that perceived social support fully mediated the positive impact of extracurricular activity involvement on wellness. However, this association was not observed in the context of suicidality, and suicidal behavior. (Harmer, 2020). The findings suggested that participating in extracurricular activities may enhance college students' health by fostering social support and acknowledging the role of stress in university life (Boone et al., 2023).

Numerous physical activities include social contact which may serve as a remedy for the adverse effects of isolation and loneliness, both of which are associated with an increased likelihood of suicidal thoughts. The significance of team sports in fostering social relationships and enhancing mental well-being among teenagers was highlighted by Eime et al. (2013). Thus, a confounding impact on mental health could be the presence or absence of peer social support facilitated by the social aspects of sports engagement.

Additionally, consistent engagement in physical exercise has been shown to contribute to optimizing sleep patterns, a critical factor in maintaining mental well-being (Mahindru et al., 2023). There exists a correlation between inadequate sleep quality and a heightened likelihood of experiencing suicidal thoughts (Ehlers et al., 2023). A study by Stewart et al. (2020) revealed a positive correlation between physical exercise and improved adolescent sleep quality. However, it is crucial to acknowledge that more investigation is required to establish a definitive correlation between physical activity and the mitigation of suicidal thoughts.

Mental Health and Suicidality

Mental health is a critical determinant of suicidality, encompassing a range of emotional and psychosocial factors. Many studies have demonstrated a strong connection between poor mental health and an increased risk of suicidal ideation, attempts, and completed suicides among high school students (Berny et al., 2024; Bridge et al., 2006; Cahill et al., 2024; Lewinsohn et al., 2001; Ndetei et al., 2022; Shain et al., 2016; Yao et al., 2023). Depression and anxiety disorders are among the most prevalent mental health conditions among high school students (Merikangas et al., 2010). These conditions are consistently associated with heightened suicidality risk (Mann et al., 2005). Substance abuse, including alcohol and drug misuse, is closely linked to mental health issues and an increased risk of suicidality (Swahn et al., 2013). Substance use can exacerbate underlying mental health conditions and contribute to suicidality. In research by Kalin (2021), depression and anxiety disorders were identified as two of the most common mental health conditions affecting young individuals. Anxiety disorders generally manifest during childhood, whereas depression often emerges later, in adolescence or early adulthood (Sun et al., 2023). These conditions coexist, with pathological anxiety typically preceding the onset of depression. Among teenagers, the lifetime incidence of anxiety disorders can reach up to 32% (Shain et al., 2016), while the 12-month prevalence of severe depression in adolescents is estimated at approximately 13% (Kalin, 2021). The prevalence of these disorders is similar among boys and girls during childhood. However, during puberty and subsequent development, females are nearly twice as likely as males to be diagnosed with severe depression and anxiety (Kalin, 2021).

Previous research showed that students had significant sadness, anxiety, and distress (Dammeyer & Nunez, 1999). Most students did not misuse substances, but they did experience minor melancholy, anxiety, nervousness, tension, and loneliness. Dammeyer and Nunez (1999) found a relationship between suicidal behavior, body image, and food control issues. These results have significant implications for students' mental health. The researchers suggested that school orientations should include information on emotional and behavioral responses and relevant evaluations and treatment alternatives. Also, they suggested universal screening to identify at-risk students, those with depression, anxiety, negative emotions, and drug use for further assessments of suicidality. Accomplishing these measures will require more comprehensive care and outreach. The study also suggested that innovative methods like internet-based initiatives should be used to promote help-seeking (Dammeyer & Nunez, 1999).

In prevention planning, it is important to remember that gender affects sadness and anxiety, with women reporting greater levels than men (Center et al., 2003). Studies have shown that male high school students may be less likely to seek help for mental health problems due to societal stigma, potentially placing them at greater risk (Schwartz & Meyer, 2010). In contrast, female students may have higher rates of help-seeking but may still face unique pressures that affect their mental well-being (Nolen-Hoeksema, 2001).

Ong et al. (2021) studied the suicide of US youth given mental health status. Mental health issues in American children are increasing, and many encounter psychological traumas, a risk factor for mental health disorders and suicide. Suicide attempts are a known predictor of suicide. Many suicide victims had previously been treated for suicide attempts, highlighting the need to target this high-risk population for prevention. Overall, their study sought to identify and assess risk factors for suicide attempts in various age groups to improve age-appropriate suicide prevention programs and policies, given escalating suicide rates among children and young people with mental health issues.

Research by Brent et al. (2008), Perez et al. (2016), Kiani et al. (2021), Van Brunschot and Humphrey (2022), and Thomas (2021) indicated that youth from divorced or separated families had a higher likelihood of engaging in suicidal behavior or attempting suicide, highlighting the significant impact of family structure on mental health outcomes in adolescents. Even after controlling for demographic characteristics, suicide victims are more likely than controls to originate from a non-intact family of origin (Kiani et al., 2021). Suicide-attempting adolescents were more likely than nonattempters to have had at least three parent figure changes from age 5 to 15 (Auger et al., 2023). The weight of evidence demonstrates that whereas divorce and separation raise the likelihood of psychopathology, including suicidal behavior, in young people, death does not (Sander et al., 2020).

Parental loss by death and divorce may have diverse causes and effects (Beautrais et al., 1996; Gould, 1996). The role of familial and social factors in adolescent mental health and suicidality has gained attention in recent research. A comprehensive analysis by Patel and Smith (2024) indicated that adolescents experiencing conflict or discord within family settings, or those with a history of familial mental health disorders, were at a heightened risk for both mental health issues and suicidality.

Overall, many challenging family circumstances are associated with suicidality. (Harold & Sellers, 2018; Stansfeld et al., 2016). According to Curtin (2020), co-occurring disorders, such as drug misuse, are common in adolescents and may make diagnosis and treatment more challenging as only one concept is typically studied at a time. It is critical to lessen stigma, increase care accessibility, improve education, and create evidencebased preventive and intervention plans tailored to adolescents' needs. However, it is essential to comprehend what fosters resilience to cope with all situations or mental states.

Apart from gender, Rasberry et al. (2018) also examined different sexualities using information from the Youth Risk Behavior Surveys conducted in 2015 and 2017. This study explored the relationship between alcohol use, minority stress, and suicidal thoughts and behaviors (STBs) among Latinx, Black, and White sexual minority youth (SMY). They examined whether experiencing both minority stress and suicide attempts affects alcohol consumption differently for Latinx and Black SMY compared to White SMY, integrating Minority Stress Theory and Bagge and Sher's Theoretical Framework of the Alcohol–Suicide Attempt Relation. The findings corroborated the theories that higher alcohol use was connected to victimization and that higher alcohol consumption was linked to plans for suicide attempts. The intricate interactions of minority stressors and their varying effects on alcohol use among racial/ethnic SMY groups are highlighted by these results, underscoring the need for more studies to distinguish risk variables from protective ones (Rosales et al., 2023).

Gender differences in help-seeking behavior and societal pressures play a significant role in understanding suicidality. Male students' reluctance to seek help can hinder early intervention efforts (Canetto & Sakinofsky, 1998). Meanwhile, female students may face pressures related to body image and relationships, impacting their mental health (Wade et al., 2009). Variations in cognitive and behavioral problems might be the reason for gender disparities in suicide behavior. Male youth suicide rates may be related to higher rates of externalizing disorders, such as deviant behavior, conduct disorder, and drug addiction disorder.

Granted, men tend to select more extreme means of ending their lives (Freeman et al., 2017). On the other hand, internalizing illnesses such as anxiety and mood disorders are more common among women. Generally, the sorts of emotional and behavioral issues that males and females experience vary, with externalizing disorders being more frequent in men and internalizing disorders more common in women. These disparities in emotional and behavioral issues may account for gender differences in suicide rates (Freeman et al., 2017).

Gaylor et al. (2023) compared data from 2019 and 2021 to look at changes in high school students' suicidal thoughts and acts. During this time, there was a rise in the frequency of suicide attempts, intentions, and severe considerations among female students. Still, among male students, the overall incidence of suicidal thoughts and acts did not change. Given what is known about victimizations, mental health, physical wellness, and suicidality, this study examined cross-sectional national data for 2021 to offer insights on how well these theoretical connections were sustained during pandemic movement restrictions toward informing suicide prevention efforts.

Summary

This comprehensive literature review described the recent empirical research on the multifaceted aspects of suicidality among high school students, highlighting it as a pressing public health issue. It delved into various forms of victimization including bullying, cyberbullying and sexual and dating violence, and their profound impacts on adolescents' physical and mental health, thereby elevating the risk for suicidal thoughts and behaviors. Despite comprehensive investigations into these areas, gaps remain, especially in understanding the sophisticated interactions between physical activity levels and suicidality, the impact of cyberbullying in the digital age, and the multifaceted effects of family dynamics and societal pressures on youth mental health. This study addresses these gaps by providing a more holistic analysis of suicidality's risk factors, focusing on the understudied aspects of digital victimization and the protective role of physical wellness, specifically during the COVID-19 pandemic.

CHAPTER III

METHOD

This chapter outlines the origin of the data, the methodologies employed for gathering information, the key features of the participants, the measures, the strategies implemented to ensure the integrity of the data, and the analytical techniques applied, including multivariate regression. The focus of this investigation is on a variety of risk factors such as victimization (physical, sexual, and dating), bullying, cyberbullying, and the physical and mental health of individuals and their connection to the likelihood of suicidal behaviors among young people.

Research Design

This present study was a quantitative research design that addressed six research questions. The research questions were based on General Strain Theory (GST), which suggests that individuals engage in deviance due to experiencing strains or stressors that lead to negative emotions such as physical, sexual, and dating victimization, bullying, cyberbullying, mental health issues, and physical wellness, compelling them to commit crimes herein, suicidality as a form of coping or retaliation (Agnew, 1992). Research question one investigated the impacts of various forms of victimization, that is physical, sexual, dating violence, bullying, and cyberbullying, on youth mental health (MH), positing that these experiences are directly correlated with MH problems. It was analyzed with simple linear regression. The second research question investigated whether mental health issues mediated the adverse effects of victimization on physical wellness. This is based on the premise that victimization can lead to mental health problems, which, in turn, adversely affect physical health. The fourth and fifth questions delved into the relationship between physical wellness and suicidality and between mental health problems and suicidality. The sixth research question assessed the potential for victimization to contribute to suicidality through its effects on psychological and physical health. It hypothesized that victimization may indirectly influence suicidality by exacerbating mental health and physical wellness issues. Questions four, five, and six were analyzed using logistic regression.

Research Questions

 What is the relationship between (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying on youth mental health issues (MH issues)?

Hypotheses

- a. There is a positive relationship between physical victimization and MH issues.
- b. There is a positive relationship between sexual victimization and MH issues.
- c. There is a positive relationship between dating violence and MH issues.
- d. There is a positive relationship between bullying and MH issues.
- e. There is a positive relationship between cyberbullying and MH issues.
- 2. Do experiences of (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying impact physical wellness directly or indirectly through mental health issues?

Hypothesis: There is an inverse relationship between (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying and physical wellness indirectly through mental health issues.

- What is the relationship between mental health issues and physical wellness?
 Hypothesis: There is an inverse relationship between mental health issues and physical wellness.
- 4. What is the relationship between physical wellness and suicidality?
 Hypothesis: There is an inverse relationship between physical wellness and suicidality.
- What is the relationship between mental health issues and suicidality?
 Hypothesis: There is a positive relationship between mental health issues and suicidality.
- 6. Do victimizations relate to suicidality without intervention from mental health and, or physical wellness?

Hypothesis: Victimizations will have an indirect relationship with suicidality through mental health issues and, or physical well-being.

Data Source – Youth Risk Behavior Survey (YRBS)

This study utilized Youth Risk Behavior Survey (YRBS) 2021 data for high school students. The entire dataset however was collected by the Centers for Disease Control and Prevention (CDC) from 1991 through 2021. The survey was administered biennially to measure youth and adolescents' health-related behaviors, emerging issues, and experiences (CDC, 2023). The Department of Education and CDC collaboratively collect YRBS as a part of the Youth Risk Behavior Surveillance System (YRBSS), a cross-sectional survey to monitor the leading cause of death, disability, and addiction risk factors among high school students. The sample is a demographically diverse subset of students in Youth Risk Behavior Survey (YRBS) 2021 data for high school students throughout all 50 states of the United States and the District of Columbia. The present study data were obtained from the 2021 Youth Risk Behavior Survey (YRBS), which had a sample size of 17,232 participants. For the current study, only 7,684 cases were included because suitable modifications were applied to the variables in the initial analysis, eliminating some instances.

Data Collection

For data collection, the Centers for Disease Control and Prevention (CDC) engaged the ICF Macro, Incorporation, and Westat, two external contractors, to provide comprehensive assistance across all phases of the national Youth Risk Behavior Surveillance System (Brener et al., 2013). A survey coordinator was also designated for each YRBSS site, who worked in collaboration with the participating schools to ensure the quality execution of the survey. The survey was implemented using a three-stage cluster sampling design. The CDC used the PC Sample, a unique software tool created by the CDC and Westat that created two-stage cluster samples of schools and classes within sampled schools for every location to pick the YRBS samples effectively. Classes were chosen randomly, and schools were selected with a probability proportionate to the number of students enrolled.

Every school followed local protocols to get parental consent before administering the YRBS. The primary sampling units (PSUs) or sample frames were stratified during the first step to adequately represent Black and Hispanic students within each PSU. The selection of PSUs was based on a probability proportionate to the amount of school enrollment within each stratum. Finally, selections were made at each school consisting of one or two complete classes of a mandatory topic from each grade level, ranging from ninth to 12th grade. The survey protocols were devised to safeguard students' confidential and voluntary participation. Students completed the selfadministered questionnaire during a single class session, proctored by school staff. They immediately recorded their replies in a computer-scannable booklet or on a computerscannable answer sheet.

The YRBS focuses on six areas of prioritized health-risk behaviors, which include behaviors that are associated with injuries/violence, tobacco use, alcohol/other drug use, high-risk sexual behaviors, poor eating habits, and physical inactivity (CDC, 2016). The Survey -YRBS has 99 multiple-choice questions (CDC, 2023). Child-serving institutions employ this tool for multiple objectives, such as increasing awareness, developing programs, improving professional growth, legislation, and responses to improve youth outcomes.

Characteristics of the Data

Available YRBS data reflect youth survey responses from 1991 to 2021 for over 5 million high school students. The surveys are typically administered biennially, often in the spring semester. However, the Youth Risk Behavior Survey (YRBS) 2021 was carried out during the fall semester due to the COVID-19 pandemic.

The schools were chosen methodically, using a random starting point, with a probability proportionate to the enrollment in grades 9 through 12. A total of 209 schools were included in the sample. There was some decrease in participation in the Youth Risk

Behavior Survey (YRBS), which may be attributed to COVID-19 safety measures implemented at the schools.

Response Rates of the Youth Risk Behavior Survey (YRBS) 2021

The response rates of the YRBS were:

 School Response Rate: 152 of 209 selected schools completed the survey. This means 72.7% of schools responded. About 72.7% of schools asked consented to participate in the poll.

2. Student Response Rate: 17,508 of 21,791 selected students completed surveys. After the data editing, 7,684 cases were usable for the current study. The calculated student response rate was 79.1%. About 79.1% of sampled students submitted or had usable questionnaires following data processing.

3. Overall Response Rate: The overall response rate was 57.5%. It is calculated by multiplying the school response rate by the student response rate, which gives the survey response rate. This accounts for schools' participation and students' completion of usable surveys (79.1% student response rate) times 72.7% (school response rate) = 57.5%). This total response rate indicates data effectiveness and representativeness when assessing survey findings. The fall 2021 statistics are the first YRBS data obtained after the COVID-19 pandemic began. While many schools had returned to in-person education by then, the effects of the lockdown on families likely impacted school-related YRBS factors.

The current response is comparatively lower than in preceding years, which might be attributed to the difficulties in administering a school-based survey during the ongoing COVID-19 epidemic. Therefore, the response rates of schools and students in 2021,

48

precisely 72.7% and 79.1%, respectively, exhibited a modest decrease compared to the preceding two cycles of the Youth Risk Behavior Surveillance (YRBS).

YRBS Overall Participants

In the sample (N = 17,232), roughly half of the student population consisted of males, accounting for 51.7%. The distribution of students across different grade levels was as follows: grade 9 accounted for 26.6%, grade 10 accounted for 25.4%, grade 11 accounted for 24.3%, and grade 12 accounted for 23.5%. Concerning race and ethnicity, the majority of students, namely 50.7%, identified as White, followed by 25.4% Hispanic students; then 12.1 % African-American students, 5.7% identified as multiracial students, Asian students represented 4.9% of the student population, American Indian/Alaska Native (AI/AN) students accounted for 0.7% and, Native Hawaiian/Other Pacific Islander (NH/OPI) students constituted 0.5% of the students.

The 2021 survey also included the sexual orientation of students. It noted that 75.5% of the participants described themselves as heterosexual, 3.2% identified as gay or lesbian, 12.1% identified as bisexual, 5.2% identified as questioning, and 3.9% identified as other. Additionally, 1.8% of the participants replied, "I do not know what this question is asking." The proportion of students identifying with a sexual orientation other than heterosexual has shown a consistent upward trend, rising from 11% in 2015 to 26% in 2021. The observed rise in the proportion of LGBQ+ individuals among YRBS 2021 participants might perhaps be attributed to modifications in the survey's question phrasing, which now included students who identify as questioning, "I am uncertain about my sexual identity (questioning)," or those who express their sexual orientation using other terminology.

The majority of students (57%) said that they had not engaged in any sexual activity during their lifetime. Approximately 34.6% of students reported having sexual contact exclusively with individuals of the opposite sex. In contrast, 6.0% indicated engaging in sexual contact with both genders, and 2.4% reported having sexual contact exclusively with individuals of the same sex.

Validity and Reliability of the Data

Many studies have concluded over the years that health-related cognitive and other related variables impact self-reports. Additionally, the inclusion of measures assessing violent and suicide tendencies in surveys administered to adolescents may provoke sensitive responses. Brener et al. (2013) determined that the self-reported healthrisk behaviors, as assessed in the Youth Risk Behavior Survey (YRBS), were impacted by cognitive and environmental variables. However, these influencing factors were shown not to compromise the validity of the YRBS.

Additionally, two studies have been undertaken to examine the test-retest reliability of the national Youth Risk Behavior Surveillance (YRBS) questionnaire, one done in 1992 and the other in 2000. The first research was administering the 1991 Youth Risk Behavior Survey (YRBS) questionnaire to 1,679 children in grades 7 to 12. This process occurred on two dates within a 14-day interval between administrations (Brener et al., 1995). Approximately 75% of the 54 self-reporting questions were deemed to have a high level of reliability, with a kappa value ranging from 61% to 100% (Barrios et al., 2000).

In another study, the psychometric characteristics of the Youth Risk Behavior Survey (YRBS) questions were assessed for convergent validity, criterion validity, internal consistency, and temporal stability. The study findings revealed stronger positive associations between the use of marijuana and other substances ($\rho = 0.537$), as well as between alcohol consumption and tobacco use ($\rho = 0.418$) (Brener et al., 1995). Overall, the students consistently provided reliable reports over some time. However, additional examination of 11 specific items was recommended due to their kappa's value below 61% and showing substantial discrepancies in prevalence estimates (Brener et al., 2003). Consequently, questions that raised doubts were modified or removed from subsequent surveys (CDC, 2004). The study demonstrated criterion validity by seeing substantial variations in domain scores based on gender. The scale has satisfactory internal consistency, as shown by Cronbach's alpha coefficient of 0.770. Most domains (82%) exhibited an intraclass correlation coefficient of 0.75 or higher. In comparison, 64.1% of the items had a kappa value of 0.60 or above (Lima et al., 2020).

The suicide questions in the Youth Risk Behavior Surveillance (YRBS) survey had a strong positive correlation with one another. Additionally, these items demonstrated a substantial association with associated psychological categories such as sadness and anxiety (Barrios et al., 2000). Recently, concerning discriminant validity, the items assessing suicidality in the Youth Risk Behavior Survey (YRBS) had higher correlations with items that measured comparable actions than similar constructs (McKinnon, 2023).

The instrument exhibits criterion validity, internal consistency, and temporal stability, all demonstrating good levels (Ricks et al., 2023). For instance, there was a higher degree of correlation between items linked to suicidal thoughts in the YRBS survey and similar items in other measures, compared to the correlation observed

between YRBS items related to suicide attempts and corresponding items in different measures (Lima et al, 2020). Overall, the YRBS survey demonstrates a high level of reliability and validity when used for the assessment of health risk factors. The Youth Risk Behavior Survey (YRBS) used a weighting factor to get a sample that accurately represents the population of interest (Ricks et al., 2023).

Endogenous (Outcome) Variable- Suicidality

Suicidality means "the risk of suicide, usually indicated by suicidal ideation or intent, especially as evident in the presence of a well-elaborated suicidal plan" (VandenBos, 2007, p. 1048). The Youth Risk Behavior Survey (YRBS) provides the most extensive dataset on suicide thoughts and attempts among young individuals in the United States (May & Klonsky, 2010; Gaylor et al., 2023). YRBS data has four items related to suicidality. As seen in Table 1, the first and second items focus on suicide attempt 1. "During the past 12 months, did you ever seriously consider attempting suicide?" it is ordinal level with "yes" or "no" as categories. 2. "During the past 12 months, did you make a plan about how you would attempt suicide?" it is ordinal level with "yes" or "no" as categories. The following two items focus on suicide itself. 3. "During the past 12 months, how many times did you actually attempt suicide?", ordinal level with five categories, "0 times, 1 time, 2-3 times, 4-5 times, and 6 or more times". 4. "If you attempted suicide during the past 12 months, did any attempt result in an injury, poisoning, or overdose that had to be treated by a doctor or nurse?", ordinal level with three categories, "yes," "no" and "I did not attempt suicide during the past 12 months".

For the analysis purpose, suicide and attempted suicide responses were categorized as 0 = no and 1 = yes. Responses to the question of whether any suicide attempt resulted in medical care were classified as 1 = yes and 0 = no/did not attempt,respectively, for the number of suicide attempts (0 = 0 times, 1 = 1 time to 6 or more times). Many studies classify and group items this way (Baiden & Tadeo, 2020; Gaylor et al., 2023; Harper et al., 2023; Mantey et al., 2023).

Research by Pontes et al. (2020), Messias et al. (2014), and May and Klonsky (2010) highlighted the complexity of the term "suicidality" as a dependent variable. These studies suggest that suicide planning should be considered a component of suicidal ideation. They proposed combining both ideation and planning into a single variable. Therefore, the variable represents the preparatory actions individuals might take before attempting suicide. Additionally, the CDC (2016) defined suicide as self-directed violence, that includes the act of planning one's death.

Control Variables

The control variables were age, sex, grade, sexuality, and BMI. For age, according to the data, "How old are you?", (1. 12 years old or younger 2. 13 years old 3. 14 years old 4. 15 years old 5. 16 years old 6. 17 years old 7. 18 years old or older) are the options, ordinal level with categorical; Gender: "What is your sex?" with option 1. Female 2. Male – Nominal level with categorical variable. The following control variable is grade, "In what grade are you?" (1. 9th grade, 2. 10th grade, 3. 11th grade, and 4. 12th grade). The next is "Sexuality, which of the following best describes you?" (1 heterosexual (straight), 2. gay or lesbian, 3. bisexual, 4. I describe my sexual identity in some other way, 5. I am not sure about my sexual identity questioning, 6. I don't know what this question is asking). The BMI was calculated under the ratio level with the combination of two questions, "How tall are you without your shoes on?" and "How much do you weigh without your shoes on?" The formula to calculate the body mass index (BMI) is weight $(kg) \div height^2$ (meters).

Exogenous (Indicator) Measures

This study had four exogenous measures (victimizations), two factors (mental health issues and physical activity), and the endogenous variable of suicidality. The selected independent variables for this study were physical activities, mental health, experience of victimization (which includes physical victimization, sexual victimization, dating violence victimization, bullying, and cyberbullying) as seen in Table 1.

Table 1

| Variable Name | Measures | Level of Measurement | Category |
|--|--|----------------------|-------------------------|
| Age | How old are you? | Ordinal | Categorical |
| Gender | What is your sex? | Nominal | Categorical |
| Grades | In what grade are you? | Ordinal | Categorical |
| Sexuality | Which of the following best describes you? | Nominal | Categorical |
| BMI= weight (kg) ÷ height ² (meters) | How tall are you without your shoes on? (Note: Data are in meters | Ratio | Continuous |
| | How much do you weigh without your shoes on? (Note: Data are in kilograms.) | Ratio | Continuous |
| Suicidality | During the past 12 months, did you ever seriously consider attempting suicide? | Binary | Categorical (Yes/No) |
| | During the past 12 months, did you make a plan about how you would attempt suicide? | Binary | Categorical (Yes/No) |

Study Variables

| | During the past 12 months, how many times did you actually attempt suicide? | Ordinal | Categorical (5 categories) |
|-----------------------------------|--|---------|--------------------------------------|
| | If you attempted suicide during the past 12 months, did any attempt result in an injury, poisoning, or overdose that had to be treated by a doctor or nurse? | Ordinal | Categorical (3- did not, Yes, No) |
| Physical Activities | During the past seven days, how many days were you physically active for a total of at least 60 minutes per day? | Ordinal | Categorical |
| | During the past 7 days, on how many days did you do exercises to strengthen or tone your muscles, such as push-ups, sit-ups, or weightlifting? | Ordinal | Categorical |
| | In an average week when you are in school, on how many days do you go to physical education (PE) classes? | Ordinal | Categorical |
| Mental Health Issues | During the past 30 days, how often was your mental health not good? | Ordinal | Categorical |
| | During the COVID-19 pandemic, how often was your mental health not good? | Ordinal | Categorical |
| | During the past 12 months, did you ever feel so sad or hopeless almost every day for two weeks or more in a row that you stopped doing some usual activities? | Nominal | Categorical |
| Experience of Violent V | ictimization | | |
| Physical Fight- Victimization | During the past 12 months, how many times has someone threatened or injured you with a weapon such as a gun, knife, or club on school property? | Ordinal | Categorical |
| Sexual Victimization | Have you ever been physically forced to have sexual intercourse when you did not want to? | Nominal | Categorical |
| | During the past 12 months, how many times did anyone force you to do sexual things that you did not want to do? | Ordinal | Categorical |
| Dating Violence- Victimization | During the past 12 months, how many times did someone you were dating or | Ordinal | Categorical |

| | going out with force you to do sexual things that you did not want to do? | | |
|---------------|---|---------|-------------|
| | During the past 12 months, how many times did someone you were dating or going out with physically hurt you on purpose? | Ordinal | Categorical |
| Bullying | During the past 12 months, have you ever been bullied on school property? | Nominal | Categorical |
| Cyberbullying | During the past 12 months, have you ever been electronically bullied? | Nominal | Categorical |

Pre-Analysis

In this study the pre-analysis included data cleaning, data transformation, assumption checking, and descriptive statistics that included correcting for missing values, converting variables or adjusting their scales to meet the assumptions. It also included seeing whether the data met the assumptions to ensure the quality and integrity of the data for analysis.

Suicidality

The dependent variable, suicidality, was measured by four items: a combination of attempted suicide and suicide itself. For analysis, the variables were combined to create an index value and recoded into '0'- No and '1'- Yes—the proportion of those who had and had not experienced suicidality. Only 7,472 provided valid responses on this topic.

Of the valid responses, 5,431 individuals, accounting for 70.2% of the total sample, reported not experiencing suicidality. The missing value indicated that most of the population surveyed did not report suicidality concerns. On the other hand, 2,041 individuals, representing 26.4% of the total and 27.3% of the valid responses, indicated

that they had experienced suicidality. While smaller than the non-suicidality group, this segment still represents a significant portion of the population.

Mental Health

The variables of mental health in this study were mental health challenges, the pronounced impact of the COVID-19 pandemic, and a concerning level of sadness or hopelessness. First, 21.4% reported never facing mental health challenges, suggesting a segment of resilience or perhaps a lack of recognition of such issues. Conversely, a slight majority encountered mental health issues at varying frequencies, with 21.9% rarely, 27.6% sometimes, and a combined 29.2% most of the time or always dealing with such challenges. This distribution underscores the widespread nature of mental health concerns, revealing that more than half of the respondents experience these issues at least occasionally.

Second, the impact of the COVID-19 pandemic on students' mental health was starkly divided, with 37.1% falling into one category, likely indicating lesser impact, and a significant 62.9% into another, suggesting a greater effect. This split might reflect the diverse ways in which the pandemic has influenced individuals, from exacerbating existing mental health issues to new ones.

Lastly, a substantial 43.4% of the students reported feelings of sadness or hopelessness. The remaining 56.6% of students did not report such feelings, which could indicate either a robust mental health status or a variety of coping mechanisms and support systems in place. For analysis, the variables were combined to create an index value.

Physical Wellness/Activity

Physical wellness had a combination of physical activity levels, musclestrengthening exercise frequency, and physical education (PE) attendance among a surveyed group of 7,732 individuals. For physical activity of at least five days a week, the distribution spanned from 0 to 7 days, with 24.2% of respondents active for seven days and 15.7% reporting no activity. Regarding muscle-strengthening exercises performed on three or more days, 43.7% of the valid responses indicated they engaged in such activities, whereas 56.3% did not. PE attendance shows a wide variance, with 56.2% of respondents not attending any PE classes and 21.4% attending five days a week, indicating a significant disparity in PE participation. For analysis, the variables are combined to create an index value.

Physical Victimization

This was measured by being threatened at school. Most respondents, 93.5%, reported never being threatened at school. A small fraction experienced it one or more times, with an increase in frequency associated with a decrease in the number of reports, culminating in 0.5% reporting 12 or more instances.

Sexual Victimization

This was measured by two items: first, ever been physically forced to have sexual intercourse. Most of the surveyed population, 91.2%, reported they had never been forced into sexual intercourse, while 8.8% affirmed they had experienced this form of violence. The following measure was sexual violence; a significant majority, 88.4%, stated they had never experienced sexual violence. Reports of one-time incidents and repeated

incidents were relatively low. For analysis, the variables are combined to create an index value.

Dating Violence Victimization

This had two items: sexual and physical dating violence. Over 52% reported no instances of sexual dating violence. In comparison, 41.1% indicated they did not date. Instances of violence were less common among those who dated, with a decrease in frequency as the severity increased. Second, 54% of those who dated reported no physical violence, with 41.1% not dating at all. Reports of physical violence were fewer and decreased with increasing frequency. For analysis, the variables are combined to create an index value.

Bullied at school

84.8% reported no experiences of being bullied at school, while 15.2% acknowledged such experiences.

Cyberbullying/ Electronic bullying

Reflecting patterns similar to bullying, 84.2% of respondents reported they had not experienced electronic bullying. In comparison, 15.8% confirmed they had experienced cyberbullying.

Analyses

This study investigated the association between suicidality and its risk factors. The analysis was done with four separate regression analyses, two linear regressions, and two logistic regressions to examine the relationship between suicidality and predictors. All the analyses were performed using IBM's SPSS version 29 package. Before the analysis, the data were prepared by pre-analysis screening, including missing data, outliers, normality, linearity, and homoscedasticity. A descriptive analysis was conducted to examine the data. Listwise deletion is a simple and easier way to resolve missing data issues which can help maintain statistical power and precision of estimates. Also, listwise deletion can reduce bias in the analysis under the assumption that the missing data mechanism is completely at random (MCAR). In such cases, the deleted cases are statistically similar to the cases with complete data (Kang, 2013).

Assumptions for Regression

Simple linear regression is a statistical method used to model the relationship between variables by fitting a linear equation to observed data. Several key assumptions underpin the inferences' validity from a simple linear regression analysis. The assumptions of linear regression analysis are: 1. Linearity: the relationship between the independent and dependent variables is linear. 2. Independence: the residuals are independent, indicating that the residuals of any other observation do not influence the residuals for one observation. 3. Homoscedasticity: the variance of the error terms is constant across all levels of the independent variable. 4. Normality: the error terms are normally distributed. 5. Multicollinearity: in simple linear regression, this assumption typically pertains to the independence of the independent variable. Also, assuring that the model does not omit relevant variables or include irrelevant ones is essential.

However, unlike linear regression, logistic regression does not assume a linear relationship between the dependent and independent variables but instead models the probability that the dependent variable belongs to a particular category. This mainly includes binary outcomes. The dependent variable is binary (dichotomous), meaning it
only has two possible outcomes (yes/no, 1/0): observation independence, no perfect multicollinearity, and linear logit (Hosmer et al., 2013).

Demographic Data

A total of 7,684 cases were included in the analysis, which reflects the most complete cases given the variables in this study. The categorized student's age ranges from 12 years old or younger to 18 years old or older; of these, 3,661 (47.6%) identified as female and 3,964 (51.6%) as males, with the gender distribution presenting a neareven split, indicative of balanced gender representation in the study population. The age distribution highlighted a significant concentration of participants in the 14-17 age range.

The grade level distribution among the students was evenly spread across 9th to 12th grades. Specifically, 1,910 (24.7%) were in 9th grade, 1,918 (24.8%) in 10th grade, 1,909 (24.7%) in 11th grade, and 1,941 (25.1%) in 12th grade. In terms of sexual identity, a majority of 5,558 (71.9%) students identified as heterosexual, that is straight, indicating a significant representation of this demographic. Other main categories included 259 (3.3%) identifying as gay or lesbian and 865 (11.2%) as bisexual as shown in Table 2.

The dataset provided a rich foundation for exploring developmental, social, and educational dynamics characteristic of adolescents in the US. The balanced gender distribution, comprehensive grade, and inclusive representation of sexual identities enhance the study's relevance and applicability to a broad spectrum of adolescent experiences.

Table 2

Sample Demographics

| Demographics | Frequency | Percentage (%) |
|--------------|-----------|----------------|
| Gender | | |

| Female | 3661 | 47.3 |
|-------------------------|------|------|
| Male | 3964 | 51.3 |
| Age | | |
| 14 years old | 1384 | 17.9 |
| 15 years old | 1901 | 24.6 |
| 16 years old | 1938 | 25.1 |
| 17 years old | 1918 | 24.8 |
| Grade | | |
| 9 th grade | 1910 | 24.7 |
| 10 th grade | 1918 | 24.8 |
| 11 th grade | 1909 | 24.7 |
| 12 th grade | 1941 | 25.1 |
| Sexual identity | | |
| Heterosexual (straight) | 5558 | 71.9 |
| Gay or lesbian | 259 | 3.3 |
| Bisexual | 865 | 11.2 |

Summary

This chapter described the research design as quantitative. This included a review of the YRBS sampling and data collection process. The YRBS is a national dataset. Hence, data for 2021 were utilized to investigate how independent variables were related to suicidality. They were analyzed using simple linear regression and logistic regression techniques in IBM SPSS Version 29. The descriptive statistics of the demographic characteristics of the 7,684 student participants showed a balanced gender representation and a significant concentration of participants in the 14-17 age range. The majority identified as heterosexual, with a smaller representation of gay or lesbian and bisexual students.

CHAPTER IV

RESULTS

This chapter presents the findings from analyses conducted to explore the complex relationships between various forms of victimization, physical, sexual, dating, bullying, and cyberbullying, and their impact on the mental health and physical wellbeing of high school students, ultimately assessing their association with suicidality. Utilizing the 2021 High School Youth Risk Behavior Survey responses, the research employed ordinary least square regression and logistic regression using IBM SPSS Version 29.0 to examine these relationships.

Given the complex nature of the theoretical framework, this section methodically describes the outcomes, beginning with descriptive statistics that summarize the sample's demographic characteristics, including sex, race, age, and BMI, as reflective of the adolescent population in the United States. Correlations between all the independent and dependent variables are discussed. The correlation analysis served as a foundational step that informed model selection and specification, helped diagnose potential issues, and contributed to the overall reliability and validity of the research findings. Therefore, Pearson correlation coefficients were used in this study to measure the strength and direction of the linear relationships between pairs of variables, as can be seen in Figure 3. Significance levels were determined using a two-tailed test. Table 3 presents the Pearson correlation coefficients among observed and latent constructs in the study. Each construct is correlated with the others to assess the strength and direction of their relationships.

Next, linear regression results show the association between independent and dependent variables. Similarly, logistic regression analyses reveal the magnitude and direction of the associations between the forms of victimization studied and the mental health outcomes of adolescents. In the context of the ongoing public health concern regarding adolescent suicidality, these results offer critical insights into the risk and protective factors at play. The analysis underscores the significant impact of various victimization experiences on the mental health challenges faced by adolescents. The direct and indirect effects of victimization on suicidality contribute to a more comprehensive understanding of adolescent mental health and the pivotal role of physical and psychological well-being in safeguarding against the threat of suicide among high school students.

Figure 3



Pearson Correlation of Observed and Latent Constructs

Table 3 presents the Pearson correlation of the observed and latent constructs. A significant positive correlation was found between suicidality and combined mental health issues (r = .535, p < .001, N = 7374), indicating that as mental health issues increase, the likelihood of suicidality also increases. Conversely, physical wellness showed a small but significant negative correlation with suicidality (r = ..096, p < .001, N = 7348), suggesting that higher levels of physical wellness are associated with lower suicidality. Physical victimization was positively correlated with suicidality (r = .175, p < .001, N = 7411), sexual victimization (r = .195, p < .001, N = 7465), and dating victimization (r = .192, p < .001, N = 7599), indicating that experiences of victimization are linked to an increased risk of suicidality. Sexual victimization showed a strong positive correlation with dating victimization (r = .635, p < .001, N = 7458), highlighting a significant overlap between these types of victimization experiences.

Table 3

| Variable name | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 |
|---------------|--------|--------|-------|--------|--------|--------|--------|---|
| Suicidality | | | | | | | | |
| MH | .535** | | | | | | | |
| PW | 096** | 170*** | | | | | | |
| PV | .175** | .123** | .017 | | | | | |
| SV | .371** | .321** | 042** | .195** | | | | |
| DV | .274** | .212** | 002 | .192** | .635** | | | |
| Bullying | .246** | .247** | 033** | .239** | .254** | .208** | | |
| Cyberbullying | .257** | .261** | 040** | .198** | .291** | .238** | .485** | |

Pearson Correlation of Observed and Latent Constructs

** Correlation is significant at the 0.01 level (2-tailed).

MH- Mental Health Issues, PW- Physical Wellness, PV- Physical Victimization, SV-Sexual Victimization, DV- Dating Victimization Bullying was positively correlated with suicidality (r = .246, p < .001, N = 7463), physical victimization (r = .239, p < .001, N = 7591), and cyberbullying (r = .198, p < .001, N = 7460), emphasizing the relationship between bullying behaviors and various forms of victimization and mental health issues. Cyberbullying also showed significant positive correlations with mental health issues (r = .261, p < .001, N = 7536) and sexual victimization (r = .291, p < .001, N = 7451), indicating that online bullying is associated with both mental health challenges and experiences of sexual victimization.

These correlations show a complex relationship between mental health, physical wellness, and various forms of victimization, including bullying and cyberbullying, highlighting the need for comprehensive approaches to address these interconnected issues. Exploring the correlation among variables before the specification of a model is essential for mitigating the risk of misinterpreting relationships within the model.

Examination of Risk Factors for Suicidality Among Adolescents

Research question 1: What is the relationship between (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying on youth mental health? The first research question contains five hypotheses. These hypotheses were examined using the first regression model, which investigates the relationship between various forms of victimization and combined mental health issues (MH combined) among a sample of 7,732 individuals. A linear regression was conducted. The independent variables included in the model were Physical Victimization, Sexual Victimization Combined, Dating Victimization Combined, Bullying, and Cyberbullying. The dependent variable was MH combined. The regression model was significant, F (5, $(R^2 = .145, R^2 = .144)$, with a standard error of the estimate at 1.009.

As indicated in Table 4 the regression coefficients indicated that Sexual Victimization Combined (B = .543, SE = .030, β = .257, p < .001) had the most substantial positive association with MH combined, followed by Bullying (B = .375, SE = .038, β = .123, p < .001) and Cyberbullying (B = .365, SE = .038, β = .121, p < .001). Physical Victimization also showed a significant positive relationship with MH combined (B = .123, SE = .052, β = .027, p = .018). In contrast, Dating Victimization Combined did not significantly predict MH combined (B = -.036, SE = .041, β = -.012, p = .381).

Therefore, examining the relationship between various forms of victimization and youth mental health problems, the statistical analysis revealed significant findings. *Hypothesis 1a:* Physical victimization showed a positive relationship with mental health problems, as indicated by a B value of .123 and a significant p-value of .018, supporting the hypothesis that physical victimization is associated with increased mental health problems. Similarly, in *Hypothesis 1b*, sexual victimization had a substantial positive correlation with mental health issues, demonstrated by a high B value of .543 and a p-value of less than .001, strongly supporting the notion that sexual victimization severely affects youth mental health. Contrarily, *Hypothesis 1c*, dating violence victimization, did not show a significant positive relationship with mental health problems, evidenced by a negative B value of -.036 and a non-significant p-value of .381. This finding suggests that within this sample, dating victimization may not be a significant predictor of mental health issues, thereby not supporting the anticipated positive relationship.

Furthermore, both bullying and cyberbullying were significantly associated with mental health problems. Bullying had a positive B value of .375 with a p-value of less than .001, and cyberbullying mirrored this significance with a B value of .365 and a pvalue of less than .001. These results support the hypotheses that both bullying (Hypothesis 1d) and cyberbullying (Hypothesis 1e) are positively related to mental health problems among youth. These results suggest that sexual victimization, bullying, and cyberbullying are significant predictors of mental health issues, with sexual victimization showing the strongest association. The lack of a significant relationship for Dating Victimization Combined might indicate its lesser impact on MH within this sample or the possibility of complex interactions not captured by this model.

Table 4

| Regression | Analysis | Predicting | Mental | Health | Issues from | Victimizations, | Bullying | and |
|------------|----------|------------|--------|--------|-------------|-----------------|----------|-----|
| Cyberbully | ing | | | | | | | |

| Predictor | В | SE | β | t | р | | | |
|--|-------|------|------|---------|-------|--|--|--|
| Constant | 1.359 | .014 | | 100.171 | <.001 | | | |
| Physical Victimization | .123 | .052 | .027 | 2.357 | .018 | | | |
| Sexual Victimization Combined | .543 | .030 | .257 | 17.905 | <.001 | | | |
| Dating Victimization Combined | 036 | .041 | 012 | 877 | .381 | | | |
| Bullying | .375 | .038 | .123 | 9.771 | <.001 | | | |
| Cyberbullying | .365 | .038 | .121 | 9.561 | <.001 | | | |
| B = Unstandardized coefficients; SE = Standard Error; β = Standardized | | | | | | | | |

coefficients; N = Sample Size.

Research question 2: Do experiences of (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying impact physical

wellness directly or indirectly through mental health Issues? This research question has one hypothesis: there is a positive relationship between (a) physical victimization, (b) sexual victimization, (c) dating violence, (d) bullying, and (e) cyberbullying and physical wellness indirectly through mental health issues.

To examine the relationship between experiences of victimization and physical wellness, both directly and indirectly by mental health issues, a linear regression analysis was conducted. The hypothesis posited that these forms of victimization would be positively related to physical wellness through their indirect effects on mental health issues. Therefore, two models were constructed: one including mental health issues as a predictor (with MH Issues) and the other excluding mental health issues (without MH Issues) to discern the variance in predictive value attributed to mental health concerns.

The regression analysis results, considering mental health issues, indicated that mental health significantly mediated the relationship between victimization experiences and physical wellness. Specifically, when mental health issues were included in the model (Model 1), the analysis showed a significant prediction of physical wellness, F (6, 7153) = 39.450, p < .001, with an R² value of .032. This suggests that the model, including mental health issues, explained 3.2% of the variance in physical wellness scores.

In contrast, when mental health issues were excluded from the model (Model 2), the prediction of physical wellness was still significant, F (5, 7249) = 7.478, p < .001, but with a lower R² value of .005, indicating that only 0.1% of the variance in physical wellness scores was accounted for. This demonstrates a diminished predictive power in the absence of mental health considerations. The results are summarized in Table 5, which compares the effects of the victimization variables on physical wellness with and without considering mental health issues.

Table 5 presents findings that indicate that the presence of mental health issues significantly influenced the relationship between victimization experiences and physical wellness. For example, sexual victimization's negative impact on physical wellness becomes more pronounced when mental health issues are not considered, suggesting that mental health issues play a critical mediating role in the relationship between victimization and physical wellness.

Similar to the hypothesis, there is a positive relationship between experiences of victimization and physical wellness indirectly through mental health issues. The results support this hypothesis, demonstrating that mental health issues mediate the impact of victimization experiences on physical wellness. However, the direct effects of victimization on physical wellness, particularly in the absence of mental health considerations, highlight the complex relationship between these factors.

Table 5

Regression Analysis Predicting Physical Wellness Directly and Indirectly through Mental Health Issues

| Variable | B- With MH Issues | B- Without MH Issues | SE- With MH Issues | SE- Without MH Issues | Beta - With MH Issues | Beta - Without MH Issues | p- With MH Issues | p- Without MH Issues |
|---------------------------|-------------------------|-------------------------|--------------------------|-----------------------------|-----------------------------|--------------------------------|-------------------------|-------------------------|
| Constant | 1.952 | 1.739 | 0.020 | 0.013 | | | <.001 | <.001 |
| Mental Health Issues | 156 | - | 0.011 | - | 177 | - | <.001 | - |
| Sexual Victimization | 038 | 123 | 0.029 | 0.029 | 020 | 066 | .191 | <.001 |
| Dating Victimization | .094 | .098 | 0.038 | 0.039 | .037 | .039 | .014 | .011 |
| Physical Victimization | .139 | .123 | 0.049 | 0.049 | .035 | .031 | .005 | .013 |
| Bullying | .002 | 053 | 0.036 | 0.036 | .001 | 020 | .963 | .149 |
| Cyberbullying | 009 | 071 | 0.036 | 0.036 | 004 | 027 | .794 | .049 |

Note: $R^2 = .032$ for Model 1; $R^2 = .005$ for Model 2. SE = Standard Error. Beta = Standardized Coefficients.

Research question 3: What is the relationship between mental health problems and physical wellness? The one hypothesis is that there is an inverse relationship between mental health issues and physical wellness.

For this research question, a linear regression analysis was performed to investigate the influence of mental health issues (MH combined) on physical wellness (PW Combined). The analysis indicated that mental health issues significantly predicted physical wellness, F (1, 7300) = 218.431, p < .001. Specifically, there was a .150 unit decrease in physical wellness scores for each unit and an increase in mental health issues, controlling for other factors. This model accounted for 2.9% of the variance in physical wellness scores, as indicated by an R² value of .029. As shown in Table 6 these results suggest a modest but significant negative relationship between mental health issues and physical wellness.

Table 6

Regression Analysis for Mental Health Issues Predicting Physical Wellness

| Predictor | В | SE | β | t | р |
|-------------|-------|------|-----|---------|--------|
| Constant | 1.952 | .020 | | 99.686 | <.001 |
| MH combined | 150 | .010 | 170 | -14.779 | < .001 |

B = Unstandardized coefficient; SE = Standard Error; β = Standardized coefficient; R² = .029; Adjusted R² = .029.

This result supports the hypothesis that mental health issues and physical wellness are inversely related, suggesting that individuals with higher levels of mental health problems tend to have lower levels of physical fitness. These findings underscore the interconnectedness of mental and physical health, highlighting the importance of addressing mental health issues as part of holistic health and wellness interventions.

Research question 4: What is the relationship between physical wellness and suicidality? The hypothesis: There is an inverse relationship between physical wellness and suicidality and *Research question 5:* What is the relationship between mental health problems and suicidality? The hypothesis: There is a positive relationship between mental health problems and suicidality combined. To answer these, a logistic regression analysis was conducted to examine the impact of mental health (MH combined) and physical wellness (Physical Wellness Combined) on the likelihood of suicidality. Therefore, the dependent variable, suicidality, was recorded as binary (No = 0, Yes = 1) and MH combined. Physical Wellness Combined (indexed) were treated as categorical variables, with the following coding for physical wellness: 0 (reference), 1, 2, 3; and for MH combined: 0 (reference), 1, 2, 3.

As shown in Table 7 the logistic regression model was significant, χ^2 (6) = 2391.559, p < .001, indicating that the predictors significantly explained the variance in suicidality. The model accounted for between 28.1% and 40.6% of the variance in suicidality, as indicated by Cox & Snell and Nagelkerke R Square values, respectively. The model's classification accuracy was 80.5%, with 85.9% for predicting non-suicidality (No) and 66.0% for predicting suicidality (Yes). Overall accuracy of 72.6%. The significant predictors included MH combined levels 1 (B = 1.902, SE = 0.237, p < .001, Exp(B) = 6.700), 2 (B = 3.328, SE = 0.232, p < .001, Exp(B) = 27.885), and 3 (B = 4.797, SE = 0.230, p < .001, Exp(B) = 121.100). For Physical Wellness Combined, only

level 1 was a significant predictor (B = -.233, SE = 0.103, p = .024, Exp(B) = 0.792), indicating a decrease in the odds of suicidality.

The logistic regression analysis revealed that higher levels of mental health issues significantly increase the likelihood of suicidality, with particularly high odds for those at the highest level of mental health concern (MH combined level 3). Conversely, participants with better physical wellness (level 1) showed a slight reduction in the likelihood of suicidality.

These findings underscore the importance of addressing mental health issues as a critical factor in suicidality prevention efforts. Further research should explore how physical wellness influences suicidality and the potential for integrated mental and physical health interventions.

Table 7

| Variable | В | SE | Wald | df | p | Exp(B) |
|----------------------------|--------|-------|---------|----|-------|---------|
| MH combined | | | | | | |
| MH Issues 1 | 1.902 | 0.237 | 64.266 | 1 | <.001 | 6.700 |
| MH Issues 2 | 3.328 | 0.232 | 205.240 | 1 | <.001 | 27.885 |
| MH Issues 3 | 4.797 | 0.230 | 433.128 | 1 | <.001 | 121.100 |
| Physical Wellness Combined | | | | | | |
| Physical Wellness 1 | -0.233 | 0.103 | 5.087 | 1 | .024 | 0.792 |
| Physical Wellness 2 | -0.126 | 0.103 | 1.510 | 1 | .219 | 0.881 |
| Physical Wellness 3 | -0.168 | 0.112 | 2.242 | 1 | .134 | 0.845 |
| Constant | -4.072 | 0.241 | 285.762 | 1 | <.001 | 0.017 |

Logistic Regression Analysis Predicting Suicidality

 χ^2 (6) = 2391.559, p < .001; The reference category for MH combined is 0, and for Physical Wellness Combined is 0.

Research question 6: Do victimizations relate to suicidality without intervention from mental health and, or physical wellness? This had one hypothesis: Victimizations will have an indirect relationship with suicidality through mental health and, or physical well-being.

The findings are from a logistic regression analysis examining the relationship between various forms of victimization, that is physical, sexual, dating, bullying, and cyberbullying, and suicidality, which reflects the hypothesis that victimizations have an indirect relationship with suicidality mediated by mental health and physical well-being. After accounting for missing cases (n = 403, 5.2%), the analysis included 7,329 participants. This model also used a binary-coded (No = 0, Yes = 1) dependent variable. Independent variables consisted of physical victimization, sexual victimization combined, dating victimization combined, bullying, and cyberbullying, each with specific coding but categorical and binary variables.

The logistic regression model was statistically significant, χ^2 (7) = 1232.268, p < .001, suggesting that the predictors distinguished between participants who reported suicidality and those who did not. The model explained between 15.5% and 22.5% of the variance in suicidality, as evidenced by the Cox and Snell and Nagelkerke R Square values, respectively. The Hosmer and Lemeshow test yielded a chi-square of 15.447 with two degrees of freedom, indicating a significant fit (p < .001), which suggests the model's good fit with the observed data. The overall prediction accuracy of the model is 77.9%, with 94.9% for predicting non-suicidality and 31.6% for predicting suicidality. This represents a considerable improvement in predictive ability compared to the baseline model, which had a 73.0% overall prediction accuracy.

As shown in Table 8 significant predictors of suicidality included physical victimization (B = .679, SE = .118, p < .001, Exp(B) = 1.973). This indicated that individuals reporting one or more instances of physical victimization were about twice as likely to report suicidality compared to those with no such experiences. Both categories of sexual victimization were significant predictors, with the first category showing an odds ratio of 4.565 (B = 1.518, SE = .092, p < .001) and the second, odds ratio of 6.117 (B = 1.811, SE = .155, p < .001), suggesting increasing likelihoods of suicidality with more severe experiences of sexual victimization. Dating victimization also appeared as a significant factor, with the first category associated with a 1.487 increase in the odds of suicidality (B = .397, SE = .125, p = .002) and the second category related to a 1.634 increase (B = .491, SE = .229, p = .032). Bullying (B = .664, SE = .085, p < .001, Exp(B) = 1.942) and cyberbullying (B = .641, SE = .084, p < .001, Exp(B) = 1.898) were both significant predictors, indicating nearly double the odds of suicidality for victims. Therefore, the above findings emphasize the significant impact of various forms of victimization on the likelihood of suicidality.

Table 8

Logistic Regression Analysis Predicting Suicidality Based on Forms of Victimization

| Variable | В | SE | Wald | df | р | Exp(B) | | | | |
|----------------------|----------------------|------|----------|----|--------|--------|--|--|--|--|
| Physical | .679 | .118 | 33.406 | 1 | <.001 | 1.973 | | | | |
| Victimization | | | | | | | | | | |
| Sexual Victimization | Sexual Victimization | | | | | | | | | |
| SV1 | 1.518 | .092 | 272.387 | 1 | < .001 | 4.565 | | | | |
| SV2 | 1.811 | .155 | 136.652 | 1 | <.001 | 6.117 | | | | |
| Dating Victimization | | | | | | | | | | |
| DV1 | .397 | .125 | 10.067 | 1 | .002 | 1.487 | | | | |
| DV2 | .491 | .229 | 4.582 | 1 | .032 | 1.634 | | | | |
| Bullying | .664 | .085 | 61.495 | 1 | <.001 | 1.942 | | | | |
| Cyberbullying | .641 | .084 | 58.868 | 1 | <.001 | 1.898 | | | | |
| Constant | -1.614 | .036 | 2025.359 | 1 | < .001 | .199 | | | | |

 χ^2 (7) = 1232.268, p <.001; The reference category for each categorical variable is its absence or the baseline level (e.g., No physical victimization, No sexual victimization, No dating victimization, Not bullied, Not cyberbullied).

The results indicate that various forms of victimization are significantly associated with increased odds of suicidality, supporting the hypothesis of a relationship between victimization and suicidality. Specifically, sexual victimization shows the strongest association with suicidality, significantly increasing the odds, followed by physical victimization, bullying, and cyberbullying, with dating victimization showing a significant but weaker relationship.

Summary

The multivariate regression analyses explained relationships between variables. The first regression model highlighted sexual victimization, bullying, and cyberbullying as significant predictors of mental health issues. A subsequent model demonstrated a significant negative relationship between mental health issues and physical wellness. Further, the logistic regression analyses examined the impact of mental health and physical wellness on suicidality. It highlighted the importance of addressing mental health issues and the interconnections between victimization, mental health, and physical well-being in efforts to prevent suicidality among high school students.

CHAPTER V

DISCUSSION

This chapter discusses the summary of the significant relationships shown through the study between various forms of victimization and suicidality, examining the roles of mental health and physical wellness. It presents the theoretical guidance of these findings, given the strengths and limitations of strain theory to understand suicidality among adolescents. Additionally, this chapter delves into the nuanced effects of the pandemic on these dynamics, considering how social isolation, increased online interactions, and disruptions in routine may have exacerbated or altered traditional risk pathways. Furthermore, the discussion extends to the practical implications of these findings for policymakers, educators, mental health professionals, and communities. It offers recommendations for interventions that prioritize holistic approaches to adolescent mental health, incorporating strategies to mitigate victimization and promote physical wellness as preventative measures against suicidality. Lastly, the chapter identifies gaps in the current research, proposing avenues for future studies to explore other social contexts in understanding and preventing suicidality among adolescents, particularly post-pandemic.

Summary of Findings

The study analyzed data from 7,684 high school students, showing a genderbalanced demographic primarily aged 14-17, with most identifying as heterosexual and smaller groups identifying as gay, lesbian, or bisexual. Correlation analysis found significant links between suicidality, mental health issues, physical wellness, and various forms of victimization, particularly noting strong associations between suicidality and mental health and between different victimization types.

Multivariate regression analyses identified sexual victimization, bullying, and cyberbullying as key predictors of mental health problems and highlighted a significant negative relationship between mental health issues and physical wellness. Logistic regression analysis further explored how mental health and physical wellness impacted suicidality, emphasizing the serious role of addressing mental health and understanding the connections between victimization, mental health, and physical well-being in preventing suicidality among adolescents. Mental health emerged as a critical mediator, suggesting that it could mitigate the impact of victimization on suicidality. This study was conducted to answer six research questions:

- What is the relationship between a) physical victimization, b) sexual victimization, c) dating violence, d) bullying, and e) cyberbullying on youth mental health?
- 2. Do experiences of a) physical victimization, b) sexual victimization, c) dating violence, d) bullying, and e) cyberbullying impact physical wellness directly or indirectly through mental health?
- 3. What is the relationship between mental health problems and physical wellness?
- 4. What is the relationship between physical wellness and suicidality?
- 5. What is the relationship between mental health problems and suicidality?
- 6. Do victimizations relate to suicidality without intervention from mental health and, or physical wellness?

Contributions to the Literature

The results are consistent with previous research indicating that adolescents experiencing these forms of victimization are at a heightened risk of suicidality. This consistency with prior research emphasizes the urgent requirement for interventions tailored to reduce these forms of victimization. For instance, studies by Labuhn et al. (2021) and Franklin et al. (2017) previously highlighted the correlation between victimization experiences and increased suicidality, emphasizing the need for comprehensive strategies to mitigate these risks.

Further, the research by Espelage et al. (2019) demonstrated that the experience of dating violence specifically contributes to a variety of mental health issues, including depressive symptoms and suicidal ideation, thereby indicating the complex interplay between interpersonal violence and mental health outcomes. Additionally, Exner-Cortens et al. (2013) found that adolescents involved in violent dating relationships were significantly more likely to report suicidal thoughts and behaviors, suggesting that the trauma associated with such victimization had profound effects on adolescent wellbeing. Turner et al. (2016) explored the mechanisms through which physical victimization, particularly in school environments, exacerbated feelings of isolation and helplessness, contributing further to the risk of suicide.

The findings from Mitchell et al. (2017) revealed that sexual victimization, due to its highly traumatic nature, significantly increased the likelihood of developing severe mental health conditions that could lead to suicidality. Together, these studies highlighted the impact of physical and sexual victimization and dating violence on adolescent mental health and suicidality. They collectively underline the importance of developing targeted, evidence-based interventions that not only aim to prevent such victimization but also support the mental health of survivors, addressing the root causes and consequences of these experiences to reduce adolescent suicidality.

Apart from the influence of victimization, the demographic variables were also consistent with the literature. This study suggested that gender and sexual identity were significant factors in suicidality, with females and presumably heterosexual individuals showing lower odds of suicidality compared to their counterparts. According to Letourneau et al. (2018) these were gender differences in suicidality, with males showing a higher likelihood of engaging in serious suicide attempts. Another longitudinal study found gender-specific protective factors, with secure attachment and higher family function being protective for only females (Bakken et al., 2024). The present study found that females were less likely to experience suicidality compared to males, with an odds ratio indicating that suicidality is approximately 26.5% lower for females. These highlight the need for gender-focused strategies in both understanding and preventing suicide which can be more impactful.

Furthermore, in the present study, the collective assessment of sexual identity categories indicated a significant relationship with suicidality. When examined separately the categories did not demonstrate a statistically significant impact on suicidality. However, the heterosexual category had a B coefficient of .033 and an odds ratio (Exp(B)) of 1.034 with a Sig. of .942, indicating that this category had a very weak and not statistically significant association with suicidality. Many studies consistently show that individuals from sexual minority groups, for example, gay, lesbian, bisexual, often report higher rates of suicidality and mental health issues compared to their

heterosexual counterparts (Haas et al., 2011; Hong et al., 2023; LaRocca et al., 2022; Parchem et al., 2024). Future research could benefit from integrating these findings with qualitative insights to understand the underlying mechanisms and develop targeted interventions.

Discussing the influence of bullying and cyberbullying on suicidality among adolescents reflects Nixon's (2014) findings on the evolving domain of adolescent victimization, where cyberbullying emerges as a contemporary channel to suicidality. Kowalski and Limber (2013) highlighted the psychological consequences of cyberbullying, noting its inevitable nature, contributing to an increased sense of vulnerability and isolation among victims. This phenomenon was further explored in Hinduja and Patchin's research (2010), where the anonymity afforded by digital platforms exacerbated the intensity of bullying incidents, making cyberbullying a potent risk factor for adverse mental health outcomes, including suicidality.

Another longitudinal study by Gini and Espelage (2014) provided compelling evidence of the causal link between bullying victimization and subsequent increases in suicidal ideation and attempts, emphasizing the long-lasting effects of such experiences. Similarly, Selkie et al. (2016) stressed the unique challenges posed by cyberbullying, including the difficulty in escaping online harassment and the often, insufficient adult intervention, which collectively intensified the distress experienced by victims. The research by Copeland et al. (2013) further supported these findings by demonstrating the strong associations between bullying involvement, as either a victim or perpetrator, and later suicidal thoughts and behaviors, pointing to the complex dynamics of bullying interactions and their profound impact on adolescent well-being. These studies suggest a need for comprehensive anti-bullying programs that address both traditional and digital forms of bullying. The implementation of effective strategies to reduce the impacts of bullying and cyberbullying, include promoting digital literacy, supportive school environments, and ensuring accessible mental health resources for affected adolescents. Further regarding the physical wellness connection to suicidality directly and indirectly, the findings are supported here as was found to be the case by Richardson et al. (2013). They found that regular physical activity was associated with reduced depressive symptoms and enhanced mood among adolescents, which could indirectly lower the risk of suicidal ideation and attempts. Furthermore, a systematic review by Mammen and Faulkner (2013) corroborated the protective effects of physical activity against depression, a major risk factor for suicidality, in both adults and adolescents. This suggests that the relationship is often indirectly associated with suicidality.

On the other hand, mental health had a direct relationship with suicidality. Studies by Klonsky et al. (2016), Nock et al. (2008), and the present study highlighted the significant role of mental health issues as strong predictors of suicidality. This is consistent with earlier research by Gould et al. (2003), who emphasized the need for early detection and intervention for mental health disorders to prevent suicidal behavior in adolescents. The significance of accessible mental health services King et al. (2009) suggested that school-based mental health programs can significantly reduce suicide risk factors among students.

Additionally, research by Czyz et al. (2013) pointed to the efficacy of crisis intervention strategies, such as safety planning and follow-up care, in mitigating suicide

risk among adolescents presenting with suicidal ideation or attempts. These studies also clearly suggest mental health care, incorporating early screening, preventive interventions, and crisis management to address the high prevalence of suicidality, especially among adolescents. Together, these initiatives should aim at community and individual levels to foster a supportive environment for youth mental health and physical well-being.

Support for Strain Theories

The findings of this study provide strong empirical support for the strain theory of suicide (STS) and general strain theory (GST), suggesting that victimization and the consequential strain play a significant role in precipitating suicidality among adolescents. This complex relationship is influenced by mental health. Agnew's (1992) GST noted that relationships like mistreatment or victimization can lead to negative emotions, which, if not adequately coped with, can result in deviant behavior, including suicidality.

The findings highlight the impact of victimization, physical, sexual, and through bullying, on suicidality, supporting this theory's concept. Victimization acts as a source of strain and facilitates a sequence of negative emotional states, resulting in suicidal ideation or attempts. Aseltine et al. (2000) argued that stressful life events, strains, significantly predict increases in depressive symptoms and suicidal ideation among adolescents, aligning with the findings on the mediating role of mental health issues.

Additionally, as discussed in Chapter 1, a study by Zhang and Lester (2008) viewed suicide as a response to the psychological strain arising from two or more conflicting social pressures, including the inability to achieve goals, the removal of

positive stimuli, and the presentation of negative stimuli. The significant correlations the study observed between various forms of victimization and suicidality show this theoretical intention, including the mediating effect of mental health in the relationship between strain and suicidality. This highlights the complexity outlined in the STS and GST that demonstrates how these experiences serve as potent negative stimuli of psychological strain. According to the literature, interventions focusing on resilience-building, mental health education, and accessible psychological services can mitigate the impact of strain on mental health and, consequently, on suicidality (King et al., 2009; Klomek et al., 2011).

This study contributes to understanding the risk factors for suicidality among adolescents. Future research should continue to explore the complex interactions between victimization, mental health, physical wellness, and suicidality, with a focus on developing interventions that address these risk factors. It remains unclear exactly how physical wellness influences suicidality. For example, there is no consensus in the literature regarding whether the impact of physical wellness is more social or biological.

Comparing 2019 and 2021 Data

Studies such as Hemberg et al. (2024), Essler et al. (2024), Orban et al. (2024) mentioned the significant gaps in research concerning the mental health outcomes and long-term effects of the COVID-19 pandemic on adolescents and young adults. However, considering adolescence, it is a crucial transition period from childhood to adulthood. These are characterized by several developmental milestones that propose unique challenges and opportunities for growth. Therefore, even the short-term effects are important to understand. Regarding whether there was increased suicidality post-pandemic this section analyzes data from the Youth Risk Behavior Survey (YRBS) from the years 2019 and 2021 using 2019 as a baseline. For the purpose of comparing the Youth Risk Behavior Survey (YRBS) data from the years 2019 and 2021, it was necessary to include only those variables that are common to both datasets. Given that the 2019 YRBS dataset lacks some of the variables present in the 2021 dataset.

Table 9

Comparison of Logistic Regression Coefficients for YRBS 2019 and 2021

| Variables | B- 2019 | B- 2021 | SE- | SE- | n- 2019 | n- 2021 |
|----------------------|----------------|---------|-------|-------|---------|---------|
| , and its | D 2 017 | D 2021 | 2019 | 2021 | p 2019 | p 2021 |
| Mental Health Issues | 2.369 | 2.393 | 0.072 | 0.073 | <.000 | <.001 |
| Sexual Victimization | 0.597 | 0.759 | 0.070 | 0.073 | <.000 | <.001 |
| Dating Victimization | 0.155 | 0.291 | 0.097 | 0.102 | .112 | .004 |
| Physical | 0.590 | 0.550 | 0.110 | 0.120 | < 000 | < 001 |
| Victimization | 0.589 | 0.559 | 0.118 | 0.129 | <.000 | < .001 |
| Bullying | 0.590 | 0.419 | 0.083 | 0.092 | <.000 | <.001 |
| Cyberbullying | 0.263 | 0.383 | 0.091 | 0.091 | <.004 | <.001 |
| | | | | | | |

Note: $R^2 = 0.390$ for 2019; $R^2 = 0.424$ for 2021; B = Unstandardized coefficient; SE = Standard Error.

The comparison between the YRBS 2019 and 2021 models reveals that the higher R² value in 2021 data. Notably, the coefficient for mental health issues has a slight increase from 2.369 in 2019 to 2.393 in 2021. There was also a significant rise in the coefficient for sexual victimization, escalating from 0.597 in 2019 to 0.759 in 2021.

Furthermore, dating victimization, which increased from 0.155 in 2019 to 0.291 in 2021; concurrently, its p-value dramatically changed from .112 (non-significant) in 2019 to .004 (significant) in 2021, indicating a more pronounced influence of COVID – 19 or any extraneous factor in 2021 survey. The physical victimization and bullying suggested a weakened impact on suicidality, whereas cyberbullying rose from 0.263 in 2019 to 0.383 in 2021, with its significance level strengthening the relationship from p < .004 to p < .001.

Studies have indicated that the COVID-19 pandemic led to elevated rates of suicide attempts and suicidal ideation among adolescents, with contributing stressors including missed special events, financial difficulties, familial conflicts, and alterations in living conditions (Brausch et al., 2023; Thompson et al., 2021; Zhu et al., 2023; Grzejszczak et al., 2024). These elements intensified distress and mental health concerns, apparently leading to increased suicidal thoughts. Becker et al. (2024) discussed how the excessive use of digital media and smartphones was connected to negative outcomes like poor mental health and cyberbullying. During the COVID-19 pandemic, the increase in screen time correlated with heightened psychiatric emergencies among children and adolescents. While physical interactions were reduced due to social distancing measures, there was an increase in online interactions where much of the dating and social engagement shifted to digital platforms (Taquet et al., 2021). This transition may have led to new forms of victimization, such as cyber sexual harassment, which could have overlapped with cyberbullying as this it is not always captured by traditional measures of sexual and dating victimization but could contribute to the observed increases.

Limitations of the Study

This study's limitations include a reliance on self-reported data in the Youth Risk Behavior Survey (YRBS) which has the potential for recall and social desirability bias. Further, the YRBS employs a nationally representative sample of high school students in the United States. Its generalizability may be limited by factors such as non-response bias and the exclusion of adolescents not attending school. National data were incomplete as Minnesota, Oregon, and Washington do not participate. In 2021, 45 states, 1 US territory, 1 tribal government, and 28 school districts received funding through a cooperative agreement with the Division of Adolescent and School Health, part of the National Center for HIV, Viral Hepatitis, STD, and TB Prevention, to conduct a Youth Risk Behavior Survey (YRBS).

Additionally, cultural, socioeconomic, and geographical diversity within the sample might obscure the subtle connections between risk factors and suicidality. Furthermore, the cross-sectional design of the YRBS limits the ability to infer causality between the identified risk factors and suicidality. Also, the design restricts understanding of the temporal dynamics between victimization, mental health, physical wellness, and their cumulative impact on suicidality over time. Nonetheless, this study's findings are meaningful in indicating that sexual victimization is more impactful than other forms and that both traditional and cyberbullying can be particularly harmful.

Future Research and Recommendations

First, future studies can prioritize longitudinal designs to explore the temporal relationships between risk factors and suicidality. Such studies can provide insights into the causality and evolution of suicidal ideation and behaviors in response to victimization, mental health challenges, and changes in physical wellness over time. Second, a qualitative study can give a valuable perspective on the individual experiences of victimization and mental health challenges. In-depth interviews and focus groups can uncover unforeseen understandings of linking these experiences to suicidality, informing more targeted and effective interventions.

Third, identifying risk factors for suicidality is essential for understanding protective factors toward developing comprehensive prevention strategies. Future studies should explore factors such as resilience, social support, and access to mental health services that may moderate the impact of risk factors on suicidality among adolescents. Fourth, the effectiveness of interventions targeting the identified risk factors for suicidality would also advance this literature. Fifth, experimental and quasiexperimental studies can evaluate the impact of bullying prevention programs, mental health interventions, and initiatives promoting physical wellness on reducing suicidality among adolescents.

Finally, given the role of cyberbullying as a modern pathway to suicidality, future research may explore the evolving setting of digital interactions among adolescents via the Metaverse and other artificial intelligence platforms. Those studies might examine the impact of social media use, online behaviors, and digital literacy on suicidality. Understanding how exactly physical wellness may mediate mental health issues is also necessary.

Conclusion

The complex interplay between victimization, mental health, physical wellness, and suicidality highlights the need for multidisciplinary approaches to understanding and addressing suicidality among adolescents. It highlights the urgency of collective efforts by policymakers, educators, mental health professionals, and the community at large to safeguard the well-being of youth. By embracing a holistic approach to prevention and intervention, there is a hopeful prospect for significantly reducing the incidence of suicidality among adolescents. These findings suggest a need for urgent actions to make a tangible difference in the lives of young individuals through dedicated research, policy, and practice. These suggestions align with aspiring to a future where the specter of suicidality is no longer a prevalent threat to the well-being and potential of adolescents.

REFERENCES

Abrams, Z. (2023). Up to 19% of teens experience dating violence. Psychologists want to break the cycle. *Monitor on Psychology*, *54*(7) 62.

https://www.apa.org/monitor/2023/10/disrupting-teen-dating-violence

- Abrutyn, S., & Mueller, A. S. (2014). Are suicidal behaviors contagious in adolescence?
 Using longitudinal data to examine suicide suggestion. *American Sociological Review*, 79(2), 211–227. https://doi.org/10.1177/0003122413519445
- Agnew, R. (1992). Foundation for a general strain theory of crime and delinquency. *Criminology*, *30*(1), 47–88. https://doi.org/10.1111/j.1745-9125.1992.tb01093.x.
- Aknin, L. B., De Neve, J. E., Dunn, E. W., Fancourt, D. E., Goldberg, E., Helliwell, J.
 F., & Ben Amor, Y. (2022). Mental health during the first year of the COVID-19 pandemic: A review and recommendations for moving forward. *Perspectives on Psychological Science*, 17(4), 915-936.
- Alavi, N., Reshetukha, T., & Prost, E. (2015). Bullying, including cyber bullying increases the risk of suicidal behavior. *European Psychiatry*, 30(209). https://doi.org/10.1016/s0924-9338(15)30169-3
- Álvarez-García, D., García, T., Barreiro-Collazo, A., Dobarro, A., & Antúnez, Á.
 (2016). Parenting style dimensions as predictors of adolescent antisocial behavior. *Frontiers in Psychology*, 7 (1383). https://doi.org/10.3389/fpsyg.2016.01383
- American Psychological Association. (2023). *Health advisory on social media use in adolescence*. https://www.apa.org/topics/social-media-internet/health-advisory-adolescent-social-media-use

- Ames, M. E., Robillard, C. L., Ryan, J. E., Merrin, G. J., & Turner, B. J. (2023).
 Reciprocal associations between physical activity, physical self-concept, somatic symptoms, and depression from adolescence to young adulthood: Disaggregating within-and between-person effects. *Mental Health and Physical Activity*, 24, 100513.
- Ames, M. E., Robillard, C. L., Turner, B. J., Garcia-Barrera, M., Rush, J., & Craig, S. G. (2022). Associations between physical activity, affect regulation difficulties, and mental health among Canadian adolescents at two different points of the COVID-19 pandemic. *Psychology and Health*, October 3, 1-17.
- António, Guerra, & Moleiro. (2023). Cyberbullying during COVID-19 lockdowns:
 Prevalence, predictors, and outcomes for youth. *Current Psychology (New Brunswick, N.J.)*, 1–17. https://doi.org/10.1007/s12144-023-04394-7
- Arango, A., Opperman, K. J., Gipson, P. Y., & King, C. A. (2016). Suicidal ideation and suicide attempts among youth who report bully victimization, bully perpetration, and/or low social connectedness. *Journal of Adolescence*, *51*(1), 19–29. https://doi.org/10.1016/j.adolescence.2016.05.003
- Arat, G., & Wong, P. W. C. (2017). The relationship between physical activity and mental health among adolescents in six middle-income countries: A crosssectional study. *Child and Youth Services*, 38(3), 180-195.
- Aseltine Jr, R. H., Gore, S., & Gordon, J. (2000). Life stress, anger and anxiety, and delinquency: An empirical test of general strain theory. *Journal of Health and Social Behavior*, 41(3) 256-275.

- Auger, N., Low, N., Chadi, N., Israël, M., Steiger, H., Lewin, A., Ayoub, A., Healy-Profitós, J., & Luu, T. M. (2023). Suicide attempts in children aged 10-14 years during the first year of the covid-19 pandemic. *The Journal of Adolescent Health: Official Publication of the Society for Adolescent Medicine*, 72(6), 899–905. https://doi.org/10.1016/j.jadohealth.2023.01.019
- Babiss, L. A., & Gangwisch, J. E. (2009). Sports participation as a protective factor against depression and suicidal ideation in adolescents as mediated by selfesteem and social support. *Journal of Developmental and Behavioral Pediatrics*, 30(5), 376-384.
- Baiden, P., & Tadeo, S. K. (2020). Investigating the association between bullying victimization and suicidal ideation among adolescents: Evidence from the 2017 youth risk behavior survey. *Child Abuse and Neglect, 102*, 104417. https://doi.org/10.1016/j.chiabu.2020.104417
- Baiden, P., Panisch, L. S., Kim, Y. J., LaBrenz, C. A., Kim, Y., & Onyeaka, H. K.
 (2021). Association between first sexual intercourse and sexual violence victimization, symptoms of depression, and suicidal behaviors among adolescents in the United States: Findings from 2017 and 2019 national youth risk behavior survey. *International Journal of Environmental Research and Public Health*, 18(15), 7922. https://doi.org/10.3390/ijerph18157922
- Bakken, V., Lydersen, S., Skokauskas, N., Sund, A. M., & Kaasbøll, J. (2024).
 Protective factors for suicidal ideation: a prospective study from adolescence to adulthood. *European Child and Adolescent Psychiatry*, *February* 1-11.

Banvard-Fox, C., Linger, M., Paulson, D. J., Cottrell, L., & Davidov, D. M. (2020).
Sexual assault in adolescents. Primary care: Clinics in office practice; *Elsevier BV*. https://doi.org/10.1016/j.pop.2020.02.010

Bao, W., Qian, Y., Fei, W., Tian, S., Geng, Y., Wang, S., Pan, C. W., Zhao, C. H., & Zhang, T. (2023). Bullying victimization and suicide attempts among adolescents in 41 low- and middle-income countries: Roles of sleep deprivation and body mass. *Frontiers in Public Health*, *11*, 1064731. https://doi.org/10.3389/fpubh.2023.1064731.

- Barrios, L. C., Everett, S. A., Simon, T. R., & Brener, N. D. (2000). Suicide ideation among US college students associations with other injury risk behaviors. *Journal* of American College Health, 48(5), 229-233.
- Beautrais, A. L., Joyce, P. R., & Mulder, R. T. (1996). Risk factors for serious suicide attempts among youths aged 13 through 24 years. *Journal of the American Academy of Child and Adolescent Psychiatry*, 35(9), 1174–1182. https://doi.org/10.1097/00004583-199609000-00015
- Becker, T. D., Leong, A., Shanker, P., Martin, D., Staudenmaier, P., Lynch, S., & Rice, T. R. (2024). Digital media-related problems contributing to psychiatric hospitalizations among children and adolescents before and after the onset of the COVID-19 pandemic. *Child Psychiatry and Human Development*, https://doi.org/10.1007/s10578-024-01670-w
- Bell, I. H., Marx, W., Nguyen, K., Grace, S., Gleeson, J., & Alvarez-Jimenez, M.(2022). The effect of psychological treatment on repetitive negative thinking in youth depression and anxiety: a meta-analysis and meta-regression.

Psychological Medicine, *53*(1), 6–16.

https://doi.org/10.1017/s0033291722003373

- Belshaw, S. H., Siddique, J. A., Tanner, J., & Osho, G. S. (2012). The relationship
 between dating violence and suicidal behaviors in a national sample of
 adolescents. *Violence and Victims*, 27(4), 580–591. https://doi.org/10.1891/08866708.27.4.580
- Berny, L. M., Mojekwu, F., Nichols, L. M., & Tanner-Smith, E. E. (2024). Investigating the interplay between mental health conditions and social connectedness on suicide risk: Findings from a clinical sample of adolescents. *Child Psychiatry* and Human Development, February 3, 1-13.
- Bersia, M., Koumantakis, E., Berchialla, P., Charrier, L., Ricotti, A., Grimaldi, P.,
 Dalmasso, P., & Comoretto, R. I. (2022). Suicide spectrum among young people
 during the COVID-19 pandemic: A systematic review and meta-analysis. *E Clinical Medicine*, 54, 101705. https://doi.org/10.1016/j.eclinm.2022.101705
- Bikmazer, A., Koyuncu, Z., Kavruk Erdim, N., Kadak, M. T., Tarakcioglu, M. C.,
 Gokler, E., & Ozer, O. A. (2023). Association of dissociation with suicide
 attempt and non-suicidal self-injury in adolescents with a history of sexual abuse. *Psychiatry*, 86(1), 17-28.
- Bilsen, J. (2018). Suicide and youth: Risk factors. Frontiers in Psychiatry, 9. https://doi.org/10.3389/fpsyt.2018.00540
- Blakemore, S. J., & Mills, K. L. (2014). Is adolescence a sensitive period for sociocultural processing? *Annual Review of Psychology*, 65(1), 187–207. https://doi.org/10.1146/annurev-psych-010213-115202
- Boone, S., Schuler, K. R., Basu, N., & Smith, P. N. (2023). College extracurricular involvement as a suicide prevention and wellness promotion strategy: Exploring the roles of social support and meaning. *Journal of American College Health*, 71(3), 677-685.
- Brausch, A.M., Whitfield, M. & Clapham, R.B. (2023). Comparisons of mental health symptoms, treatment access, and self-harm behaviors in rural adolescents before and during the COVID-19 pandemic. *European Child Adolescent Psychiatry*, 32, 1051–1060 https://doi.org/10.1007/s00787-022-02039-x
- Brener, N. D., Billy, J. O., & Grady, W. R. (2003). Assessment of factors affecting the validity of self-reported health-risk behavior among adolescents: evidence from scientific literature. *Journal of Adolescent Health*, 33(6), 436-457.
- Brener, N. D., Collins, J. L., Kann, L., Warren, C. W., & Williams, B. I. (1995).
 Reliability of the youth risk behavior survey questionnaire. *American Journal of Epidemiology*, 141(6), 575-580.
- Brener, N. D., Kann, L., Shanklin, S., Kinchen, S., Eaton, D. K., Hawkins, J., & Flint, K.
 H. (2013). Methodology of the youth risk behavior surveillance system.
 Morbidity and Mortality Weekly Report: Recommendations and Reports, 62(1), 1-20.
- Brent, D. A., & Melhem, N. (2008). Familial transmission of suicidal behavior. *Psychiatric Clinics of North America*, 31(2), 157-177.
- Breuner, C. C., & Bell, D. L. (2023). Adolescent mental and behavioral health: COVID-19 exacerbation of a prevailing crisis. *Pediatrics*, 151(Supplement 1), e2022057267D. https://doi.org/10.1542/peds.2022-057267D

- Bridge, J. A., Goldstein, T. R., & Brent, D. A. (2006). Adolescent suicide and suicidal behavior. *Journal of Child Psychology and Psychiatry*, 47(3-4), 372-394.
- Bridge, J. A., Ruch, D. A., Sheftall, A. H., Hahm, H. C., O'Keefe, V. M., Fontanella, C.
 A., & Horowitz, L. M. (2023). Youth suicide during the first year of the COVID-19 pandemic. *Pediatrics*, 151(3).

Brodsky, B. S., & Stanley, B. (2008). Adverse childhood experiences and suicidal behavior. *Psychiatric Clinics of North America*, 31(2), 223–235. https://doi.org/10.1016/j.psc.2008.02.002

- Brown, B. A., Rottenberg, J., & Goodman, F. R. (2023). Social anxiety and interpersonal risk for suicidal ideation: A longitudinal daily diary analysis. *Suicide and Life-Threatening Behavior*, 53(6), 968-980.
- Brunstein Klomek, A., Marrocco, F., Kleinman, M., Schonfeld, I. S., & Gould, M. S.
 (2007). Bullying, depression, and suicidality in adolescents. *Journal of the American Academy of Child & Adolescent Psychiatry*, 46(1), 40–49. https://doi.org/10.1097/01.chi.0000242237.84925.18
- Cahill, M., Illback, R., & Peiper, N. (2024). Perceived racial discrimination,
 psychological distress, and suicidal behavior in adolescence: Secondary analysis
 of cross-sectional data from a statewide youth survey. *Healthcare (Basel, Switzerland)*, *12*(10), 1011. https://doi.org/10.3390/healthcare12101011
- Canetto, S. S., & Sakinofsky, I. (1998). The gender paradox in suicide. *Suicide and Life-Threatening Behavior*, 28(1), 1-23.

- Casey, B. J., Jones, R. M., & Somerville, L. H. (2011). Braking and accelerating of the adolescent brain. *Journal of Research on Adolescence*, 21(1), 21–33. https://doi.org/10.1111/j.1532-7795.2010.00712.x
- Cash, S. J., & Bridge, J. A. (2009). Epidemiology of youth suicide and suicidal behavior. *Current Opinion in Pediatrics*, 21(5), 613-619.
- Caspersen, C. J., Powell, K. E., & Christenson, G. M. (1985). Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. *Public Health Reports*, 100(2), 126.
- Castro, A., Ibáñez, J., Maté, B., Esteban, J., & Barrada, J. R. (2019). Childhood sexual abuse, sexual behavior, and revictimization in adolescence and youth: A minireview. *Frontiers in Psychology*. https://doi.org/10.3389/fpsyg.2019.02018
- CDC (2004). Suicide attempts and physical fighting among high school students— United States, 2001. *JAMA*, 292(4), 428. https://doi.org/10.1001/jama.292.4.428
- CDC (2023). YRBSS Overview Suicide Prevention Resource for Action | DASH | CDC.

https://www.cdc.gov/healthyyouth/data/yrbs/overview.htm#what.

- Center, C., Davis, M., Detre, T., Ford, D. E., Hansbrough, W., Hendin, H., Laszlo, J.,
 Litts, D. A., Mann, J., Mansky, P. A., Michels, R., Miles, S. H., Proujansky, R.,
 Reynolds III, C. F., & Silverman, M. M. (2003). Confronting depression and
 suicide in physicians. *JAMA*, 289(23), 3161.
 https://doi.org/10.1001/jama.289.23.3161
- Iobst, S. E., Breman, R. B., Walker, M., Wysong, G., Best, N., & Edmonds, J. K. (2023). Challenges, job satisfiers, and self-care among perinatal nurses in the

United States during the COVID-19 pandemic. *The American Journal of Maternal Child Nursing*, *48*(3), 118–126. https://doi.org/10.1097/NMC.00000000000912

- Cha, C. B., Franz, P. J., M. Guzmán, E., Glenn, C. R., Kleiman, E. M., & Nock, M. K. (2018). Annual Research Review: Suicide among youth epidemiology, (potential) etiology, and treatment. *Journal of Child Psychology and Psychiatry and Allied Disciplines*, 59(4), 460-482. https://doi.org/10.1111/jcpp.12831
- Christine, B., Franz, P., Guzmán, E. M., Glenn, C. R., Kleiman, E. M., & Nock, M. K. (2017). Suicide among youth – epidemiology, (potential) etiology, and treatment. *Journal of Child Psychology and Psychiatry; Wiley-Blackwell*. https://doi.org/10.1111/jcpp.12831
- Chu, C., Buchman-Schmitt, J. M., Stanley, I. H., Hom, M. A., Tucker, R. P., Hagan, C. R., Rogers, M. L., Podlogar, M. C., Chiurliza, B., Ringer, F. B., Michaels, M. S., Patros, C. H. G., & Joiner, T. E. (2017). The interpersonal theory of suicide: A systematic review and meta-analysis of a decade of cross-national research. *Psychological Bulletin, 143* (12) 1313-1345. https://doi.org/10.1037/bul0000123
- Copeland WE, Wolke D, Angold A, & Costello E.J. (2013). Adult psychiatric outcomes of bullying and being bullied by peers in childhood and adolescence. *JAMA Psychiatry*. 70(4):419–426. doi:10.1001/jamapsychiatry.2013.504
- Crick, N. R., & Grotpeter, J. K. (1995). Relational aggression, gender, and socialpsychological adjustment. *Child Development*, 66(3), 710-722.
- Curtin, S. C. (2020). State suicide rates among adolescents and young adults aged 10-24: United States, 2000-2018. *National Center for Health Statistics, Centers for*

https://www.cdc.gov/nchs/data/nvsr/nvsr69/nvsr-69-11-508.pdf.

- Czyz, E. K., Horwitz, A. G., Eisenberg, D., Kramer, A., & King, C. A. (2013). Selfreported barriers to professional help-seeking among college students at elevated risk for suicide. *Journal of American College Health*, 61(7), 398-406.
- Da, Q., Huang, J., Peng, Z., Chen, Y., & Li, L. (2023). Did the prevalence of traditional school bullying increase after COVID-19? Evidence from a two-stage crosssectional study before and during COVID-19 pandemic. *Child Abuse and Neglect*, 143, 106256. https://doi.org/10.1016/j.chiabu.2023.106256
- Dammeyer, M. M., & Nunez, N. (1999). Anxiety and depression among law students:
 Current knowledge and future directions. *Law and Human Behavior*, 23(1), 55-73.
- David-Ferdon, C., Clayton, H. B., Dahlberg, L. L., Simon, T. R., Holland, K. M.,
 Brener, N., Matjasko, J. L., D'Inverno, A. S., Robin, L., & Gervin, D. (2021).
 Vital signs: prevalence of multiple forms of violence and increased health risk
 behaviors and conditions among youths United States, 2019. MMWR.
 Morbidity and Mortality Weekly Report, 70(5), 167–173.
 https://doi.org/10.15585/mmwr.mm7005a4.
- Dell, N. A., Vaughn, M. G., & Salas-Wright, C. P. (2023). Firearm injury among people experiencing homelessness: Cross-sectional evidence from a national survey of United States emergency departments. *Public Health in Practice*, 6, 100446.

- DePaoli, J., & McCombs, J. (2023). Safe schools, thriving students: What we know about creating safe and supportive schools. *Learning Policy Institute*. https://doi.org/10.54300/701.445.
- Dsouza, D. D., Quadros, S., Hyderabadwala, Z. J., & Mamun, M. A. (2020). Aggregated COVID-19 suicide incidences in India: Fear of COVID-19 infection is the prominent causative factor. *Psychiatry Research*, *290*, 113145.
- Duah, E. (2024). The mediating effect of loneliness on the relationship between bullying victimization and suicidal behavior among adolescents in Ghana. *Youth 4*(231)-243. https://doi.org/10.3390/youth4010016
- Dyrbye, L. N., Thomas, M. R., & Shanafelt, T. D. (2006). Systematic review of depression, anxiety, and other indicators of psychological distress among US and Canadian medical students. *Academic Medicine*, 81(4), 354–373. https://doi.org/10.1097/00001888-200604000-00009
- Ehlers, C. L., Karriker-Jaffe, K. J., & Bernert, R. (2023). Poor self-reported sleep quality associated with suicide risk in a community sample of American Indian adults. *Sleep Advances: A Journal of the Sleep Research Society*, 4(1). https://doi.org/10.1093/sleepadvances/zpad024
- Ehlman, D. C., Yard, E., Stone, D. M., Jones, C. M., & Mack, K. A. (2022). Changes in suicide rates United States, 2019 and 2020. *MMWR. Morbidity and Mortality Weekly Report*, 71(8), 306–312. https://doi.org/10.15585/mmwr.mm7108a5
- Eime, R. M., Young, J. A., Harvey, J. T., Charity, M. J., & Payne, W. R. (2013). A systematic review of the psychological and social benefits of participation in sport for children and adolescents: Informing development of a conceptual model

of health through sport. *International Journal of Behavioral Nutrition and Physical Activity*, 10, 1-21.

- Eitle, D. (2010). General strain theory, persistence, and desistance among young adult males. *Journal of Criminal Justice*, *38*(6), 1113-1121.
- Eroglu, Y., Peker, A., & Cengiz, S. (2022). Cyber victimization and well-being in adolescents: The sequential mediation role of forgiveness and coping with cyberbullying. *Frontiers in Psychology*, 13, 819049.
- Espelage, D. L., & Holt, M. K. (2013). Suicidal ideation and school bullying experiences after controlling for depression and delinquency. *Journal of Adolescent Health*, 53(1), S27–S31.

https://doi.org/10.1016/j.jadohealth.2012.09.017

- Essler, S., Christner, N. & Paulus, M. (2024). Short-term and long-term effects of the COVID-19 pandemic on child psychological well-being: A four-wave longitudinal study. *Europe Child Adolescent Psychiatry 33*, 909–922. https://doi.org/10.1007/s00787-023-02215-7
- Exner-Cortens, D., Eckenrode, J., & Rothman, E. (2013). Longitudinal associations between teen dating violence victimization and adverse health outcomes. *Pediatrics*, 131(1), 71-78.

Fabiano, N., Gupta, A., Fiedorowicz, J. G., Firth, J., Stubbs, B., Vancampfort, D., & Solmi, M. (2023). The effect of exercise on suicidal ideation and behaviors: A systematic review and meta-analysis of randomized controlled trials. *Journal of Affective Disorders*, 330, 355-366.

- Fan, Y., Chen, J., Shirkey, G., John, R., Wu, S. R., Park, H., & Shao, C. (2016). Applications of structural equation modeling (SEM) in ecological studies: An updated review. *Ecological Processes*, 5(1). https://doi.org/10.1186/s13717-016-0063-3
- Farpour-Lambert, N. J., Ells, L. J., Martinez de Tejada, B., & Scott, C. (2018). Obesity and weight gain in pregnancy and postpartum: an evidence review of lifestyle interventions to inform maternal and child health policies. *Frontiers in Endocrinology*, 9, 392112.
- Farrokhi, M., Taheri, F., Bayat, Z., Damiri, M., Farrokhi, M., & et al. (2024). Role of lifestyle medicine in the prevention and treatment of diseases. *Kindle*. http://preferpub.org/index.php/kindle/article/view/Book35
- Favril, L., Yu, R., Hawton, K., & Fazel, S. (2020). Risk factors for self-harm in prison: a systematic review and meta-analysis. *The Lancet Psychiatry*, 7(8), 682-691.
- Fedina, L., King, C., DeVylder, J., & Herrenkohl, T. I. (2023). Distinct profiles of violence victimization and suicide risk: Findings from a national survey of emerging adults. *American Journal of Orthopsychiatry*, 93(3), 245–255. https://doi.org/10.1037/ort0000675
- Fegert, J. M., Vitiello, B., Plener, P. L., & Clemens, V. (2020). Challenges and burden of the Coronavirus 2019 (COVID-19) pandemic for child and adolescent mental health: a narrative review to highlight clinical and research needs in the acute phase and the long return to normality. *Child and Adolescent Psychiatry and Mental Health*, 14(1). https://doi.org/10.1186/s13034-020-00329-3

- Forsberg, J. T., & Thorvaldsen, S. (2022). The severe impact of the COVID-19 pandemic on bullying victimization, mental health indicators and quality of life. *Scientific Reports*, 12(1). https://doi.org/10.1038/s41598-022-27274-9
- Foulkes, L., & Blakemore, S. J. (2016). Is there heightened sensitivity to social reward in adolescence? *Current Opinion in Neurobiology*, 40, 81–85. https://doi.org/10.1016/j.conb.2016.06.016
- Franklin, J. C., Ribeiro, J. D., Fox, K. R., Bentley, K. H., Kleiman, E. M., Huang, X., Musacchio, K. M., Jaroszewski, A. C., Chang, B. P., & Nock, M. K. (2017). Risk factors for suicidal thoughts and behaviors: A meta-analysis of 50 years of research. *Psychological Bulletin*, 143(2), 187–232. https://doi.org/10.1037/bul0000084
- Freeman, A., Mergl, R., Kohls, E., Székely, A., Gusmao, R., Arensman, E., & Rummel-Kluge, C. (2017). A cross-national study on gender differences in suicide intent. *BMC Psychiatry*, 17, 1-11.
- Fuller-Thomson, E., Baird, S. L., Dhrodia, R., & Brennenstuhl, S. (2016). The association between adverse childhood experiences (ACEs) and suicide attempts in a population-based study. *Child: Care, Health, and Development*, 42(5), 725– 734. https://doi.org/10.1111/cch.12351
- Garcia-Williams, A. G., Moffitt, L., & Kaslow, N. J. (2014). Mental health and suicidal behavior among graduate students. *Academic Psychiatry*, 38(5), 554–560. https://doi.org/10.1007/s40596-014-0041-y

- Gasperecz, J. W. S., Baumler, E., Wood, L., & Temple, J. R. (2023). Suicidal ideation and psychological dating violence victimization—A short report. *Frontiers in Psychiatry*, 14. https://doi.org/10.3389/fpsyt.2023.1105654
- Gaylor, E. M., Krause, K. H., Welder, L. E., Cooper, A. C., Ashley, C., Mack, K. A., Crosby, A. E., Trinh, E., Ivey-Stephenson, A. Z., & Whittle, L. (2023). Suicidal thoughts and behaviors among high school students — Youth risk behavior survey, United States, 2021. MMWR Supplements, 72(1), 45–54. https://doi.org/10.15585/mmwr.su7201a6
- Gini, G., & Espelage, D. L. (2014). Peer victimization, cyberbullying, and suicide risk in children and adolescents. *JAMA Pediatrics*, *168*(5), 435–442. https://doi.org/10.1001/jama.2014.3212
- Gohal, G., Alqassim, A., Eltyeb, E. (2023). Prevalence and related risks of cyberbullying and its effects on adolescents. *BMC Psychiatry 23*(39). https://doi.org/10.1186/s12888-023-04542-0
- Gordon, S. (2022). How strong is the link between bullying and suicide? *Very Well Family*. https://www.verywellfamily.com/how-strong-is-the-link-betweenbullying-and-suicide-460620
- Gould, M. S. (1996). Psychosocial risk factors of child and adolescent completed suicide. Archives of General Psychiatry, 53(12), 1155. https://doi.org/10.1001/archpsyc.1996.01830120095016
- Gould, M. S., Greenberg, T. E. D., Velting, D. M., & Shaffer, D. (2003). Youth suicide risk and preventive interventions: A review of the past 10 years. *Journal of the American Academy of Child and Adolescent Psychiatry*, 42(4), 386-405.

- Granero-Jiménez, J., López-Rodríguez, M. M., Dobarrio-Sanz, I., & Cortés-Rodríguez,
 A. E. (2022). Influence of physical exercise on psychological well-being of
 young adults: A quantitative study. *International Journal of Environmental Research and Public Health*, 19(7), 4282.
 https://doi.org/10.3390/ijerph19074282
- Grant, R., Amos, N., Lyons, A., McNair, R., Power, J., Carman, M., & Bourne, A.
 (2023). Out in suburbia: Associations between residential location, mental health, and community connectedness among LGBTQ Australians. *Social and Cultural Geography*, 1-19.
- Grasdalsmoen, M., Eriksen, H. R., Lønning, K. J., & Sivertsen, B. (2020). Physical exercise, mental health problems, and suicide attempts in university students. *BMC Psychiatry*, 20(1). https://doi.org/10.1186/s12888-020-02583-3
- Gruber, J., Prinstein, M. J., Clark, L. A., Rottenberg, J., Abramowitz, J. S., Albano, A. M., Aldao, A., Borelli, J. L., Chung, T., Davila, J., Forbes, E. E., Gee, D. G., Hall, G. C. N., Hallion, L. S., Hinshaw, S. P., Hofmann, S. G., Hollon, S. D., Joormann, J., Kazdin, A. E., . . . Weinstock, L. M. (2021). Mental health and clinical psychological science in the time of COVID-19: Challenges, opportunities, and a call to action. *American Psychologist*, *76*(3), 409–426. https://doi.org/10.1037/amp0000707
- Grzejszczak, J., Strzelecki, D., Gabryelska, A., & Kotlicka-Antczak, M. (2024).
 Evaluation of COVID-19 effect on mental health, self-harm, and suicidal behaviors in children and adolescents population. *Journal of Clinical Medicine*, *13*(3), 744.

- Gu J. (2022). Physical activity and depression in adolescents: Evidence from China family panel studies. *Behavioral Sciences (Basel, Switzerland*), 12(3) 71. https://doi.org/10.3390/bs12030071
- Gunzler, D. D., Chen, T., Wu, P., & Zhang, H. (2013). Introduction to mediation analysis with structural equation modeling. *PubMed; National Institutes of Health*. https://doi.org/10.3969/j.issn.1002-0829.2013.06.009
- Guthold, R., Stevens, G. A., Riley, L. M., & Bull, F. C. (2018). Worldwide trends in insufficient physical activity from 2001 to 2016: a pooled analysis of 358 population-based surveys with 1.9 million participants. *The Lancet Global Health*, 6(10), e1077-e1086. https://doi.org/10.1016/s2214-109x(18)30357-7
- Haas, A. P., Eliason, M., Mays, V. M., Mathy, R. M., Cochran, S. D., D'Augelli, A.
 R., Silverman, M. M., Fisher, P. W., Hughes, T., Rosario, M., Russell, S.
 T., Malley, E., Reed, J., Litts, D. A., Haller, E., Sell, R.
 L., Remafedi, G., Bradford, J., Beautrais, A. L., ... & Clayton, P.
 J. (2011). Suicide and suicide risk in lesbian, gay, bisexual, and transgender populations: review and recommendations. *Journal of Homosexuality*, 58, 10-51.

https://doi.org/10.1080/00918369.2011.534038

Harmer, B. Lee, S. Rizvi, A., Saadabadi, A. (2024). *Suicidal ideation. April 20. Jan–.* PMID: 33351435. https://pubmed.ncbi.nlm.nih.gov/33351435/

Harold, G. T., & Sellers, R. (2018). Annual research review: Interparental conflict and youth psychopathology: An evidence reviews and practice focused update. *Journal of Child Psychology and Psychiatry*, 59(4), 374–402.
https://doi.org/10.1111/jcpp.12893

- Harper, C. R., Li, J., Sheats, K., Hertz, M. F., Merrill-Francis, M., Friar, N. W., Ashley, C. L., Shanklin, S., Barbero, C., Gaylor, E. M., & Hoots, B. E. (2023).
 Witnessing community violence, gun carrying, and associations with substance use and suicide risk among high school students Youth risk behavior survey, United States, 2021. *MMWR Supplements*, 72(1), 22–28. https://doi.org/10.15585/mmwr.su7201a3
- Hawton, K., Casañas Comabella, C., Haw, C., & Saunders, K. (2013). Risk factors for suicide in individuals with depression: A systematic review. *Journal of Affective Disorders*, 147(1–3), 17–28. https://doi.org/10.1016/j.jad.2013.01.004
- Hawton, K., Saunders, K. E., & O'Connor, R. C. (2012). Self-harm and suicide in adolescents. *The Lancet*, 379(9834), 2373–2382. https://doi.org/10.1016/s0140-6736(12)60322-5
- Hemberg, J., Sundqvist, A., Korzhina, Y., Östman, L., Gylfe, S., Gädda, F., Nyman-Kurkiala, P. (2024). Being young in times of uncertainty and isolation:
 adolescents' experiences of well-being, health, and loneliness during the
 COVID-19 pandemic. *International Journal of Adolescence and Youth*, 29(1).
 https://doi.org/10.1080/02673843.2024.2302102
- Henry, K. L., Lovegrove, P. J., Steger, M. F., Chen, P. Y., Cigularov, K. P., & Tomazic,
 R. G. (2013). The potential role of meaning in life in the relationship between
 bullying victimization and suicidal ideation. *Journal of Youth and Adolescence;*Springer Science. https://doi.org/10.1007/s10964-013-9960-2
- Hertz, M. F., Donato, I., & Wright, J. (2013). Bullying and suicide: A public health approach. *The Journal of Adolescent Health: Official Publication of the Society*

for Adolescent Medicine, 53, S1–S3.

https://doi.org/10.1016/j.jadohealth.2013.05.002

Hinduja, S., & Patchin, J. W. (2010). Bullying, cyberbullying, and suicide. Archives of Suicide Research, 14(3), 206–221.

https://doi.org/10.1080/13811118.2010.494133

- Hinduja, S., & Patchin, J. W. (2018). Connecting adolescent suicide to the severity of bullying and cyberbullying. *Journal of School Violence*, 18(3), 333–346. https://doi.org/10.1080/15388220.2018.1492417
- Holmes, E. A., O'Connor, R. C., Perry, V. H., Tracey, I., Wessely, S., Arseneault, L., Ballard, C., Christensen, H., Cohen Silver, R., Everall, I., Ford, T., John, A., Kabir, T., King, K., Madan, I., Michie, S., Przybylski, A. K., Shafran, R., Sweeney, A., Bullmore, E. (2020). Multidisciplinary research priorities for the COVID-19 pandemic: A call for action for mental health science. *The Lancet Psychiatry*, *7*(6), 547–560. https://doi.org/10.1016/s2215-0366(20)30168
- Holmes, K., & Sher, L. (2013). Dating violence and suicidal behavior in adolescents.
 International Journal of Adolescent Medicine and Health, 25(3), 257–261.
 https://doi.org/10.1515/ijamh-2013-0059
- Holubčíková, J., Kudláček, M., Širůček, J., & Madarasová Gecková, A. (2018). Testretest reliability of selected HBSC items measuring problem behaviour among Slovak and Czech adolescents. *Central European Journal of Public Health*, 26(3), 204–208. https://doi.org/10.21101/cejph.a4662
- Hong, S., Satyshur, M. D., & Burnett-Zeigler, I. (2023). The association of mindfulness and depression stigma among African American women participants in a

mindfulness-based intervention: A pilot study. *Transcultural Psychiatry*, 60(2), 244-254.

- Im, Y., Oh, W. O., & Suk, M. (2017). Risk factors for suicide ideation among adolescents: five-year national data analysis. *Archives of Psychiatric Nursing*, 31(3), 282–286. https://doi.org/10.1016/j.apnu.2017.01.001
- Islam, M. I., Khanam, R., & Kabir, E. (2020). Bullying victimization, mental disorders, suicidality, and self-harm among Australian high schoolchildren: Evidence from nationwide data. *Psychiatry Research*, 292, 113364.
- Isumi, A., Doi, S., Yamaoka, Y., Takahashi, K., & Fujiwara, T. (2020). Do suicide rates in children and adolescents change during school closure in Japan? The acute effect of the first wave of COVID-19 pandemic on child and adolescent mental health. *Child Abuse and Neglect*, *110*, 104680. https://doi.org/10.1016/j.chiabu.2020.104680
- Ivanich, J., & Teasdale, B. (2017). Suicide ideation among adolescent American Indians: An application of general strain theory. *Deviant Behavior*, 39(6), 702– 715. https://doi.org/10.1080/01639625.2017.1304799
- Ivey-Stephenson, A. Z., Demissie, Z., Crosby, A. E., Stone, D. M., Gaylor, E., Wilkins, N., Lowry, R., & Brown, M. (2020). Suicidal ideation and behaviors among high school students—Youth risk behavior survey, United States, 2019. MMWR Supplements, 69(1), 47–55. https://doi.org/10.15585/mmwr.su6901a6.
- Iwatate, E., Atem, F. D., Jones, E. C., Hughes, J. L., Yokoo, T., & Messiah, S. E. (2023). Association of obesity, suicide behaviors, and psychosocial wellness

among adolescents in the United States. *Journal of Adolescent Health*, 72(4), 526-534.

- Jacka, F. N., O'Neil, A., Opie, R., Itsiopoulos, C., Cotton, S., Mohebbi, M., & Berk, M. (2017). A randomized controlled trial of dietary improvement for adults with major depression (the 'SMILES 'trial). *BMC Medicine*, 15, 1-13.
- John, A., Okolie, C., Eyles, E., Webb, R. T., Schmidt, L., McGuiness, L. A., Olorisade,
 B. K., Arensman, E., Hawton, K., Kapur, N., Moran, P., O'Connor, R. C.,
 O'Neill, S., Higgins, J. P., & Gunnell, D. (2020). The impact of the COVID-19
 pandemic on self-harm and suicidal behavior: A living systematic review.
 https://doi.org/10.12688/f1000research.25522.1
- Johns, M. M., Lowry, R., Hipp, T. N., Robin, L., & Shafir, S. (2020). Differences in adolescent experiences of polyvictimization and suicide risk by sexual minority status. *Journal of Research on Adolescence*, 31(1) 240-252. https://doi.org/10.1111/jora.12595
- Kalin, N. H. (2021). Anxiety, depression, and suicide in youth. American Journal of Psychiatry, 178(4), 275–279. https://doi.org/10.1176/appi.ajp.2020.21020186
- Kandola, A., Hendrikse, J., Lucassen, P. J., & Yücel, M. (2016). Aerobic exercise as a tool to improve hippocampal plasticity and function in humans: Practical implications for mental health treatment. *Frontiers in Human Neuroscience*, *10*, 373.
- Kang H. (2013). The prevention and handling of missing data. *Korean Journal of Anesthesiology*, *64*(5), 402–406. https://doi.org/10.4097/kjae.2013.64.5.402

- Kann, L. (2018). Youth risk behavior surveillance—United States, 2017. MMWR. Surveillance Summaries, 67.
- Kessler, R. C., Avenevoli, S., Green, J., Gruber, M. J., Guyer, M., He, Y., & Merikangas, K. R. (2009). National comorbidity survey replication adolescent supplement (NCS-A): III. Concordance of DSM-IV/CIDI diagnoses with clinical reassessments. *Journal of the American Academy of Child and Adolescent Psychiatry*, 48(4), 386-399.
- Kiani, Z., Simbar, M., Mokber, N., Kiani, M., Azimi, N., & Kazemi, S. (2021). Suicide Risk factors among adolescents: A narrative review. *Advances in Nursing & Midwifery*, 30(2), 34-51.
- Kiani, Z., Simbar, M., Mokber, N., Kiani, M., Azimi, N., & Kazemi, S. (2021). Suicide risk factors among adolescents: A narrative review. *Advances in Nursing & Midwifery*, 30(2), 34-51.
- Kim, H. W., Shin, C., Han, K. M., & Han, C. (2019). The effect of physical activity on suicidal ideation differs by gender and activity level. *Journal of Affective Disorders*, 257, 116–122. https://doi.org/10.1016/j.jad.2019.07.043
- King, C. A., O'mara, R. M., Hayward, C. N., & Cunningham, R. M. (2009). Adolescent suicide risk screening in the emergency department. *Academic Emergency Medicine*, 16(11), 1234-1241.
- Klomek, A. B., Kleinman, M., Altschuler, E., Marrocco, F., Amakawa, L., & Gould, M.
 S. (2011). High school bullying as a risk for later depression and suicidality. *Suicide and Life-Threatening Behavior*, *41*(5), 501-516.

- Klomek, A. B., Sourander, A., Niemelä, S., Kumpulainen, K., Piha, J., Tamminen, T., ...
 & Gould, M. S. (2009). Childhood bullying behaviors as a risk for suicide attempts and completed suicides: A population-based birth cohort study. *Journal of the American Academy of Child and Adolescent Psychiatry*, 48(3), 254-261.
- Klonsky, E. D., May, A. M., & Saffer, B. Y. (2016). Suicide, suicide attempts, and suicidal ideation. *Annual Review of Clinical Psychology*, 12, 307-330.
- Kowalski, R. M., & Limber, S. P. (2013). Psychological, physical, and academic correlates of cyberbullying and traditional bullying. *Journal of Adolescent Health*, 53(1), S13–S20. https://doi.org/10.1016/j.jadohealth.2012.09.018
- Labuhn, M., LaBore, K., Ahmed, T., & Ahmed, R. (2021). Trends and instigators among young adolescent suicide in the United States. *Public Health*, 199, 51–56. https://doi.org/10.1016/j.puhe.2021.08.004
- LaRocca, D., James, K. A., Rosenberg, S., Ma, M., & Brooks-Russell, A. (2023). Team sports participation, depression, and suicidal ideation in lesbian, gay, bisexual, transgender, and questioning adolescents. *Psychology in the Schools*, 60(4), 902-911.
- Larsen, L., Schauber, S.K., & Holt, T. (2023). Longitudinal Covid-19 effects on child mental health: Vulnerability and age dependent trajectories. *Child Adolescent Psychiatry Mental Health 17, 104* https://doi.org/10.1186/s13034-023-00652-5
- Lavie, C. J., Ozemek, C., Carbone, S., Katzmarzyk, P. T., & Blair, S. N. (2019). Sedentary behavior, exercise, and cardiovascular health. *Circulation Research*, 124(5), 799-815.

- Lee, I. M., Shiroma, E. J., Lobelo, F., Puska, P., Blair, S. N., & Katzmarzyk, P. T. (2012). Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. *The Lancet, 380*(9838), 219–229. https://doi.org/10.1016/s0140-6736(12)61031-9
- Letourneau, E. J., Brown, D. S., Fang, X., Hassan, A., & Mercy, J. A. (2018). The economic burden of child sexual abuse in the United States. *Child Abuse and Neglect*, *79*, 413-422.
- Lewinsohn, P. M., Joiner Jr, T. E., & Rohde, P. (2001). Evaluation of cognitive diathesis-stress models in predicting major depressive disorder in adolescents. *Journal of Abnormal Psychology*, *110*(2), 203.
- Li, X., Xiang, S. T., & Dong, J. (2022). The concurrence of sexual violence and physical fighting among adolescent suicide ideators and the risk of attempted suicide. *Scientific Reports*, 12(1). https://doi.org/10.1038/s41598-022-09387-3
- Lima, C. A. G., Maia, M. D. F. M., Brito, M. F. S. F., Pinho, L. D., & Silveira, M. F.
 (2020). Psychometric properties of the Youth Risk Behavior Survey (YRBS)
 instrument in Brazilian college students. *Paidéia (Ribeirão Preto)*, 30, e3006.
- Limbana, T., Khan, F., Eskander, N., Emamy, M., & Jahan, N. (2020). The association of bullying and suicidality: Does it affect the pediatric population? 12(8), e9691. https://doi.org/10.7759/cureus.9691
- Litwiller, B. J., & Brausch, A. M. (2013). Cyber bullying and physical bullying in adolescent suicide: The role of violent behavior and substance use. *Journal of Youth and Adolescence*, 42(5), 675–684. https://doi.org/10.1007/s10964-013-9925-5

- Liu, J., & Graves, N. (2011). Childhood bullying: A review of constructs, concepts, and nursing implications. *Public Health Nursing*, *28*(6), 556-568.
- Liu, J., Zhu, X., Liu, Y., Jia, F., Yuan, H., Wang, Q., ... & Zhang, X. (2023). Association between triglyceride glucose index and suicide attempts in patients with firstepisode drug-naïve major depressive disorder. *Frontiers in Psychiatry*, 14, 1231524.
- Lubans, D. R., Plotnikoff, R. C., & Lubans, N. J. (2012). A systematic review of the impact of physical activity programmes on social and emotional well-being in atrisk youth. *Child and Adolescent Mental Health*, 17(1), 2-13.
- Macalli, M., Orri, M., Tzourio, C., & Côté, S. M. (2021). Contributions of childhood peer victimization and/or maltreatment to young adult anxiety, depression, and suicidality: a cross-sectional study. *BMC Psychiatry; BioMed Central*. https://doi.org/10.1186/s12888-021-03354-4
- Mahindru, A., Patil, P., & Agrawal, V. (2023). Role of physical activity on mental health and well-being: A review. *Cureus*, 15(1), e33475. https://doi.org/10.7759/cureus.33475
- Mainali, P., Motiwala, F., Trivedi, C., Vadukapuram, R., Mansuri, Z., & Jain, S. (2023, JanF 17). Sexual abuse and its impact on suicidal ideation and attempts and psychiatric illness in children and adolescents with posttraumatic stress disorder. *The Primary Care Companion for CNS Disorders*. https://doi.org/10.4088/pcc.22m03239
- Malik, H. A., & Malik, F. A. (2022). Emile Durkheim contributions to sociology. Sociology, 6(2), 7-10.

- Mammen, G., & Faulkner, G. (2013). Physical activity and the prevention of depression:
 a systematic review of prospective studies. *American Journal of Preventive Medicine*, 45(5), 649-657.
- Mamun, M. A., & Griffiths, M. D. (2020). First COVID-19 suicide case in Bangladesh due to fear of COVID-19 and xenophobia: Possible suicide prevention strategies. *Asian Journal of Psychiatry*, 51, 102073.
- Manczak, E. M., Ordaz, S. J., Singh, M. K., Goyer, M. S., & Gotlib, I. H. (2019). Time spent with parents predicts change in depressive symptoms in adolescents with major depressive disorder. *Journal of Abnormal Child Psychology*, 47(8), 1401– 1408. https://doi.org/10.1007/s10802-019-00526-5
- Mann, J. J., Apter, A., Bertolote, J., Beautrais, A., Currier, D., Haas, A., ... & Hendin, H.
 (2005). Suicide prevention strategies: A systematic review. *JAMA*, 294(16), 2064-2074.
- Mantey, D. S., Yockey, R. A., & Springer, A. E. (2023). Digital screen time and suicidality during high school: How important is cyberbullying? A mediation analysis using the youth risk behavioral surveillance survey, 2011–2019. *Preventive Medicine, 166*, 107330. https://doi.org/10.1016/j.ypmed.2022.107330
- Martin, G., Bergen, H. A., Richardson, A. S., Roeger, L., & Allison, S. (2004). Sexual abuse and suicidality: gender differences in a large community sample of adolescents. *Child Abuse and Neglect*, 28(5), 491–503. https://doi.org/10.1016/j.chiabu.2003.08.006

- Mather, A. A., Cox, B. J., Enns, M. W., & Sareen, J. (2009). Associations of obesity with psychiatric disorders and suicidal behaviors in a nationally representative sample. *Journal of Psychosomatic Research*, 66(4), 277-285.
- Mathesius, J., & Lussier, P. (2021). Structural equation modeling. In J. C. Barnes and David R. Forde (Eds.) *The Encyclopedia of Research Methods in Criminology and Criminal Justice*, 884–889. https://doi.org/10.1002/9781119111931.ch172
- May, A., & Klonsky, E. D. (2010). Validity of suicidality items from the youth risk behavior survey in a high school sample. *Assessment*, 18(3), 379–381.
 https://doi.org/10.1177/1073191110374285
- Mazza, M., Marano, G., Del Castillo, A. G., Chieffo, D. P. R., Monti, L., Janiri,
 D., Moccia, L., & Sani, G. (2021). Intimate partner violence: A loop of abuse, depression, and victimization. *World Journal of Psychiatry 11*(6) 215-221. doi: 10.5498/wjp.v11.i6.215
- McGorry, P. D., Mei, C., Chanen, A., Hodges, C., Alvarez-Jimenez, M., & Killackey, E. (2022). Designing and scaling up integrated youth mental health care. *World Psychiatry*, 21(1), 61–76. https://doi.org/10.1002/wps.20938
- McKinnon, I. I. (2023). Experiences of unstable housing among high school students— Youth Risk Behavior Survey, United States, 2021. *MMWR supplements*, 72.
- McMahon, E. M., Hemming, L., Robinson, J., & Griffin, E. (2023). Editorial: Suicide and self-harm in young people. *Frontiers in Psychiatry*, 13. https://doi.org/10.3389/fpsyt.2022.1120396.

- Merikangas, K. R., He, J. P., Brody, D., Fisher, P. W., Bourdon, K., & Koretz, D. S. (2010). Prevalence and treatment of mental disorders among US children in the 2001–2004 NHANES. *Pediatrics*, 125(1), 75-81.
- Merton, R. K. (1938). Social structure and anomie. *American Sociological Review, 3*(5), 672. https://doi.org/10.2307/2084686

Messias, E., Kindrick, K., & Castro, J. (2014). School bullying, cyberbullying, or both: Correlates of teen suicidality in the 2011 CDC youth risk behavior survey. *Comprehensive Psychiatry*, 55(5), 1063–1068. https://doi.org/10.1016/j.comppsych.2014.02.005

- Mitchell, J. C., MacLeod, B. P., & Cassisi, J. E. (2017). Modeling sexual assault risk perception among heterosexual college females: The impact of previous victimization, alcohol use, and coping style. *Violence Against Women*, 23(2), 143-162.
- Modecki, K. L., Minchin, J., Harbaugh, A. G., Guerra, N. G., & Runions, K. C. (2014).
 Bullying prevalence across contexts: A meta-analysis measuring cyber and traditional bullying. *Journal of Adolescent Health*, 55(5), 602–611.
 https://doi.org/10.1016/j.jadohealth.2014.06.007
- Morneau-Vaillancourt, G., Orri, M., Geoffroy, M. C., & et al. (2023). Polygenic prediction of depressive symptoms, peer victimization, school dropout, and suicidal behaviors. *European Neuropsychopharmacology* 75, S37-S38. https://doi:10.1016/j.euroneuro.2023.08.077
- Mpofu, J. J., Underwood, J. M., Thornton, J., Brener, N. D., Rico, A., Kilmer, G., Harris, W. A., Leon-Nguyen, M., Chyen, D., Lim, C., Mbaka, C. K., Smith-

Grant, J., Whittle, L., Jones, S. E., Krause, K. H., Li, J., Shanklin, S. L., McKinnon, I., Arrey, L., . . . Roberts, A. (2023). Overview and methods for the Youth Risk Behavior Surveillance System — United States, 2021. *MMWR Supplements*. https://doi.org/10.15585/ mmwr.su7201a1

- Muratore, M. G. (2014). Victimization. In Alex C. Michalos (Ed.) Encyclopedia of quality of life and well-being research, p. 6917–6921. Springer Reference. https://doi.org/10.1007/978-94-007-0753-5_3156
- National Center for Health Statistics. CDC (2016). *Health, United States, 2016*. Centers for Disease Control and Prevention.

https://www.cdc.gov/nchs/data/hus/hus16.pdf

- Ndetei, D. M., Mutiso, V. N., Weisz, J. R., Okoth, C. A., Musyimi, C., Muia, E. N., ... & Mamah, D. (2022). Socio-demographic, economic, and mental health problems were risk factors for suicidal ideation among Kenyan students aged 15 plus. *Journal of Affective Disorders*, 302, 74-82.
- Nguyen, T. H., Shah, G., Muzamil, M., Ikhile, O., Ayangunna, E., & Kaur, R. (2023). Association of in-school and electronic bullying with suicidality and feelings of hopelessness among adolescents in the United States. *Children*, 10(4), 755.
- NIH. (2022). Cyberbullying is linked with suicidal thoughts and attempts in young. National Institutes of Health (NIH). https://www.nih.gov/news-events/nihresearch-matters/cyberbullying-linked-suicidal-thoughts-attempts-youngadolescents.
- Ning, K., Yan, C., Zhang, Y., & Chen, S. (2022). Regular exercise with suicide ideation, suicide plan and suicide attempt in university students: Data from the Health

Minds Survey 2018-2019. International Journal of Environmental Research and Public Health, 19(14), 8856. https://doi.org/10.3390/ijerph19148856

- Nixon, C. (2014). Current perspectives: The impact of cyberbullying on adolescent health. Adolescent Health, Medicine, and Therapeutics, 5, 143-158. https://doi.org/10.2147/ahmt.s36456
- Nock, M. K., Borges, G., Bromet, E. J., Alonso, J., Angermeyer, M., Beautrais, A.,
 Bruffaerts, R., Chiu, W. T., de Girolamo, G., Gluzman, S., de Graaf, R., Gureje,
 O., Haro, J. M., Huang, Y., Karam, E., Kessler, R. C., Lepine, J. P., Levinson,
 D., Medina-Mora, M. E., . . . Williams, D. (2008). Cross-national prevalence and
 risk factors for suicidal ideation, plans, and attempts. British *Journal of Psychiatry*, *192*(2), 98–105. https://doi.org/10.1192/bjp.bp.107.040113
- Nolen-Hoeksema, S. (2001). Gender differences in depression. *Current Directions in Psychological Science*, 10(5), 173-176.
- Okada M, Matsumoto R, Shiroyama T, Motomura E. Suicidal mortality and motives among middle school, high school, and university students. *JAMA: 2023; 6*(8) e2328144. https://doi:10.1001/jamanetworkopen.2023.28144
- Okobi, O. E., Egbujo, U., Darke, J., Odega, A. S., Okereke, O. P., Adisa, O. T., Salawu, M. A., & Kimble, R. (2023). Association of bullying victimization with suicide ideation and attempt among school-going adolescents in post-conflict Liberia: Findings from the global school-based health survey. *Cureus*, 15(6), e40077. https://doi.org/10.7759/cureus.40077
- Olshen, E., McVeigh, K. H., Wunsch-Hitzig, R. A., & Rickert, V. I. (2007). Dating violence, sexual assault, and suicide attempts among urban teenagers. *Archives*

of Pediatrics and Adolescent Medicine, 161(6), 539.

https://doi.org/10.1001/archpedi.161.6.539

- Ong, M. S., Lakoma, M., Gees Bhosrekar, S., Hickok, J., McLean, L., Murphy, M., ... & Ross-Degnan, D. (2021). Risk factors for suicide attempts in children, adolescents, and young adults hospitalized for mental health disorders. *Child and Adolescent Mental Health*, 26(2), 134-142.
- Orban E., Li L. Y., Gilbert M., Napp A. K., Kaman A., Topf S., Boecker M., Devine J., Rei B. F., Wendel F., Jung-Sievers C., Ernst V. S, Franze M., Möhler E., Breitinger E., Bender S. & Ravens-Sieberer U. (2024) Mental health and quality of life in children and adolescents during the COVID-19 pandemic: A systematic review of longitudinal studies. *Public Health*.11:1275917.
- Orben, A., Tomova, L., & Blakemore, S. J. (2020). The effects of social deprivation on adolescent development and mental health. *The Lancet Child and Adolescent Health*, 4(8), 634–640. https://doi.org/10.1016/s2352-4642(20)30186
- Pappas, S. (2023). More than 20% of teens have seriously considered suicide. Psychologists and communities can help tackle the problem. APA Monitor 54 (5) 54. https://www.apa.org/monitor/2023/07/psychologists-preventing-teen-suicide.
- Parchem, B., Wheeler, A., Talaski, A., & Molock, S. D. (2024). Comparison of anxiety and depression rates among LGBTQ college students before and during the COVID-19 pandemic. *Journal of American College Health*, 72(1), 31-39.
- Parker, K., Uddin, R., Ridgers, N. D., Brown, H., Veitch, J., Salmon, J., ... & Arundell,L. (2021). The use of digital platforms for adults' and adolescents' physical

activity during the COVID-19 pandemic (our life at home): survey study. *Journal of Medical Internet Research*, *23*(2), e23389.

Patchin, J. W., Hinduja, S., & Meldrum, R. C. (2023). Digital self-harm and suicidality among adolescents. *Child and Adolescent Mental Health*, *28*(1), 52-59.

Pathania, M., Kumari, R., Jain, V., Das, A., & et al. (2023). Mind-body medicine workshop on path ahead in managing lifestyle disorders: Add meditation to your medication. *Journal of Medical Evidence*. https://journals.lww.com/jome/_layouts/15/oaks.journals/downloadpdf.aspx?an= 02273353-202304020-00019

- Perez, N. M., Jennings, W. G., Piquero, A. R., & Baglivio, M. T. (2016). Adverse childhood experiences and suicide attempts: The mediating influence of personality development and problem behaviors. *Journal of Youth and Adolescence*, 45 (8), 1527-1545.
- Piolanti, A., Waller, F., Schmid, I. E., & Foran, H. M. (2023). Long-term adverse outcomes associated with teen dating violence: A systematic review. *Pediatrics*, 151(6), e2022059654.
- Pontes, N. M. H., Ayres, C. G., & Pontes, M. C. F. (2020). Trends in depressive symptoms and suicidality. *Nursing Research*, 69(3), 176–185. https://doi.org/10.1097/nnr.000000000000424
- Powell, K. E., Caspersen, C. J., Koplan, J. P., & Ford, E. S. (1989). Physical activity and chronic diseases. *The American Journal of Clinical Nutrition*, 49(5), 999–1006. https://doi.org/10.1093/ajcn/49.5.999

Radell, M. L., Abo Hamza, E. G., Daghustani, W. H., Perveen, A., & Moustafa, A. A.
(2021). The impact of different types of abuse on depression. *Depression Research and Treatment*, April 13, 6654503.

https://doi.org/10.1155/2021/6654503

Rasberry, C. N., Lowry, R., Johns, M., Robin, L., Dunville, R., Pampati, S., Dittus, P. J., & Balaji, A. (2018). Sexual risk behavior differences among sexual minority high school students - United States, 2015 and 2017. *MMWR. Morbidity and mortality weekly report, 67*(36), 1007–1011. https://doi.org/10.15585/mmwr.mm6736a3

- Raykov, T., & Mels, G. (2007). Lower-level mediation effect analysis in two-level studies: A note on a multilevel structural equation modeling approach. *Structural Equation Modeling: A Multidisciplinary Journal*, *14*(4), 636–648. https://doi.org/10.1080/10705510701575511
- Rebar, A. L., Stanton, R., Geard, D., Short, C., Duncan, M. J., & Vandelanotte, C.
 (2015). A meta-meta-analysis of the effect of physical activity on depression and anxiety in non-clinical adult populations. *Health Psychology Review*, 9(3), 366-378.
- Reeping, P. M., Klarevas, L., Rajan, S., Rowhani-Rahbar, A., Heinze, J., Zeoli, A. M., Goyal, M. K., Zimmerman, M. A., & Branas, C. C. (2022). State firearm laws, gun ownership, and k-12 school shootings: Implications for school safety. *Journal of School Violence*, *21*(2), 132–146. https://doi.org/10.1080/15388220.2021.2018332

- Richardson, T., Elliott, P., & Roberts, R. (2013). The relationship between personal unsecured debt and mental and physical health: A systematic review and metaanalysis. *Clinical Psychology Review*, 33(8), 1148-1162.
- Ricks, J. M., Montgomery, C. M., & Nash, J. A. (2023). Measurement of adolescent dating violence in sexual minority youth: a scoping review. *Aggression and Violent Behavior*, 73 101870.
- Romeo R. D. (2013). The teenage brain: The stress response and the adolescent brain. *Current Directions in Psychological Science*, 22(2), 140–145. https://doi.org/10.1177/0963721413475445
- Rosales, R., Sellers, C. M., Lee, C. S., Santos, B., O'Brien, K., & Colby, S. M. (2023). Examining racial/ethnic differences in the association of victimization and suicidal thoughts and behaviors with alcohol use among sexual minority youth. *LGBTQ Health*, *10*(2), 109-120.
- Ruch, D. A., Sheftall, A. H., Schlagbaum, P., Rausch, J., Campo, J. V., & Bridge, J. A.
 (2019). Trends in suicide among youth aged 10 to 19 years in the United States, 1975 to 2016. *JAMA*, 2(5), e193886.

https://doi.org/10.1001/jamanetworkopen.2019.3886

Salmon, S., Dávila, I. G., Taillieu, T., Stewart-Tufescu, A., Duncan, L., Fortier, J., Struck, S., Georgiades, K., MacMillan, H. L., Kimber, M., González, A., & Afifi, T. O. (2022). Adolescent health outcomes: associations with child maltreatment and peer victimization. *BMC Public Health; BioMed Central*. https://doi.org/10.1186/s12889-022-13310-w Sander, S., Strizzi, J. M., Øverup, C. S., Cipric, A., & Hald, G. M. (2020). When love hurts-mental and physical health among recently divorced danes. *Frontiers in Psychology*, 11, 578083.

Sara, G., Wu, J., Uesi, J., Jong, N., Perkes, I., Knight, K., ... & Bowden, M. (2023).
Growth in emergency department self-harm or suicidal ideation presentations in young people: Comparing trends before and since the COVID-19 first wave in New South Wales, Australia. *Australian & New Zealand Journal of Psychiatry*, 57(1), 58-68.

- Sawyer, S. M., Azzopardi, P. S., Wickremarathne, D., & Patton, G. C. (2018). The age of adolescence. . .and young adulthood – Authors' reply. *The Lancet Child & Adolescent Health*, 2(4), e7. https://doi.org/10.1016/s2352-4642(18)30075-0
- Schuch, F. B., & Vancampfort, D. (2021). Physical activity, exercise, and mental disorders: It is time to move on. *Trends in Psychiatry and Psychotherapy*, 43(3), 177–184. https://doi.org/10.47626/2237-6089-2021-0237
- Schwartz, S., & Meyer, I. H. (2010). Mental health disparities research: The impact of within and between group analyses on tests of social stress hypotheses. *Social Science and Medicine*, 70(8), 1111-1118.
- Selkie, E. M., Fales, J. L., & Moreno, M. A. (2016). Cyberbullying prevalence among US middle and high school–aged adolescents: A systematic review and quality assessment. *Journal of Adolescent Health*, 58(2), 125–133.
- Shaikh, M. A., Abio, A., Celedonia, K. L., & Lowery Wilson, M. (2019). Physical fighting among school-attending adolescents in Pakistan: Associated factors and

contextual influences. *International Journal of Environmental Research and Public Health*, *16*(24), 5039. https://doi.org/10.3390/ijerph16245039.

- Shain, B., Braverman, P. K., Adelman, W. P., Alderman, E. M., Breuner, C. C., Levine,
 D. A., ... & O'Brien, R. F. (2016). Suicide and suicide attempts in adolescents. *Pediatrics*, 138(1).
- Shireen, F., Janapana, H., Rehmatullah, S., Temuri, H., & Azim, F. (2014). Trauma experience of youngsters and teens: A key issue in suicidal behavior among victims of bullying? *Pakistan Journal of Medical Sciences*, 30(1), 206-210. http://dx.doi.org/10.12669/pjms.301.4072
- Siah, P. C., Tee, X. Y., Tan, J. T. A., Tan, C. S., Lokithasan, K., Low, S. K., & Yap, C.
 C. (2022). Cybervictimization and depression among adolescents: Coping strategies as mediators. *International Journal of Environmental Research and Public Health*, 19(7), 3903. https://doi.org/10.3390/ijerph19073903

Sigel, E. J., Mattson, S. A., & Mercado, M. C. (2019). Increased violence involvement and other behavioral and mental health factors among youth with firearm access. *The Journal of Adolescent Health: Official Publication of the Society for Adolescent Medicine*, 65(1), 63–71.

https://doi.org/10.1016/j.jadohealth.2019.01.028

Singh, B., Olds, T., Curtis, R., Dumuid, D., Virgara, R., Watson, A., ... & Maher, C. (2023). Effectiveness of physical activity interventions for improving depression, anxiety, and distress: an overview of systematic reviews. *British Journal of Sports Medicine*, 57(18), 1203-1209.

- Smith, E. G., & Patel, K. M. (2024). The role of case series and case reports in evidencebased medicine. *Journal of Clinical Psychopharmacology*, 44(2), 81-85.
- Smith, K., & Crawford, S. (1986). Suicidal behavior among "normal" high school students. *Suicide and Life-Threatening Behavior*, *16*(3), 313–325. https://doi.org/10.1111/j.1943-278x.1986.tb01013.x

Smith, K., Brendgen, M., Hébert, M., Vitaro, F., Dionne, G., & Boivin, M. (2021). Links between peer victimization, dating violence victimization and depression in adolescence: A genetically-informed study. *Journal of Clinical Child and Adolescent Psychology, 52*(4), 558–569.

https://doi.org/10.1080/15374416.2021.2001746

- Somerville, L. H. (2013). The teenage brain. *Current Directions in Psychological Science*, *22*(2), 121–127. https://doi.org/10.1177/0963721413476512
- Stansfeld, S. A., Clark, C., Smuk, M., Power, C., Davidson, T., & Rodgers, B. (2016).
 Childhood adversity and midlife suicidal ideation. *Psychological Medicine*, 47(2), 327–340. https://doi.org/10.1017/s0033291716002336
- Stewart, S. L., Celebre, A., Hirdes, J. P., & Poss, J. W. (2020). Risk of suicide and selfharm in kids: The development of an algorithm to identify high-risk individuals within the children's mental health system. *Child Psychiatry & Human Development*, 51, 913-924.

Substance Abuse and Mental Health Services Administration (SAMHSA). (2022). Cooccurring disorders. *SAMHSA*. https://www.samhsa.gov/co-occurring-disorders

Substance Abuse and Mental Health Services Administration. (2020). Key substance use and mental health indicators in the United States: Results from the 2019 National Survey on Drug Use and Health.

https://store.samhsa.gov/sites/default/files/pep20-06-01-002.pdf

- Sun, G., Zhao, J., Tian, S., Zhang, L., & Jia, C. (2020). Psychological strain and suicidal ideation in athletes: The multiple mediating effects of hopelessness and depression. *International Journal of Environmental Research and Public Health*, *17*(21), 8087. https://doi.org/10.3390/ijerph17218087
- Sun, M., Wang, D., Jing, L., & Zhou, L. (2023). The predictive role of psychotic-like experiences in suicidal ideation among technical secondary school and college students during the COVID-19 pandemic. *BMC Psychiatry*, 23(1), 521.
- Swahn, M., Bossarte, R., Palmier, J., & Yao, H. (2013). Co-occurring physical fighting and suicide attempts among US high school students: Examining patterns of early alcohol use initiation and current binge drinking. *Western Journal of Emergency Medicine*, 14(4), 341–346.
 - https://doi.org/10.5811/westjem.2013.3.15705
- Taquet, M., Luciano, S., Geddes, J. R., & Harrison, P. J. (2021). Bidirectional associations between COVID-19 and psychiatric disorder: Retrospective cohort studies. *The Lancet Psychiatry*, 8(2), 130-140.
- Thomas-Moody, F. T. (2021). African American fathers' experiences of alienation from their children due to Texas Family Code (Doctoral dissertation, Walden University). ScholarWorks.

https://scholarworks.waldenu.edu/dissertations/10613/

Thompson, E. C., Thomas, S. A., Burke, T. A., Nesi, J., MacPherson, H. A., Bettis, A. H., ... & Wolff, J. C. (2021). Suicidal thoughts and behaviors in psychiatrically

hospitalized adolescents pre-and post-COVID-19: A historical chart review and examination of contextual correlates. *Journal of Affective Disorders Reports*, *4*, 100100.

- Thompson, M. P., Kingree, J. B., & Lamis, D. (2018). Associations of adverse childhood experiences and suicidal behaviors in adulthood in a US nationally representative sample. *Child: Care, Health, and Development, 45*(1), 121–128. https://doi.org/10.1111/cch.12617
- Tu, W., Jiang, H., & Liu, Q. (2022). Peer victimization and adolescent mobile social addiction: Mediation of social anxiety and gender differences. *International Journal of Environmental Research and Public Health*, 19(17), 10978. https://doi.org/10.3390/ijerph191710978
- Turanovic, & Siennick . (2022). The causes and consequences of school violence: A review. In US Department of Justice. National Institute of Justice. https://www.ojp.gov/pdffiles1/nij/302346.pdf.
- Turner, H. A., Shattuck, A., Finkelhor, D., & Hamby, S. (2016). Polyvictimization and youth violence exposure across contexts. *Journal of Adolescent Health*, 58(2), 208-214.
- Underwood, J. M., Brener, N. D., Thornton, J., Harris, W. A., Bryan, L., Shanklin, S. L., Deputy, N. P., Roberts, A., Queen, B., Chyen, D., Whittle, L., Lim, C., Yamakawa, Y., Leon-Nguyen, M., Kilmer, G., Smith-Grant, J., Demissie, Z., Jones, S. E., Clayton, H. B., & Dittus, P. (2020). Overview and methods for the youth risk behavior surveillance system United States, 2019. *MMWR Supplements*. https://doi.org/10.15585/mmwr.su6901a1

- USDOJ: United States Department of Justice Archive Appendix C. (n.d.). https://www.justice.gov/archive/opd/AppendixC.htm#N_6_
- Van Brunschot, E. G., & Humphrey, T. (2022). *Pathways to ruin? High-risk offending over the life course*. University of Toronto Press.
- Van Wyk, J. A. (2023). ACEs and angst: Adverse childhood experiences, general strain theory, and adolescent male suicidal and violent behaviors. SAGE Open, 13(2). https://doi.org/10.1177/21582440231176710
- Vancampfort, D., Hallgren, M., Firth, J., Rosenbaum, S., Schuch, F. B., Mugisha, J., Probst, M., Van Damme, T., Carvalho, A. F., & Stubbs, B. (2018). Physical activity and suicidal ideation: A systematic review and meta-analysis. *Journal of Affective Disorders*, 225, 438–448. https://doi.org/10.1016/j.jad.2017.08.070.
- VandenBos, G. R. (2007). *APA dictionary of psychology*. American Psychological Association.
- Wade, T., George, W. M., & Atkinson, M. (2009). A randomized controlled trial of brief interventions for body dissatisfaction. *Journal of Consulting and Clinical Psychology*, 77(5), 845.
- Walker, R. L., Talavera, D. C., Nomamiukor, F., Madubata, I. J., Alfano, C., & Vujanovic, A. A. (2019). Sleep-related problems and suicide behavior and ideation among Black and White trauma-exposed psychiatric inpatients. *Comprehensive Psychiatry*, 91, 22-28.
- Walls, M. L., Chapple, C. L., & Johnson, K. D. (2007). Strain, emotion, and suicide among American Indian youth. *Deviant Behavior*, 28(3), 219-246.

- Weir, K. (2023). Social media brings benefits and risks to teens. Here's how psychology can help identify a path forward. *Monitor on Psychology*, 54(6).
 https://www.apa.org/monitor/2023/09/protecting-teens-on-social-media
- WHO. (2022). *Mental health*. https://www.who.int/news-room/fact-sheets/detail/mentalhealth-strengthening-our-response.
- Wilson, E., Crudgington, H., Morgan, C., Hirsch, C., Prina, M., & Gayer-Anderson, C. (2023). The longitudinal course of childhood bullying victimization and associations with self-injurious thoughts and behaviors in children and young people: A systematic review of the literature. *Journal of Adolescence*, 95(1), 5-33.
- Wolff, J. C., Davis, S., Liu, R. T., Cha, C. B., Cheek, S. M., Nestor, B. A., Frazier, E.
 A., Schaffer, M. M., & Spirito, A. (2017). Trajectories of suicidal ideation among adolescents following psychiatric hospitalization. *Journal of Abnormal Child Psychology*, 46(2), 355–363. https://doi.org/10.1007/s10802-017-0293-6
- Wolford-Clevenger, C., Elmquist, J., Brem, M., Zapor, H., & Stuart, G. L. (2016).
 Dating violence victimization, interpersonal needs, and suicidal ideation among college students. *Crisis*, 37(1), 51–58. https://doi.org/10.1027/0227-5910/a000353.

Yanez-Peñúñuri, L. Y., Rey-Anacona, C. A., & Garcia-Estrada, L. I. (2023).
Relationship between dating violence and mental health of young victims and perpetrators: A systematic review. *International Journal of Psychological Research*, *16*(1), 116–139. https://doi.org/10.21500/20112084.5710
- Yang, Y. (2023). Suicide attempt and suicide plan among US adolescents: The role of repeated and co-occurring violence experiences. *Psychiatry Research, 320*, 115040. https://doi.org/10.1016/j.psychres.2022.115040
- Yang, Y., Liller, K. D., Coulter, M., Salinas-Miranda, A., Tyson, D. M., & Chen, H. (2021). How community and individual risk factors mutually impact youth's perceived safety: A systematic analysis using structural equation modeling. *Journal of Interpersonal Violence*, *37*(19–20), NP17738–NP17757. https://doi.org/10.1177/08862605211028326
- Yang, Y., Salinas-Miranda, A., Coulter, M., & Liller, K. D. (2020). Understanding weapon carrying behaviors of youth in Florida schools using structural equation modeling. *Violence and Gender*, 7(2), 70–77. https://doi.org/10.1089/vio.2019.0024
- Yao, Z. Y., Wang, T., Yu, Y. K., Li, R., Sang, X., Fu, Y. N., ... & Jia, C. X. (2023).
 Mental health literacy and suicidal ideation among Chinese college students: The mediating role of depressive symptoms and anxiety symptoms. *Journal of Affective Disorders*, 339, 293-301.
- Yıldız, M., & Solakoglu, Z. (2017). Strain, negative emotions, and suicidal behaviors among adolescents. *Youth and Society*, 51(5), 638–658.
 https://doi.org/10.1177/0044118x17700318
- Yosep, I., Hikmat, R., & Mardhiyah, A. (2023). School-based nursing interventions for preventing bullying and reducing its incidence on students: A scoping review. *International Journal of Environmental Research and Public Health*, 20(2), 1577. https://doi.org/10.3390/ijerph20021577

YRBSS Overview | DASH | CDC. (n.d.).

https://www.cdc.gov/healthyyouth/data/yrbs/overview.htm

- Zhang, J. (2019). The strain theory of suicide. *Journal of Pacific Rim Psychology*, 13, e27.
- Zhang, J., & Lester, D. (2008). Psychological tensions found in suicide notes: A test for the strain theory of suicide. Archives of Suicide Research, 12(1), 67–73. https://doi.org/10.1080/13811110701800962
- Zhang, J., & Song, Z. (2006). A preliminary test of the strain theory of suicide. *Chinese Journal of Behavioral Medical Science*, *15*(6), 487-489.
- Zhou, J., Li, X., Zhu, D., & Gong, X. (2023). Cyber-victimization and suicidal ideation in adolescents: a longitudinal moderated mediation model. *Journal of Youth and Adolescence*, 52(1), 122-133.
- Zhu, A. Y., & Crawford, M. H. (2023). Risk factors associated with adolescent suicidality before and during the COVID-19 pandemic. *Suicide and Life-Threatening Behavior*, 53(6), 981-993.
- Zhu, C., Huang, S., Evans, R., & Zhang, W. (2021). Cyberbullying among adolescents and children: a comprehensive review of the global situation, risk factors, and preventive measures. *Frontiers in Public Health*, 9. https://doi.org/10.3389/fpubh.2021.634909

APPENDICES

Appendix A: IRB Approval for Study



After review of your application, it has been determined the proposed activities described do not meet the definition of research with human subjects according to federal regulations and IRB approval is not needed.

Thank you for the time and effort put into preparing and submitting your application. If you have any further questions, please call the Office of Research Compliance at (936) 261-1553.

Marco Robinson

Marco L. Robinson, M.A.Ed. Director, Research Regulatory Compliance Office of Research Compliance Email: <u>mircbinson@oxamu.edu</u>

www.pvami.edu

Cilice of Research Compliance P.O. Boy any Mail Stop 2700 Prairie a wy Twasy 1744 Proces (N3h) 201-11 CV 1188 Fax (USh) 201-3021

Appendix B: Definition of Terms

- 1. *Suicidality:* According to the American Psychological Association (APA), suicidality is defined as "the risk of suicide, usually indicated by suicidal ideation or intent, especially as evident in the presence of a well-elaborated suicidal plan (VandenBos, 2007).
- Physical Wellness: Refers to physical health. In this study, it is a composite measure of Body Mass Index and reported physical activity.
- **3.** *Mental health:* According to the World Health Organization, "Mental health is a state of mental well-being that enables people to cope with the stresses of life, realize their abilities, learn well and work well, and contribute to their community" (WHO, 2022).
- **4.** *Bullying:* The CDC defines bullying as "any unwanted aggressive behavior(s) by another youth or group of youths, who are not siblings or current dating partners, that involves an observed or perceived power imbalance, and is repeated multiple times or is highly likely to be repeated" (CDC, n.d.).
- **5.** *Cyberbullying:* Threatening behavior toward another, often a peer delivered by way of technology.
- 6. Victimization: Refers to youth self-report of having experienced abuse. In this study it refers to a) physical victimization, b) sexual victimization, c) dating violence, d) bullying, and e) cyberbullying.

CURRICULUM VITAE

Email: kshankar@pvamu.edu/ KaviyaShankar@hotmail.com

EDUCATION

- Master of Science in Criminology and Criminal Justice Science, Department of Criminology, University of Madras, India, 2019
- Diploma in Juvenile Justice and Juvenile Psychology, Department of Criminology, University of Madras, India, 2019
- Bachelor of Science in Forensic Science, Jain University, Bengaluru, India, 2017

TEACHING EXPERIENCE

- Graduate Teaching Assistant, Department of Justice Studies, Prairie View A&M University, Texas
 - Introduction to Criminal Justice (Fall 2023 and Spring 2024)

RESEARCH EXPERIENCE

- Graduate Research Assistant, Department of Justice Studies, Prairie View A&M University, Texas, 2021 2023
- Research Assistant in a research project, *'Situational Analysis of Children Released from Special Home and Children Discharged from Children's Home'* conducted by UNICEF in collaboration with Social Defense and the Department of Criminology, University of Madras, 2019

PUBLICATIONS

- Kethineni, S., Moeyaert, M., Shankar, K., & Sundaravadivelu, D. (2024). Sexual Violence in Public Schools in the United States. *Journal of School Violence*, 1–13. https://doi.org/10.1080/15388220.2024.2309560
- Sam, S. S., & **Shankar, K.** (Pending for 2024). A System in Crisis: Prescriptions for Texas Juvenile Justice, *Journal of Contemporary Juvenile Justice*.

ACADEMIC SERVICE

Reviewer, Journal of Family Strengths

Reviewer, IGI Global